AGREEMENT

BETWEEN

TELUS

AND

THE INTERNATIONAL BROTHERHOOD OF ELECTRICAL WORKERS

LOCAL 348

CLERICAL/OPERATOR SERVICES EMPLOYEES

NOVEMBER 29, 1999 – DECEMBER 31, 2000

(For purposes of this Collective Agreement TELUS Communications (a Division of TELUS Communications Inc.) and TELUS Advanced Communications & TELUS Management Services (Divisions of TELUS Services Inc.)

AGREEMENT

BETWEEN

TELUS

AND

THE INTERNATIONAL BROTHERHOOD OF ELECTRICAL WORKERS

LOCAL 348

THIS AGREEMENT is made by and between TELUS designated and known as the "Employer" also referred to as the "Company" or "Management" and Local 348, the International Brotherhood of Electrical Workers, referred to as the "Union" representing Clerical and Operator Services of TELUS as referenced in the certificate issued December 6, 1998 to the International Brotherhood of Electrical Workers, Local 348.

As a result of collective bargaining negotiations by and between the Company and the Union, the following conditions of employment and wage schedules are agreed to and shall be implemented on November 29, 1999 unless otherwise specified in the Agreement.

SIGNED ON BEHALF OF TELUS:

D. Windsor DIRECTOR, LABOUR RELATIONS

And the Management Negotiating Committee:

Lydia Emanuel EMPLOYEE RELATIONS ADVISOR

Corrina Chetley Joyce Sawchuk

Patti Skogg Bruce Smith Terri Williams

SIGNED ON BEHALF OF LOCAL 348, INTERNATIONAL BROTHERHOOD OF ELECTRICAL WORKERS:

J. Ewing PRESIDENT

M. Semeniuk BUSINESS MANAGER

And the Clerical/Operator Services Negotiating Committee:

Grace Thostenson ASSISTANT BUSINESS MANAGER

Marg Casson Elma Coyle Gerrie Dakers Krista Heiligers Valerie Montgomery Sandi Mutter Stacey Properzi Anne-Marie Taylor

TABLE OF CONTENTS

		PAG	
ART	ICLE 1	I: SCOPE	1
ART		2: CLASSIFICATIONS (OP SVCS)	1
ART		B: DEFINITIONS (CLERICAL)	•4
	3.01	Employee2-	
	3.02	Probationary Employee	
	3.03	Tour Employee	
	3.04	Tours Of Duty	
	3.05	Overtime	
	3.06	Session	
	3.07	Basic Rate Of Pay	
	3.08	Basic Hourly Rate	
	3.09	Basic Wages	
	3.10	Headquarters	
	3.11	Furlough	4
ART		4: DEFINITIONS (OP SVCS)	-6
	4.01	Regular Full-Time Employee	
	4.02	Regular Part-Time Employee	
	4.03	Probationary Employee	
	4.04	Casual Employee	
	4.05	Occasional Employee	
	4.06	Casual And Occasional Employee's Probationary Period	
	4.07	Furlough	
	4.08	Basic Rate Of Pay	
	4.09	Basic Hourly Rate	
	4.10	Basic Wages	
	4.11	Overtime	
	4.12	Session	
	4.13	Meal Period	
	4.14	Tour Of Duty	
	4.15	Tour Bidding	
	4.16	Definition of a Day	

5.01 Union Recognition 7 5.02 Union Representative 7 5.03 Shop Steward List 7 5.04 Memorandum Of Agreement 7 5.05 Bulletin Boards 7 5.06 Directives 7 7 5.06 Directives 7 7 5.06 Directives 7 7 5.06 Directives 7 7 5.06 Directives 7 7 5.06 Directives 7 7 5.06 Directives 7 7 5.06 Directives 7 7 5.06 Directives 7 7 5.06 Directives 7 7 5.07 Directives 7 8.01 Local Union Representatives/Time Paid for by The Company 7 7 5.02 Time Off for Union Business/Time Billed to The Union 8 7.04 Union Negotiating Committee 8 8 7.05 Employee' Name & Address 8 7.05	5.02 Union Representative	
5.03 Shop Steward List. 7 5.04 Memorandum Of Agreement. 7 5.05 Bulletin Boards. 7 5.06 Directives. 7 ARTICLE 6: TIME OFF FOR UNION REPRESENTATIVES 7-8 6.01 Local Union Representatives/Time Paid for by The Company 7 6.02 Time Off for Union Business/Time Billed to The Union 8 6.03 Union Negotitating Committee 8 6.04 Meeting New Employees 8 ARTICLE 7: DUES DEDUCTION 8-9 7.01 Union Dues Deduction 8-9 7.02 Employees' Name & Address 8 7.03 Employeer Harmless From Claims 8 7.04 Union Members 8 7.05 Transfers Into Bargaining Unit 9 7.06 Dues Deduction Changes 9 ARTICLE 8: WORKING SUPERVISION (CLERICAL) 9 ARTICLE 9: MANAGEMENT RIGHTS 9 ARTICLE 10: CONTRACTING OUT 10 10.01 Contracting Out 10 10.02 Retain Work	5.02 Union Representative	7
5.04 Memorandum Of Agreement 7 5.05 Bulletin Boards 7 5.06 Directives 7 ARTICLE 6: TIME OFF FOR UNION REPRESENTATIVES 7-8 6.01 Local Union Representatives/Time Paid for by The Company 7 6.02 Time Off for Union Business/Time Billed to The Union 8 6.03 Union Negotiating Committee 8 6.04 Meeting New Employees 8 ARTICLE 7: DUES DEDUCTION 8-9 7.01 Union Dues Deduction 8 7.02 Employees' Name & Address 8 7.03 Employeer Harmless From Claims 8 7.04 Union Members 8 7.05 Transfers Into Bargaining Unit 9 7.06 Dues Deduction Changes 9 ARTICLE 8: WORKING SUPERVISION (CLERICAL) 9 ARTICLE 9: MANAGEMENT RIGHTS 9 ARTICLE 10: CONTRACTING OUT 10 10.01 Contracting Out 10 10.02 Retain Work 10 11.01 No Discrimination 10 11.02 No Pay Reduction </td <td></td> <td>7</td>		7
5.05 Bulletin Boards		
5.06 Directives 7 ARTICLE 6: TIME OFF FOR UNION REPRESENTATIVES 7-8 6.01 Local Union Representatives/Time Paid for by The Company 7 6.02 Time Off for Union Business/Time Billed to The Union 8 6.03 Union Negotiating Committee 8 6.04 Meeting New Employees 8 ARTICLE 7: DUES DEDUCTION 8-9 7.01 Union Dues Deduction 8 7.02 Employees' Name & Address 8 7.03 Employer Harmless From Claims 8 7.04 Union Members 8 7.05 Transfers Into Bargaining Unit 9 7.06 Dues Deduction Changes 9 ARTICLE 8: WORKING SUPERVISION (CLERICAL) 9 ARTICLE 9: MANAGEMENT RIGHTS 9 ARTICLE 10: CONTRACTING OUT 10 10.01 Contracting Out 10 10.02 Retain Work 10 11.03 Discrimination 10 11.04 No Discrimination 10 11.03 Discrimination Factors 10 <td></td> <td></td>		
ARTICLE 6: TIME OFF FOR UNION REPRESENTATIVES 7-8 6.01 Local Union Representatives/Time Paid for by The Company 7 6.02 Time Off for Union Business/Time Billed to The Union 8 6.03 Union Negotiating Committee 8 6.04 Meeting New Employees 8 ARTICLE 7: DUES DEDUCTION 8-9 7.01 Union Dues Deduction 8 7.02 Employees' Name & Address 8 7.03 Employees' Name & Address 8 7.04 Union Members 8 7.05 Transfers Into Bargaining Unit 9 7.06 Dues Deduction Changes 9 ARTICLE 8: WORKING SUPERVISION (CLERICAL) 9 ARTICLE 9: MANAGEMENT RIGHTS 9 ARTICLE 10: CONTRACTING OUT 10 10.01 Contracting Out 10 10.02 Retain Work 10 11.01 No Discrimination 10 11.02 No Pay Reduction 10 11.03 Discrimination Factors 10 12.01 Seniority 10 12.02 Seniority List <th></th> <th></th>		
6.01 Local Union Representatives/Time Paid for by The Company 7 6.02 Time Off for Union Business/Time Billed to The Union 8 6.03 Union Negotiating Committee 8 6.04 Meeting New Employees 8 ARTICLE 7: DUES DEDUCTION 8-9 7.01 Union Dues Deduction 8 7.02 Employees' Name & Address 8 7.03 Employees' Name & Address 8 7.04 Union Members 8 7.05 Transfers Into Bargaining Unit 9 7.06 Dues Deduction Changes 9 7.06 Dues Deduction Changes 9 ARTICLE 8: WORKING SUPERVISION (CLERICAL) 9 ARTICLE 9: MANAGEMENT RIGHTS 9 ARTICLE 10: CONTRACTING OUT 10 10.01 Contracting Out 10 10.02 Retain Work 10 11.01 No Discrimination 10 11.02 No Pay Reduction 10 11.03 Discrimination Factors 10 12.01 Seniority 10 12.02 Seniority <t< th=""><th>5.06 Directives</th><th>7</th></t<>	5.06 Directives	7
6.01 Local Union Representatives/Time Paid for by The Company 7 6.02 Time Off for Union Business/Time Billed to The Union 8 6.03 Union Negotiating Committee 8 6.04 Meeting New Employees 8 ARTICLE 7: DUES DEDUCTION 8-9 7.01 Union Dues Deduction 8 7.02 Employees' Name & Address 8 7.03 Employees' Name & Address 8 7.04 Union Members 8 7.05 Transfers Into Bargaining Unit 9 7.06 Dues Deduction Changes 9 7.06 Dues Deduction Changes 9 ARTICLE 8: WORKING SUPERVISION (CLERICAL) 9 ARTICLE 9: MANAGEMENT RIGHTS 9 ARTICLE 10: CONTRACTING OUT 10 10.01 Contracting Out 10 10.02 Retain Work 10 11.01 No Discrimination 10 11.02 No Pay Reduction 10 11.03 Discrimination Factors 10 12.01 Seniority 10 12.02 Seniority <t< th=""><th>ARTICLE 6: TIME OFF FOR UNION REPRESENTATIVES</th><th>7-8</th></t<>	ARTICLE 6: TIME OFF FOR UNION REPRESENTATIVES	7-8
6.02 Time Off for Union Business/Time Billed to The Union 8 6.03 Union Negotiating Committee 8 6.04 Meeting New Employees 8 ARTICLE 7: DUES DEDUCTION 8-9 7.01 Union Dues Deduction 8 7.02 Employees' Name & Address 8 7.03 Employees' Name & Address 8 7.04 Union Members 8 7.05 Transfers Into Bargaining Unit 9 7.06 Dues Deduction Changes 9 ARTICLE 8: WORKING SUPERVISION (CLERICAL) 9 ARTICLE 9: MANAGEMENT RIGHTS 9 ARTICLE 10: CONTRACTING OUT 10 10.01 Contracting Out 10 10.02 Retain Work 10 10.02 Retain Work 10 11.03 Discrimination 10 11.03 Discrimination Factors 10 12.01 Seniority 10 12.02 Seniority List 10 ARTICLE 13: SERVICE DATE 11 ARTICLE 14: GRIEVANCE PROCEDURE 11-14		
6.04 Meeting New Employees 8 ARTICLE 7: DUES DEDUCTION 8-9 7.01 Union Dues Deduction 8 7.02 Employees' Name & Address 8 7.03 Employees' Name & Address 8 7.04 Union Dues Deduction 8 7.05 Employer Harmless From Claims 8 7.06 Dues Deduction Changes 9 7.06 Dues Deduction Changes 9 ARTICLE 8: WORKING SUPERVISION (CLERICAL) 9 ARTICLE 9: MANAGEMENT RIGHTS 9 ARTICLE 10: CONTRACTING OUT 10 10.01 Contracting Out 10 10.02 Retain Work 10 11.01 No Discrimination 10 11.02 No Pay Reduction 10 11.03 Discrimination Factors 10 12.02 Seniority 10 12.02 Seniority 10 12.02 Seniority List 10 12.02 Seniority List 10 12.02 Seniority List 10 <		
6.04 Meeting New Employees 8 ARTICLE 7: DUES DEDUCTION 8-9 7.01 Union Dues Deduction 8 7.02 Employees' Name & Address 8 7.03 Employees' Name & Address 8 7.04 Union Dues Deduction 8 7.05 Employer Harmless From Claims 8 7.06 Dues Deduction Changes 9 7.06 Dues Deduction Changes 9 ARTICLE 8: WORKING SUPERVISION (CLERICAL) 9 ARTICLE 9: MANAGEMENT RIGHTS 9 ARTICLE 10: CONTRACTING OUT 10 10.01 Contracting Out 10 10.02 Retain Work 10 11.01 No Discrimination 10 11.02 No Pay Reduction 10 11.03 Discrimination Factors 10 12.02 Seniority 10 12.02 Seniority 10 12.02 Seniority List 10 12.02 Seniority List 10 12.02 Seniority List 10 <	6.03 Union Negotiating Committee	8
7.01 Union Dues Deduction 8 7.02 Employees' Name & Address 8 7.03 Employer Harmless From Claims 8 7.04 Union Members 8 7.05 Transfers Into Bargaining Unit 9 7.06 Dues Deduction Changes 9 ARTICLE 8: WORKING SUPERVISION (CLERICAL) 9 ARTICLE 9: MANAGEMENT RIGHTS 9 ARTICLE 10: CONTRACTING OUT 10 10.01 Contracting Out 10 10.02 Retain Work 10 11.01 No Discrimination 10 11.02 No Pay Reduction 10 11.03 Discrimination Factors 10 12.01 Seniority 10 12.02 Seniority 10 12.02 Seniority List 10 1		
7.01 Union Dues Deduction 8 7.02 Employees' Name & Address 8 7.03 Employer Harmless From Claims 8 7.04 Union Members 8 7.05 Transfers Into Bargaining Unit 9 7.06 Dues Deduction Changes 9 ARTICLE 8: WORKING SUPERVISION (CLERICAL) 9 ARTICLE 9: MANAGEMENT RIGHTS 9 ARTICLE 10: CONTRACTING OUT 10 10.01 Contracting Out 10 10.02 Retain Work 10 11.01 No Discrimination 10 11.02 No Pay Reduction 10 11.03 Discrimination Factors 10 12.01 Seniority 10 12.02 Seniority 10 12.02 Seniority List 10 1		<u>ه</u> م
7.02 Employees' Name & Address 8 7.03 Employer Harmless From Claims 8 7.04 Union Members 8 7.05 Transfers Into Bargaining Unit 9 7.06 Dues Deduction Changes 9 ARTICLE 8: WORKING SUPERVISION (CLERICAL) 9 ARTICLE 9: MANAGEMENT RIGHTS 9 ARTICLE 10: CONTRACTING OUT 10 10.01 Contracting Out 10 10.02 Retain Work 10 11.01 No Discrimination 10 11.02 No Pay Reduction 10 11.03 Discrimination Factors 10 12.01 Seniority 10 12.02 Seniority List 10 12.02 Seniority List 10 12.02 Seniority List 10 12.02 Seniority List 10 12.03 SERVICE DATE 11 ARTICLE 14: GRIEVANCE PROCEDURE 11-14 ARTICLE 15: DISCIPLINE, RESIGNATION AND DISMISSAL 14-15		
7.03 Employer Harmless From Claims 8 7.04 Union Members 8 7.05 Transfers Into Bargaining Unit 9 7.06 Dues Deduction Changes 9 ARTICLE 8: WORKING SUPERVISION (CLERICAL) 9 ARTICLE 9: MANAGEMENT RIGHTS 9 ARTICLE 10: CONTRACTING OUT 10 10.01 Contracting Out 10 10.02 Retain Work 10 11.01 No Discrimination 10 11.02 No Pay Reduction 10 11.03 Discrimination Factors 10 12.02 Seniority 10 12.02 Seniority 10 12.02 Seniority List 10 ARTICLE 13: SERVICE DATE 11 ARTICLE 14: GRIEVANCE PROCEDURE 11-14 ARTICLE 15: DISCIPLINE, RESIGNATION AND DISMISSAL 14-15		
7.04 Union Members 8 7.05 Transfers Into Bargaining Unit 9 7.06 Dues Deduction Changes 9 ARTICLE 8: WORKING SUPERVISION (CLERICAL) 9 ARTICLE 9: MANAGEMENT RIGHTS 9 ARTICLE 10: CONTRACTING OUT 10 10.01 Contracting Out 10 10.02 Retain Work 10 ARTICLE 11: DISCRIMINATION 10 11.01 No Discrimination 10 11.02 No Pay Reduction 10 11.03 Discrimination Factors 10 12.02 Seniority 10 12.02 Seniority 10 12.02 Seniority List 10 ARTICLE 13: SERVICE DATE 11 ARTICLE 14: GRIEVANCE PROCEDURE 11-14 ARTICLE 15: DISCIPLINE, RESIGNATION AND DISMISSAL 14-15		
7.05 Transfers Into Bargaining Unit 9 7.06 Dues Deduction Changes 9 ARTICLE 8: WORKING SUPERVISION (CLERICAL) 9 ARTICLE 9: MANAGEMENT RIGHTS 9 ARTICLE 10: CONTRACTING OUT 10 10.01 Contracting Out. 10 10.02 Retain Work. 10 11.01 No Discrimination 10 11.02 No Pay Reduction 10 11.03 Discrimination Factors 10 11.03 Discrimination Factors 10 12.01 Seniority 10 12.02 Seniority List 10 12.03 Senvice DATE 11 ARTICLE 14: GRIEVANCE PROCEDURE 11-14 ARTICLE 15: DISCIPLINE, RESIGNATION AND DISMISSAL 14-15		
7.06 Dues Deduction Changes 9 ARTICLE 8: WORKING SUPERVISION (CLERICAL) 9 ARTICLE 9: MANAGEMENT RIGHTS 9 ARTICLE 10: CONTRACTING OUT 10 10.01 Contracting Out. 10 10.02 Retain Work. 10 ARTICLE 11: DISCRIMINATION 10 11.01 No Discrimination 10 11.02 No Pay Reduction 10 11.03 Discrimination Factors 10 11.03 Discrimination Factors 10 12.01 Seniority 10 12.02 Seniority List 10 ARTICLE 13: SERVICE DATE 11 ARTICLE 14: GRIEVANCE PROCEDURE 11-14 ARTICLE 15: DISCIPLINE, RESIGNATION AND DISMISSAL 14-15	7.04 Union Members	0
ARTICLE 8: WORKING SUPERVISION (CLERICAL) 9 ARTICLE 9: MANAGEMENT RIGHTS 9 ARTICLE 10: CONTRACTING OUT 10 10.01 Contracting Out. 10 10.02 Retain Work 10 ARTICLE 11: DISCRIMINATION 10 11.01 No Discrimination. 10 11.02 No Pay Reduction. 10 11.03 Discrimination Factors 10 12.01 Seniority. 10 12.02 Seniority List. 10 12.03 Seniority List. 10 11 ARTICLE 14: GRIEVANCE PROCEDURE 11-14 ARTICLE 15: DISCIPLINE, RESIGNATION AND DISMISSAL 14-15		
ARTICLE 9: MANAGEMENT RIGHTS. 9 ARTICLE 10: CONTRACTING OUT 10 10.01 Contracting Out. 10 10.02 Retain Work. 10 ARTICLE 11: DISCRIMINATION 10 11.01 No Discrimination. 10 11.02 No Pay Reduction 10 11.03 Discrimination Factors 10 12.01 Seniority. 10 12.02 Seniority List 10 ARTICLE 13: SERVICE DATE. 11 ARTICLE 14: GRIEVANCE PROCEDURE 11-14 ARTICLE 15: DISCIPLINE, RESIGNATION AND DISMISSAL 14-15	7.00 Dues Deduction Changes	5
ARTICLE 10: CONTRACTING OUT 10 10.01 Contracting Out. 10 10.02 Retain Work. 10 ARTICLE 11: DISCRIMINATION 10 11.01 No Discrimination. 10 11.02 No Pay Reduction. 10 11.03 Discrimination Factors 10 12.01 Seniority. 10 12.02 Seniority List 10 12.02 Seniority List 10 ARTICLE 13: SERVICE DATE 11 ARTICLE 14: GRIEVANCE PROCEDURE 11-14 ARTICLE 15: DISCIPLINE, RESIGNATION AND DISMISSAL 14-15	ARTICLE 8: WORKING SUPERVISION (CLERICAL)	9
10.01 Contracting Out		
10.01 Contracting Out	ARTICLE 9: MANAGEMENT RIGHTS	9
10.02 Retain Work 10 ARTICLE 11: DISCRIMINATION 10 11.01 No Discrimination 10 11.02 No Pay Reduction 10 11.03 Discrimination Factors 10 ARTICLE 12: SENIORITY 10 12.01 Seniority 10 12.02 Seniority List 10 ARTICLE 13: SERVICE DATE 11 ARTICLE 14: GRIEVANCE PROCEDURE 11-14 ARTICLE 15: DISCIPLINE, RESIGNATION AND DISMISSAL 14-15		
11.01 No Discrimination.1011.02 No Pay Reduction.1011.03 Discrimination Factors10ARTICLE 12: SENIORITY.1012.01 Seniority.1012.02 Seniority List1010 ARTICLE 13: SERVICE DATE.11ARTICLE 14: GRIEVANCE PROCEDURE11-14ARTICLE 15: DISCIPLINE, RESIGNATION AND DISMISSAL14-15	ARTICLE 10: CONTRACTING OUT	. 10
11.01 No Discrimination.1011.02 No Pay Reduction.1011.03 Discrimination Factors10ARTICLE 12: SENIORITY.1012.01 Seniority.1012.02 Seniority List1010 ARTICLE 13: SERVICE DATE.11ARTICLE 14: GRIEVANCE PROCEDURE11-14ARTICLE 15: DISCIPLINE, RESIGNATION AND DISMISSAL14-15	ARTICLE 10: CONTRACTING OUT	. 10 . 10
11.02 No Pay Reduction1011.03 Discrimination Factors10ARTICLE 12: SENIORITY1012.01 Seniority1012.02 Seniority List10ARTICLE 13: SERVICE DATE11ARTICLE 14: GRIEVANCE PROCEDURE11-14ARTICLE 15: DISCIPLINE, RESIGNATION AND DISMISSAL14-15	ARTICLE 10: CONTRACTING OUT 10.01 Contracting Out 10.02 Retain Work	. 10 . 10 . 10
11.03 Discrimination Factors 10 ARTICLE 12: SENIORITY 10 12.01 Seniority 10 12.02 Seniority List 10 ARTICLE 13: SERVICE DATE 11 ARTICLE 14: GRIEVANCE PROCEDURE 11-14 ARTICLE 15: DISCIPLINE, RESIGNATION AND DISMISSAL 14-15	ARTICLE 10: CONTRACTING OUT 10.01 Contracting Out 10.02 Retain Work ARTICLE 11: DISCRIMINATION	. 10 . 10 . 10 . 10
ARTICLE 12: SENIORITY	ARTICLE 10: CONTRACTING OUT 10.01 Contracting Out. 10.02 Retain Work ARTICLE 11: DISCRIMINATION 11.01 No Discrimination.	. 10 . 10 . 10 . 10 . 10
12.01 Seniority	ARTICLE 10: CONTRACTING OUT 10.01 Contracting Out 10.02 Retain Work ARTICLE 11: DISCRIMINATION 11.01 No Discrimination 11.02 No Pay Reduction	. 10 . 10 . 10 . 10 . 10 . 10
12.02 Seniority List 10 ARTICLE 13: SERVICE DATE 11 ARTICLE 14: GRIEVANCE PROCEDURE 11-14 ARTICLE 15: DISCIPLINE, RESIGNATION AND DISMISSAL 14-15	ARTICLE 10: CONTRACTING OUT 10.01 Contracting Out 10.02 Retain Work ARTICLE 11: DISCRIMINATION 11.01 No Discrimination 11.02 No Pay Reduction	. 10 . 10 . 10 . 10 . 10 . 10
12.02 Seniority List 10 ARTICLE 13: SERVICE DATE 11 ARTICLE 14: GRIEVANCE PROCEDURE 11-14 ARTICLE 15: DISCIPLINE, RESIGNATION AND DISMISSAL 14-15	ARTICLE 10: CONTRACTING OUT 10.01 Contracting Out. 10.02 Retain Work. ARTICLE 11: DISCRIMINATION 11.01 No Discrimination. 11.02 No Pay Reduction. 11.03 Discrimination Factors ARTICLE 12: SENIORITY.	. 10 . 10 . 10 . 10 . 10 . 10 . 10
ARTICLE 14: GRIEVANCE PROCEDURE 11-14 ARTICLE 15: DISCIPLINE, RESIGNATION AND DISMISSAL 14-15	ARTICLE 10: CONTRACTING OUT 10.01 Contracting Out 10.02 Retain Work. ARTICLE 11: DISCRIMINATION 11.01 No Discrimination. 11.02 No Pay Reduction 11.03 Discrimination Factors ARTICLE 12: SENIORITY 12.01 Seniority	. 10 . 10 . 10 . 10 . 10 . 10 . 10 . 10
ARTICLE 15: DISCIPLINE, RESIGNATION AND DISMISSAL	ARTICLE 10: CONTRACTING OUT 10.01 Contracting Out 10.02 Retain Work. ARTICLE 11: DISCRIMINATION 11.01 No Discrimination. 11.02 No Pay Reduction 11.03 Discrimination Factors ARTICLE 12: SENIORITY 12.01 Seniority	. 10 . 10 . 10 . 10 . 10 . 10 . 10 . 10
ARTICLE 15: DISCIPLINE, RESIGNATION AND DISMISSAL	ARTICLE 10: CONTRACTING OUT 10.01 Contracting Out. 10.02 Retain Work. ARTICLE 11: DISCRIMINATION 11.01 No Discrimination. 11.02 No Pay Reduction. 11.03 Discrimination Factors. ARTICLE 12: SENIORITY. 12.01 Seniority 12.02 Seniority List	. 10 . 10 . 10 . 10 . 10 . 10 . 10 . 10
ATTICLE TO DIGUILENCE, REGISTATION AND DIGUIGOAL	ARTICLE 10: CONTRACTING OUT 10.01 Contracting Out. 10.02 Retain Work. ARTICLE 11: DISCRIMINATION 11.01 No Discrimination. 11.02 No Pay Reduction. 11.03 Discrimination Factors ARTICLE 12: SENIORITY. 12.01 Seniority 12.02 Seniority List. ARTICLE 13: SERVICE DATE.	. 10 . 10 . 10 . 10 . 10 . 10 . 10 . 10
15.01 Just Cause 14	ARTICLE 10: CONTRACTING OUT 10.01 Contracting Out 10.02 Retain Work ARTICLE 11: DISCRIMINATION 11.01 No Discrimination 11.02 No Pay Reduction 11.03 Discrimination Factors ARTICLE 12: SENIORITY 12.01 Seniority 12.02 Seniority List ARTICLE 13: SERVICE DATE ARTICLE 14: GRIEVANCE PROCEDURE	. 10 . 10 . 10 . 10 . 10 . 10 . 10 . 10

15.0	2 Discipline	. 14	
15.0	3 Meetings	. 15	
	4 Notice Of Resignation		
	5 Dismissal		
		_	
ARTICI F	16: HOURS OF WORK AND TOURS OF DUTY (CLERICAL)	5-18	
	1 Overtime		
	2 Tours		
10.01	(a) Tour Of Duty		
	(b) Day Tour		
	(c) Evening Tour		
	(d) Night Tour		
	(e) Relief Or Meal Period		
	(f) Basis Of Hours Of Work		
	(g) Meal Period On Saturday, Sunday, Holiday, And Evening Night Tours		
10.0			
	3 Calendar Work Week		
	4 Day Tours Of Duty		
	5 Tour Rotation		
	6 Change Of Tour Payment		
	7 Overtime		
	8 Change of Tour Request		
	9 Days Off - Consecutive		
	0 Deviation From Normal Basic Hours Of Work		
	1 Short Tour		
	2 Break Period		
16.1	3 Meal Period	. 18	
	17: HOURS OF WORK AND TOURS OF DUTY (OP SVCS)		
	1 Tour Employee		
17.0	2 Tours		
	(a) Tour Of Duty	. 18	
	(b) Day Tour	. 18	
	(c) Afternoon Tour	. 18	
	(d) Evening Tour		
	(e) Night Tour	. 18	
	(f) Length of Sits	. 19	
17.0	3 Split Tours		
	4 Minimum Paid Period Of Work For Occasional Or Regular Part-Time		
	5 Relief Period		
	6 Sickness During A Session		
	7 Basic Hours of Work		
	8 Calendar Work Week		
	9 Basic Hours Of Work		
	0 Week's Work		
		. 13	

17.11	Posting of Tour Schedules	
17.12	Tour Changes	
	Rotation of Tours	
ARTICLE '	18: FAMILY DAYS (CLERICAL)	20-21
18.01	Eligibility And Pay Entitlement	
	Scheduling Of Family Days	
	19: OVERTIME - APPLICATION OF OVERTIME REQUIREMENTS	21-24
	Overtime Rate	
	Callout Overtime	
	Required to Work Days Off	
	· Overtime Breaks	
	Authorization Requirement for Overtime	
	Rest Period	
	Vacation Overtime	
	Part-Time Employees	
13.00		
ARTICLE	20: HOLIDAYS AND HOLIDAY PAY	25-28
20.01	Company Holidays	
	Civic Holidays	
	Easter Monday	
	Reduction Of Calendar Work Week	
20.05	Holiday Pay	
	Overtime	
-	21: ANNUAL VACATIONS	
	Vacation Year	
	Vacation Credits	
21.03	Vacation Entitlement Table	30-31
21.04	Vacation Pay, Regular Employees	31
	Vacation Wage Advance	
21.06	Vacation Scheduling	32-33
21.07	Vacation Entitlement Waived / Carry Over	33-34
21.08	Open Vacation Period	
	Travelling Time For Vacation	
21.10	Company Holiday Occurring During Vacation Period	

21.11 Employee Absent Under Sick Leave Regulations	
Example: Vacation Carry Over Calculations	
ARTICLE 22: PAY IN LIEU OF VACATION	
22.01 Casual/Occasional Vacation Pay	
(a) Vacation Pay / Time Off For Hourly Rated Employees	
22.02 Regular Employees Whose Employment Ceases	
ARTICLE 23: MATERNITY LEAVE	
23.01 Entitlement To Maternity Leave	
23.02 Notice To Commence Maternity Leave	
23.03 Notice To Change Length Of Maternity Leave	
23.04 Duration Of Maternity Leave	
23.05 Unable To Perform Function Of Job	
23.06 Returning From Maternity Leave	
23.07 Benefit Contributions During Maternity Leave	
23.08 Suspension Of Increases	
23.09 Pay Deductions	
ARTICLE 24: CHILD CARE LEAVE	
24.01 Entitlement To Child Care Leave	
24.02 Notice To Commence Or Change Child Care Leave	
24.03 Returning From Child Care Leave	
24.04 Child Care Leave Split Between Two Employees	
24.05 Benefit Contributions During Child Care Leave	
24.06 Suspension Of Increases	
24.07 Pay Deductions	
ARTICLE 25: ADOPTION LEAVE	
25.01 Entitlement To Adoption Leave	40
25.02 Notice To Commence Adoption Leave	40
25.03 Returning From Adoption Leave	40
25.04 Benefit Contributions During Adoption Leave	40
25.05 Employer Benefit Contributions During Adoption Leave	40
25.06 Resumption of Union Dues	
25.07 Voluntary Contribution Of Union Dues	
ARTICLE 26: SAFETY	
26.01 Safety Requirements	
26.02 Company Safety Roles	
26.03 Union Safety Roles	41
26.04 Safety Training & Inspections	
26.05 Refusal To Work	
26.06 Refusal To Work Limitations	

26.07 Safety Footwear	41
26.08 Pregnant Employees And Video Display Terminals	
26.09 Safety Representatives	
ARTICLE 27: ACCIDENT COMPENSATION	42-43
27.01 Reporting Accidents	
27.02 Accident Compensation	
27.03 Worker's Compensation For Casual And Occasional Employees	
27.04 Accident Compensation Claim Within 65 Days Of Previous Claim	
27.05 Re-Establishing Accident Compensation Credits	
27.06 Anniversary Date Passed While On Accident Compensation	
27.07 Anniversary Date Passed Between Disabling Work Injury Absences	
27.08 Anniversary Date Passed But Not Increasing Compensation Credits	
ARTICLE 28: SICK LEAVE REGULATIONS	44-49
28.01 Notifying Supervisor of Sick Leave Absence	
28.02 Computing of Sick Leave Credits	
28.03 Sick Leave Schedule	
28.04 Documentation	
28.05 Employee Unable To Continue Work Because Of Illness	
28.06 Frequent Or Extended Amount Of Absence	
28.07 Quarantine	
28.08 Exhaustion of Sick Leave Credits, 22 working days – 70%	
28.09 Resumption of Service Before Exhaustion of Sick Leave Benefits-	
22 Working Days (new illness)	
28.10 Resumption of Service Before Exhaustion of Sick Leave Benefits-	
65 Working Days	
28.11 Sick Leave Credits Exhausted	
28.12 Establishment of New Sick Leave Credits - 65 Working Days	
28.13 Sick Leave Credits And Anniversary Date	
(a) Anniversary Date Passed While On Sick Leave	
(b) Anniversary Date Passed Between Sickness	
(c) Anniversary Date Passed But Not Establishing Increased Credits	
(d) Exhaustion Of Sick Leave Credits	
28.14 Vacation and Holidays During Sick Leave Absence	
28.15 Sickness Or Accident During Vacation Or Furlough	
28.16 Discontinuing Payment Of Benefits	
28.17 Legislative Changes to Sick Leave Regulations	

ARTICLE 29: ALBERTA HEALTH CARE INSURANCE PLAN, SUPPLEMENTARY		
HEALTH, VISION, DENTAL CARE PLANS AND GROUP INSURANCE PLAN	40.50	
29.01 Alberta Health Care Insurance Plan		
29.02 Supplementary Health Plan		
29.02 Supplementary Health Flah		
29.03 Dental Flan		
29.05 Group Insurance Plan		
29.06 Employee Benefits		
ARTICLE 30: REGULAR PART-TIME BENEFITS	50-53	
30.01 Benefit Premiums	50	
30.02 Company Holidays	50-51	
30.03 Paid Absences		
30.04 Progressional Increases	51	
30.05 Seniority	51	
30.06 Seventy-Five Percent Rule	51	
30.07 Sick Leave	51-52	
30.08 Two Pay Periods	53	
30.09 Vacations	53	
ARTICLE 31: TECHNOLOGICAL CHANGE (CLERICAL)	53	
31.01 Decrease Of Work		
31.02 Separation Allowance: Less Than Five Years Of Service		
31.03 Separation Allowance: Five Or More Years Of Service		
31.04 Re-Engaged And Subsequent Separation		
ARTICLE 32: TECHNOLOGICAL CHANGE (OP SVCS)	54	
32.01 Decrease Of Work		
32.02 Separation Allowance: Less Than Five Years Of Service	54	
32.03 Separation Allowance: Five Or More Years Of Service	54	
32.04 Re-Engaged And Subsequent Separation		
	E 1 E C	
ARTICLE 33: LAYOFF (CLERICAL)		
33.01 Order Of Layoff		
33.03 Notice Of Layoff To Union And Establishment Of Joint Committee		
33.04 Notice Of Layoff To Employees		
33.05 Order Of Recall		
33.06 Recall Responsibilities Of Employees		
33.07 Recall And Former Credits		
33.08 Severance Pay		
33.09 Right To Waive Recall		

ARTICLE 34: LAYOFF (OPERATOR SERVICES)	56-58
34.01 Order Of Layoff	56
34.02 Layoff Of Casual And Occasional Employees	
34.03 Notice Of Layoff To Union And Establishment Of Joint Committee	56-57
34.04 Notice Of Layoff To Employees	
34.05 Order Of Recall	
34.06 Recall Responsibilities Of Employees	57
34.07 Recall And Former Credits	
34.08 Severance Pay	58
34.09 Right To Waive Recall	58
ARTICLE 35: JOB EVALUATION	58-60
35.01 Jobs to be Evaluated	
35.02 Requests For Job Evaluation And Job Descriptions	58
35.03 Job Evaluation Appeals	59
35.04 Job Evaluation Steering Committee	59
35.05 Wage Adjustments	60
35.06 Job Evaluation Plan & Breakpoints	60
ARTICLE 36: UPGRADING COURSES	61
ARTICLE 37: JOB POSTINGS, TRANSFERS, SURPLUS PLACEMENTS AND REDEPLOYMENT	61-65
37.01 Job Postings	61
37.02 Job Postings: Out Of Scope Vacancies	
37.02 Job Posting Restrictions	
37.04 Job Posting Contents	
37.05 Job Posting Copies To Union	
37.06 Cannot Fulfill Position	
37.07 Posting Exceptions	
37.08 Selection	
37.09 Notification And Transfer Date	
37.10 Transfers - Operator Services	63
37.11 Redeployment	64-65
ARTICLE 38: TEMPORARY MOVEMENT OF EMPLOYEES	65-69
38.01 Temporary Assignments To Management: Clerical	65-66
38.02 Temporary Assignments To Management: Operator Services	
38.03 Temporary Assignments To A Higher Rated Job	
38.04 Temporary Assignments To Other Bargaining Units	
38.05 Other Temporary Assignments Within The Bargaining Unit	
38.06 Acting Assignments: Clerical/Operator Services	
ARTICLE 39: DIFFERENTIALS AND SPECIAL PAYMENTS(CLERICAL)	69-70

- VIII -

39.01 Tour Differentials	
39.02 Night Tour Differential	70
39.03 Consecutive Saturday Differential	70
39.04 Sunday Premium	
39.05 Training Differential	
39.06 Christmas And New Year's Eve Special Payment	70
ARTICLE 40: DIFFERENTIALS AND SPECIAL PAYMENTS (OP SVCS)	
40.01 Tour Differential	
40.02 Sunday Premium	
40.03 Split Tour Differential	
40.04 Differential Where Relief Is Not Provided	
40.05 Service Assistant (Special Services) Differential	
40.06 Christmas Eve And New Year's Eve Special Payment	
40.07 Consecutive Saturday Differential	
40.08 Training Differential	73
ARTICLE 41: PAY PROVISIONS FOR EMPLOYEES ABSENT FROM DUT	Y 74-77
41.01 Court Witness	74
41.02 Jury Duty	74
41.03 Elections	
41.04 Bereavement Leave	75
41.05 Funeral Leave	75
41.06 Pallbearer Leave	75
41.07 Serious Distress	76
41.08 Medical, Dental Or Other Special Appointments	
41.09 Working For Another Company	76
41.10 Time Lost Other Than Sickness	
41.11 Furlough	
41.12 Absence (Clerical)	
41.13 Absence (Operator Services)	77
ARTICLE 42: MEALS, LODGING, TRANSPORTATION AND	
TRAVELLING TIME	
42.01 Expenditures Necessarily Incurred	78
42.02 Meals	
42.03 Lodging	
42.04 Incidental Expenses	
42.05 Extraordinary Expenses	
42.06 Overtime Meal Allowance	
42.07 Travel Time	
42.08 Transportation Provided	
42.09 Transportation Allowed	80

ARTICLE 43: PAY PROVISIONS AND WAGE SCHEDULES	80-87
43.01 Non-Permanent Experienced Employees	80
43.02 Payments	
43.03 Wage Increases	
43.04 Efficiency, Promotion or Re-evaluation Increases	
43.05 Wage Adjustment Upon Promotion	
43.06 Suspension Of Increases	
43.07 Administration Of Progression Wage Schedule	82
43.08 Amount of Increases	82
43.09 Special Increases	82
43.10 Work Assignment	82
43.11 Variable Pay	
43.12 Wage Schedules	
ARTICLE 44: DURATION AND NEGOTIATIONS	88
44.01 Duration	

44.01		00
44.02	First Negotiating Meeting	88
	Notice to Negotiate or Notice to Terminate	
44.04	Mediation / Conciliation	88

	S OF UNDERSTANDING	
1.	Letters Of Understanding/Letters Of Intent	
2.	Benefits Committee	
3.	Compressed Hours of Work Program Information Technology	
	(Computer Operators)	
4.	Compressed Work Week	
5.	Electronic Monitoring	
6.	Family Days	
7.	Job Allocation Appeal Procedure	
8.	Job Sharing	
	100-103	
9.	Joint Committee Grievance Review	104
10.	Medical And Compassionate Transfers	
11.	Microcomputer Consultant Positions – PLAnet Help Desk	106
12.	Occasional Employees – Residential Sales & Service	107-108
13.		
14.	Sales Incentive Plan	111
15.	SAP Administration	112-114
16.	Service Assistant (Special Services), Customer Service Assistar	nt And
	Service	
	Assistant And Support Duties (Operator Services)	115
17.		
18.	Supplementary Health and Vision Care Plans	117

96

19.	Taxis	118
20.	Telecommunications Concession	119
21.	Trading Of Tours Between Services (Operator Services)	120
22.	Vacation Entitlement	121
23.	Work Experience Programs	122

APPENDICES

Appendix A Travel and Transfer Expenses (Clerical)		
General		
Travel And Expenses Outside The Province		
Travel And Expenses Within The Province		
Transfer Allowances And Expenses		
Employees Travelling To And From Out-Of-Town Jobs		
Transportation		
Appendix B Travel and Transfer Expenses (Op Svcs)		
General		
Travel And Expenses Outside The Province	127	
Travel And Expenses Within The Province		
Transfer Allowances And Expenses		
Employees Travelling To And From Out-Of-Town Jobs		
Transportation		
Appendix C (Op Svcs)		
Working Supervision Policy		

ARTICLE 1 - SCOPE

1.01 The provisions of this Agreement apply to all employees of TELUS Communications, (a Division of TELUS Communications Inc.), TELUS Advanced Communications, and TELUS Management Services, each being Divisions of TELUS Services Inc. as listed in the Canada Industrial Relations Board Certificate (Order No.: 7504-U) issued December 6, 1998 certifying the International Brotherhood of Electrical Workers, Local 348 as the bargaining agent for the Clerical and Operator Services employees of the above-named Employers and such other employees as the parties may agree to include or the Canada Industrial Relations Board may direct.

ARTICLE 2 - CLASSIFICATIONS (Operator Services)

2.01 The provisions of the Operator Services sections of this agreement apply to the following classifications of employees:

Video Conference Room Co-ordinator Morris Plus Co-ordinator Office Assistant Teleconference Representative Video Conference Representative Deaf/Hearing Liaison Network Operator Scheduling Co-ordinator Customer Service Assistant

Payment for these classifications will be in accordance with the wage schedules that form part of this agreement. Employees performing the duties of:

> Service Assistant and Support Service Assistant (Special Services)

shall be paid differentials in accordance with Clause 40.05 and Clause 38.02.

ARTICLE 3 - DEFINITIONS (Clerical)

- 3.01 Employee means any of the following:
 - (a) Regular Full-Time Employee means a person employed in the service of the Company, as covered under the terms of this Agreement and whose employment is expected to continue indefinitely subject to terminating action by either party.
 - (b) Regular Part-Time Employee means a person hired for regular employment on a continuous basis and works a minimum of seventy-five (75) hours averaged over two (2) pay periods. This employee may be required to work full-time to meet the demands of service. The employee shall not work more than seventy-five percent (75%) of the hours of a full-time employee in any calendar year. Such employees will be covered on a prorated basis under the terms of this Agreement and the Pension Plan.

If an employee works more than seventy-five percent (75%) of the hours of a full-time employee then they will be offered regular full-time employment.

(c) A Casual Employee shall work the hours of a Regular Full-Time Employee. Casual Employee means a person hired to fill a vacancy which the Company considers to be temporary or to carry out casual work assignments. Casual Employees will be covered under the terms of this Agreement except for Group Insurance, Alberta Health Care Insurance, Supplementary Health, Vision and Dental Care Plans, Pension Plan, Company Sick Leave and Accident Compensation Benefits.

A Casual Employee shall not be employed for a period beyond nine (9) months. A Casual Employee may replace a Regular Employee absent on maternity leave, child care leave, and/or adoption leave for a period equal to the length of the absence, plus four (4) weeks if training is required or nine (9) months, whichever is greater. There shall be a three (3) month break between each term of employment.

(d) Occasional Employees are employees employed on an hourly or daily basis to cover situations that cannot reasonably be handled by Regular Full-Time, Regular Part-Time or Casual Employees. Such employees will be covered under the terms of this Agreement where terms and conditions are considered applicable, except for Group Insurance, Alberta Health Care Insurance, Supplementary Health, Vision and Dental Care Plans, Company Sick Leave and Accident Compensation Benefits.

Upon completion of six (6) months of employment, Occasional Employees shall be paid an amount equal to five percent (5%) of their gross regular earnings in lieu of health benefits.

- 3.02 Probationary Employee means an employee who has not completed their first six (6) months of service.
- 3.03 Tour Employee

Tour Employee means an employee who is regularly posted on a tour of duty covering day, evening, and/or night tours, Saturdays, Sundays and/or holidays on a regular rotation basis.

3.04 Tours of Duty

The Company reserves the right to establish, for employees in any classification listed herein, presently or in the future, special work weeks starting on any day of the week consistent with Article 16 covering Hours of Work and Tours of Duty.

- 3.05 Overtime means all hours or portion thereof worked by an employee in excess of the basic hours of work as provided for in this Agreement.
- 3.06 Session

Session means the continuous time, not including overtime but including the break period which an employee is scheduled to work without a meal period. A session shall be one-half (1/2) the duration of a tour.

3.07 Basic Rate of Pay means the specified amount of money per day calculated as follows: basic hourly rate times seven and one-half (7 1/2).

- 3.08 Basic Hourly Rate means the specified amount of money per hour which, in accordance with the Wage Schedules that form part of this Agreement, is paid to an employee for each hour worked and all other amounts payable to any employee are completely excluded from the term "basic hourly rate".
- 3.09 Basic Wages means the basic hourly rate accumulated over a period of time, e.g. payroll period.
- 3.10 Headquarters
 - (a) For the purposes of this Agreement, headquarters will be the city or town limits.
 - (b) For employees who have been approved for a Company paid relocation, headquarters shall mean:
 - (1) City of Edmonton: City limits and includes Sherwood Park, St. Albert and Namao.
 - (2) All other cities and towns: City or Town limits.
- 3.11 Furlough

Furlough for the purpose of this Agreement means leave of absence without pay or benefits except as otherwise provided in this Agreement. This does not include Maternity Leave, Adoption Leave or Child Care Leave.

ARTICLE 4 - DEFINITIONS (Operator Services)

Employee means any of the following:

4.01 Regular Full-Time Employee - means a person employed in the services of the Company as covered under the terms of this Agreement and whose employment is expected to continue indefinitely, subject to terminating action by either party.

4.02 Regular Part-Time Employee - means a person hired for regular employment on a continuous basis and works a minimum of seventy (70) hours averaged over a two (2) pay period basis. This employee may be required to work full-time to meet the demands of service. The employee shall not work more than seventy-five percent (75%) of the hours of a full-time employee in any calendar year. Such employees will be covered on a prorated basis under the terms of this Agreement and the Pension Scheme.

If a Regular Part-Time Employee works more than seventy-five percent (75%) of the hours of a full-time employee, then they will be offered regular full-time employment.

- 4.03 Probationary Employee means an employee who has not completed their first six (6) months of service.
- 4.04 Casual Employee means a person hired at an hourly rate to fill a vacancy which the Company considers to be temporary or to carry out casual work assignments. Such employees will be covered under the terms of this Agreement except for Group Life Insurance, Alberta Health Care Insurance, Supplementary Health, Dental and Vision Care Plans, Pension, Sick Leave and Accident Compensation Benefits.

A Casual Employee shall not be employed for a period beyond nine (9) months. There shall be three (3) months between each term of employment.

A Casual Employee shall work regular full-time hours.

4.05 Occasional Employees are employees employed on an hourly or daily basis to cover situations that can not reasonably be handled by Regular Full-Time, Regular Part-Time or Casual Employees. Such employees will be covered under the terms of this Agreement where terms and conditions are considered applicable, except for Group Life Insurance, Alberta Health Care Insurance, Supplementary Health, Dental and Vision Care Plans, Sick Leave and Accident Compensation Benefits.

Upon completion of six (6) months of employment, occasional employees shall be paid an amount equal to five percent (5%) of their gross regular earnings in lieu of health benefits.

4.06 Casual and Occasional Employees hired as Regular Full-Time having three (3) months or more accumulated service will have a three (3) month probationary period.

- 4.07 Furlough for the purposes of this Agreement, furlough is defined as leave of absence without pay. This does not include maternity leave, adoption leave or child care leave.
- 4.08 Basic Rate of Pay means the specified amount of money per day calculated as follows: Basic hourly rate times seven (7).
- 4.09 Basic Hourly Rate means the specified amount of money per hour which, in accordance with the Wage Schedules that form part of this Agreement, is paid to an employee for each hour worked and all other amounts payable to any employee are completely excluded from the term "basic hourly rate".
- 4.10 Basic Wages means the basic hourly rate accumulated over a period of time, e.g. payroll period.
- 4.11 Overtime means all hours or portion thereof worked by an employee in excess of the basic hours of work as provided for in this Agreement.
- 4.12 Session means the continuous time, not including overtime, but including the relief period which an employee is scheduled to work without a meal period. A session shall be approximately one-half (1/2) the duration of a tour.
- 4.13 Meal Period shall not exceed one hour. This does not apply to split tours.
- 4.14 Tour of Duty means the scheduled period of work for an employee on any working day.
- 4.15 (a) A "Tour Bidding" Office shall be an Operator Services Office where Regular Full Time and Regular Part-Time employees have the right to select tours according to seniority and availability.
 - (b) Tour bidding shall be established or discontinued in a particular office by a secret ballot majority vote of the regular employees that are directly affected by the tours (e.g. Network Operator). A tour bidding vote shall be held no more than once in a twelve (12) month period.
- 4.16 A day shall mean the twenty-four (24) hour period beginning at midnight on one day and ending on the following midnight. There shall be a minimum of ten (10) hours between the end of a tour and the start of the next tour. If ten (10) hours is not provided, then overtime shall be paid for the time that infringes.

ARTICLE 5 - UNION RECOGNITION

- 5.01 The Company recognizes the Union as the sole collective bargaining **agent** for the employees covered by this Agreement and hereby consents and agrees to negotiate with the Union or its representatives in any and all matters pertaining to this Agreement which may affect the relationship between the Company and the employees.
- 5.02 The Company also agrees that the Union may have the assistance of such representatives as it may designate in any negotiations or discussions between representatives of the parties hereto subject to Article 6 Time Off For Union Representatives.
- 5.03 The Union agrees to furnish the Company in writing a current list and amendments containing the name, department and location of each Shop Steward, Union Executive member, and negotiating committee member.
- 5.04 When during the life of this Agreement both parties agree that a change is required to any article or item, a Memorandum of Agreement shall be drafted and signed by both parties upon reaching mutual agreement.
- 5.05 Bulletin Boards

The Company will provide bulletin boards suitable for the posting of Union material and shall permit such material to be posted. Any disputes regarding the appropriateness of the material posted shall be referred to the Business Manager of the IBEW and the Director of **Labour** Relations for discussion. Should agreement not be reached, the issue may be submitted to Arbitration.

5.06 Directives

The Company will furnish the Union with copies of directives or letters which affect the working conditions of the employees covered in this Agreement.

ARTICLE 6 - TIME OFF FOR UNION REPRESENTATIVES

- 6.01 Local Union Representatives may investigate and settle grievances, attend joint Company and Union committee meetings or attend to other business of the Union pertaining to the affairs of the Company during working hours for a reasonable length of time, provided however, that arrangements are made with the supervisor(s), subject to service requirements. All time granted accordingly, shall be reported as Union Leave (UL) and paid for by the Company.
- 6.02 Employees shall be allowed time off to attend to other business of the Union, including necessary travelling time. This will be subject to service requirements and management approval. The Union must notify the Employer stating

purpose, name(s) of employee(s) and the time required. Where possible, notification will be given five (5) weeks prior to the days off required. All time granted for this purpose shall be billed to the Union.

- 6.03 Members of the Union Negotiating Committee shall receive their normal basic rate of pay while travelling and attending a negotiation meeting with the Company. Should negotiations extend into an employee's regular day off, equivalent time off will be allowed at a later date. If negotiations extend beyond normal daily working hours, additional or overtime payment shall not be paid.
- 6.04 Shop Stewards shall be given fifteen (15) minutes (private) paid time to meet new employees working in the area.

ARTICLE 7 - DUES DEDUCTIONS

- 7.01 All employees shall have deducted from their wages, deductions levied in accordance with the Union's Constitution and Bylaws, however, membership in the Union shall be at the employee's choice.
- 7.02 The Employer shall forward the deductions to the Union on or before the last day of each month accompanied by a list of employees from whom the deductions have been made. The list shall include employees' home addresses unless the employee makes a written request to the Employer to restrict this information.
- 7.03 The Union will save the Employer harmless from any and all claims which may be made to it by any employee for amounts deducted as herein provided.
- 7.04 (a) All employees covered under the terms of this Agreement who are members of Local 348, IBEW, shall as a condition of employment, remain members of Local 348, IBEW, for the life of this Agreement; provided that, an employee who is a member of the Union is given a period of thirty (30) days prior to the terminating date of this Agreement within which to elect either to remain a Union member or to withdraw from membership.
 - (b) If an employee fails to remain a member of the Union as herein required, the employee shall, notwithstanding their date of employment, be specifically subject to the deductions as provided in Clause 7.01.
- 7.05 All employees transferred into the bargaining unit shall, as a condition of employment, pay the deductions specified in Clause 7.01, concurrent with their transfer date.

7.06 The Union shall notify the Employer in writing ninety (90) days in advance of any approved change in the deductions authorized in accordance with its Constitution and Bylaws.

ARTICLE 8 - WORKING SUPERVISION (Clerical)

8.01 Managerial or Professional (MP) employees shall not do work which is normally carried out by Clerical bargaining unit employees. However, occasions may arise when MP employees are required to do work in training demonstrations, restoration of service where an accident occurs or when other unforeseeable or unpreventable circumstances occur. These occasions should be considered exceptions and not normal operating practice.

ARTICLE 9 - MANAGEMENT RIGHTS

9.01 The management of the operations of the Company and the direction of the working forces, including the right to direct, plan, and control operations and to schedule working hours and the right to hire, promote, demote, transfer, suspend, or discharge employees for just cause or to release employees because of lack of work or the right to introduce new and improved methods or facilities and to manage the operations in the traditional manner, is vested exclusively in the Company, subject to the provisions of this Agreement.

ARTICLE 10 - CONTRACTING OUT

- 10.01 The Employer agrees to have work performed by employees, although at times it may be necessary to have a variety of work performed by outside contractors. The Employer shall give notice in writing to the Union with a copy to the **Director Labour Relations** before contracting out any work that has historically been done by the employees in the bargaining unit. Such notice shall state the reasons for the proposed contracting out.
- 10.02 In an effort to retain work, within the Clerical and Operator Services bargaining unit, which would otherwise be contracted out, the Company agrees to meet with the Union at least once per year, or more frequently where agreed by both parties, to discuss and identify ways of achieving this objective.

ARTICLE 11 - DISCRIMINATION

- 11.01 No discrimination against employees on account of affiliation or relations they may or may not have, will be permitted. All those in charge of employees must take precautions to see that this is not violated.
- 11.02 Nothing herein shall be construed to reduce the pay of an employee now getting a higher rate of pay for the work classified in the wage schedule.
- 11.03 The Company shall not threaten, intimidate, harass, or unlawfully discriminate against an employee for reasons of race, national or ethnic origin, colour, religion, age, sex, marital status, family status, disability, political affiliation, conviction for which a pardon has been granted, or for exercising any rights under this collective agreement or any other federal or provincial regulation.

ARTICLE 12 - SENIORITY

- 12.01 Seniority is the **accumulated** length of service in the employ of the Company **in a bargaining unit position or positions** and shall start from the date the employee last entered **employment with the Company.**
- 12.02 The Employer agrees to provide monthly to the Union on disk or paper copy if required, separate seniority lists which shall include the names, seniority dates, personnel subgroup (classification), pay scale group, and description of job (job title) for all clerical and operator services employees. The lists shall be printed in order of seniority. The format of the reports shall continue to be as are currently provided or as the parties may agree.

ARTICLE 13 - SERVICE DATE

13.01 An employee's service date for the purpose of this Agreement shall be the date the employee entered the service of the Company, and remained continuously employed.

When an Occasional or Casual employee attains regular employment without a break in service, their service date shall be adjusted according to the established Company formula, which will determine sick leave, vacation entitlement and any other entitlement in this Agreement based on service.

ARTICLE 14 - GRIEVANCE PROCEDURE

If any differences concerning the interpretation, application, operation or any alleged violation of the Agreement arises between the Employer and the Union, or between an employee or employees bound by the Collective Agreement and the Employer, including any questions as to whether any difference is arbitrable, it shall be processed according to the following grievance procedure.

In the event that a grievance or grievances involve(s) a group or groups of employees, or in the event that the Union initiates a grievance with respect to the application, operation, interpretation, or any alleged violation of the Collective Agreement, then the grievance shall be submitted in writing to the Director of Labour Relations who shall call a meeting and attempt to resolve the grievance and render a decision within thirty (30) days of receiving the grievance. If the parties are unable to resolve the matter within thirty (30) days following the meeting, the grievance may be advanced to Arbitration.

A grievance in the case of dismissal shall be submitted in writing within thirty (30) days of the date of dismissal and shall commence at Step 2 of the Grievance Procedure.

In the event that a designated official of the Company is not available or the Employer does not respond as required under any step of the grievance procedure, the Union may advance the grievance to the next step of the grievance procedure.

With the consent of the parties the time within which any step or appointment must or should be taken or made may be extended for such period of time as may be agreed between the parties.

With mutual consent of the parties, step(s) of the grievance procedure may be waived.

In the event of the Employer filing a grievance with respect to the application, operation, interpretation, or any alleged violation of the Collective Agreement, it shall be submitted

directly from the Director of **Labour** Relations to the Business Manager of the Union who shall call a meeting to attempt to resolve the grievance and render a decision within thirty (30) days of receiving the grievance. If the parties are unable to resolve the matter within thirty (30) days following the meeting, the grievance may be advanced to Arbitration.

All meetings and/or hearings from Step 2 onward are to be held in Edmonton or Calgary, unless another location is mutually agreed to.

Time periods for the purpose of this Article, shall be calculated on consecutive calendar days.

STEP 1 An employee who believes they have a justifiable grievance, shall discuss the matter with the immediate Manager within thirty (30) days of occurrence and the Manager shall give a decision within five (5) days following the discussion. The employee may be accompanied by a Shop Steward during the discussion if they choose.

If the employee is not satisfied with the decision, the employee may within fifteen (15) days, discuss the matter with a Shop Steward, and the Shop Steward may within fifteen (15) days discuss the matter with the employee's Manager. The decision of the Manager shall be rendered within five (5) days.

STEP 2 If a satisfactory settlement has not been obtained under the previous step, then the employee may within thirty (30) days of the date of the decision under Step 1, render the grievance in writing to the Union. The Union may, within **twenty (20)** days of the date of delivery of the grievance in writing, forward the grievance to the Director of the department concerned, with a copy to the Director of **Labour** Relations. The Director, **or designate** of the department concerned shall call a meeting, attempt to resolve the grievance and render a decision within thirty (30) days of receipt of the grievance.

- STEP 3 If a satisfactory settlement has not been obtained under the previous step, the Union may, within seven (7) days forward the grievance to the Vice-President, or designate of the department concerned. The Vice-President, or designate of the department concerned shall establish a date for the meeting with the Union, the employee (if they choose), and the immediate Manager of the employee concerned (or in the case of a job posting grievance, the hiring manager) within ten (10) days from receipt of the grievance. The Vice-President of the department concerned will render a decision within ten (10) days of the meeting being held.
- **STEP 4** Should the parties agree, any matter may be referred to a single arbitrator.

A grievance which is not resolved through the previous steps shall be adjudicated in accordance with the following:

- (a) Either of the parties shall, within **ninety (90)** days, notify the other party, in writing, of its desire to submit the grievance to arbitration. The notice shall contain the name of the notifying party's appointee to an Arbitration Board and/or the names of up to three individuals acceptable as a single arbitrator.
- (b) The party receiving the notice shall within ten (10) days following receipt of the notice, inform the other party of the name of its appointee to the Arbitration Board or its acceptance of a name from those proposed as a single arbitrator.
- (c) If agreement on a single arbitrator cannot be reached, the matter shall revert to an Arbitration Board.
- (d) If the recipient of the notice fails to name an appointee to the Arbitration Board within the prescribed time limits, the appointment shall be made by the Minister of Labour upon request of either party.
- (e) The two (2) appointees shall, within ten (10) days of the second appointment, appoint a third person who shall be the Chairman of the Arbitration Board. If the two (2) appointees fail to agree upon a Chairman within the ten (10) day limit, either party may request the Minister of Labour to appoint a Chairman.

(f) After the Arbitration Board has been formed by the above procedure or the single Arbitrator selected, the Board or Arbitrator shall meet within fifteen (15) days of the appointment of the Chairman of the Arbitration Board or Arbitrator and hear such evidence as the parties may desire to present to assure a full, fair hearing, and shall render a decision in writing to the parties within fifteen (15) days after the completion of the hearing, unless such time is extended by mutual consent between the parties.

The Arbitration Board or Arbitrator shall have such power as defined and provided in the governing legislation.

- (g) The Chairman shall have the authority to render the decision with the concurrence of either of the other members and a decision thus rendered shall be final and binding on the parties hereto. If no majority decision is reached then the Chairman shall render the decision.
- (h) A single Arbitrator shall have the authority to render a decision which shall be final and binding on the parties.
- (i) The Arbitration Board or Arbitrator by its decision, shall not alter, amend or change the terms of this Collective Agreement.
- (j) Each of the parties to this Agreement shall bear the expense of its nominee to the Arbitration Board. The fees and expenses of the Chairman or single Arbitrator shall be borne equally by the two (2) parties to the dispute.

ARTICLE 15 - DISCIPLINE, RESIGNATION AND DISMISSAL

- 15.01 Employees shall not be disciplined or dismissed except for just cause.
- 15.02 Discipline

When an employee is disciplined for just cause by way of written reprimand, suspension, dismissal or demotion, the Supervisor shall give the employee notice in writing as to the reason(s) for such action, and shall forward a copy of the notice to the Union.

15.03 Meetings

The Employer must identify the purpose of a meeting prior to its commencement. The Employer must advise the employee of their right to have a Shop Steward present at meetings as outlined below in (1) and (2). An employee may request the presence of a Shop Steward, at any meeting which is:

- (1) identified by the Employer as being disciplinary in nature; or,
- (2) for purposes of investigation and where Security staff or Employer representative(s) other than the employee's immediate Supervisor are present.

Where a Shop Steward is available, the request shall be granted. All reasonable effort will be made to schedule the meeting at a time when a Shop Steward is available.

15.04 Notice of Resignation

A regular full-time or regular part-time employee resigning from the Company's employ shall be expected to give at least two (2) weeks' notice in writing.

15.05 Dismissal

Employees dismissed shall be paid in full for all wages due up to the time of dismissal.

ARTICLE 16 - HOURS OF WORK AND TOURS OF DUTY (Clerical)

The work week shall be thirty-seven and one-half (37 1/2) hours.

- 16.01 Overtime means all hours or portion thereof worked by an employee in excess of seven and one-half (7 1/2) hours in a day or thirty-seven and one-half (37 1/2) hours in a week.
- 16.02 (a) Tour of Duty means the scheduled period of work for an employee on any working day.
 - (b) Day Tour a seven and one-half (7 1/2) hour tour which falls between 7:00 A.M. and 6:00 P.M.
 - (c) Evening Tour a seven and one-half (7 1/2) hour tour ending between 6:01 P.M. and 12:00 midnight.

- (d) Night Tour a seven and one-half (7 1/2) hour tour ending between 12:01 A.M. and 8:00 A.M.
- (e) No more than two and one-half (2 1/2) hours shall be scheduled on duty during either a day or evening tour without either a relief or meal period.
- (f) Thirty-seven and one-half (37 1/2) hours per week shall be the basis of the hours of work.
- (g) Saturday, Sunday, Holiday, Evening and Night tours shall have a one-half (1/2) hour meal period.
- 16.03 Calendar Work Week shall be a period of seven (7) consecutive days commencing 12:01 A.M. on Sunday and ending at midnight on the following Saturday.
- 16.04 Day tours of duty shall be from 8:00 A.M. to 4:30 P.M. but other hours may be assigned to meet service requirements except that a Day tour of duty shall not be assigned to commence before 7:00 A.M. nor terminate later than 6:00 P.M.
- 16.05 All tours shall be rotated as equally as possible amongst all affected employees.
- 16.06 Change of Tour Payment Less than Seven (7) Days' Notice

All tour schedules **for regular and casual employees** shall be posted five (5) weeks in advance of the day they are to become effective.

Occasional employees are call in employees, however where the requirement is known, they may be scheduled according to the notice period in effect for their area. The Company shall provide occasional employees the opportunity to identify their availability (i.e. number of days available per week and specific days or times not available) one (1) week prior to the schedule being posted. Occasional employees who are scheduled or called in shall not have their scheduled or call in tours cancelled. Once an occasional employee is scheduled, or accepts a call in, they are responsible to work the tour or arrange a trade.

When an employee's tour schedule is changed without seven (7) days' notice, the employee shall be paid their basic rate of pay and in addition, a change of tour payment calculated as follows: one (1) times their basic rate of pay on the first tour of the change only. Normal pay treatment will apply on the second and subsequent days of the tour schedule.

Employees away from work shall be notified by the Supervisor in charge, of changes posted seven (7) full days in advance. A change of tour payment is not applicable when an employee is required to work overtime.

- **16.07** An employee who has completed their tour of duty and is called back within a twenty-four (24) hour period from the start of their last tour, (e.g. 8:00 A.M. to 8:00 A.M. or 4:00 P.M. to 4:00 P.M.), to cover a vacant tour for any reason shall be paid overtime in accordance with the overtime regulations.
- **16.08** An employee may change their tour with any other qualified employee at the discretion of their Supervisor. The other qualified employee shall be responsible for the tour coverage at no extra cost to the Company.
- **16.09** Employee's days off shall be consecutive.
- **16.10** (a) Any request for deviation from normal basic hours of work except for regularly scheduled tour workers shall be subject to fourteen (14) days notice in writing to the Union and the employees. Such notice shall contain the reasons for the deviation, the number of, and names of employees, and departments affected.

In the event the parties, through joint consultation which may be done by telephone, letter or where necessary a meeting, fail to mutually agree to the deviation within fourteen (14) days of the notice, the Company may proceed as they deem necessary and the Union may refer the matter to Step 4 of the Grievance Procedure.

(b) For specific work items, temporary deviations of no more than ten (10) consecutive working days, which can only be broken by days off in a sixty (60) day period, will be allowed without the requirement of written notice and mutual agreement of the parties. Such deviation will require at least seven (7) days written notice to the employee(s). If seven (7) days written notice is not given, then in addition to their basic rate of pay, an employee will be paid for the first day only of the temporary deviation: one (1) times their basic hourly rate for each hour worked. Such payments are not applicable when an employee is scheduled or called out on an overtime basis.

- **16.11** No Occasional or Regular Part-Time employee shall be paid for less than four (4) hours in any one (1) day and this short tour can only be broken by a meal period of one-half (1/2) hour maximum duration.
- 16.12 Break Period

A paid break period, not to exceed fifteen (15) minutes, will be given during each session subject to emergency service requirements.

16.13 Meal Period

Meal period shall not exceed one (1) hour. Employee requests for one-half (1/2) hour meal periods shall not be unreasonably denied.

ARTICLE 17 - HOURS OF WORK AND TOURS OF DUTY (Operator Services)

- 17.01 Tour Employee means an employee who is regularly posted on a tour schedule covering day, afternoon, evening or night tours.
- 17.02 (a) Tour of Duty means the scheduled period of work for an employee on any working day.
 - (b) Day Tour Seven (7) hours commencing not earlier than 6:00 A.M. and terminating not later than 6:00 P.M. and the following scheduling provisions shall apply:
 - (1) A session may start and end on the hour, half-hour or the quarter hour.
 - (2) Saturday, Sunday and Holiday tours will have one half (1/2) hour lunch breaks.
 - (c) Afternoon Tour Seven (7) hours, terminating between 6:00 P.M. and 9:00 P.M. and scheduling provisions apply as in (b) above.
 - (d) Evening Tour Seven and one-half (7 1/2) continuous hours, terminating after 9:00 P.M., inclusive of a one-half (1/2) hour unpaid lunch break.
 - (e) Night Tour Seven (7) hours, 12:00 midnight to 7:30 A.M.
 - (f) No more than two (2) hours shall be scheduled on duty without relief or meal period except as provided in Article 40.04.

- 17.03 Split Tours Seven (7) hours commencing not earlier than 6:00 A.M. and terminating not later than 9:00 P.M.
 - (a) There will not be less than two (2) hours nor more than a five (5) hour period between the two portions of a split tour.
 - (b) Split tours shall not be scheduled on Sundays or Holidays.
- 17.04 No Occasional or Regular Part-Time employee shall be paid for less than four (4) hours in any one (1) day and this short tour can only be broken by a meal period of one-half (1/2) hour maximum duration.
- 17.05 Relief of fifteen (15) minutes shall be given on each session as close to the middle of the session as possible.
- 17.06 An employee commencing a session who is unable to continue on duty on account of sickness shall be assumed to have worked the session and will be paid accordingly.
- 17.07 Basic Hours of Work means the hours of work per day and per week established in this Agreement.
- 17.08 For the purposes of this Agreement, "calendar work week" shall be a period of seven (7) consecutive days commencing 12:01 A.M. on Sunday and ending midnight on the following Saturday.
- 17.09 Thirty-five (35) hours per week shall be the basis of the hours of work.
- 17.10 (a) An employee who is required to work in excess of seven (7) basic hours of work per day shall be paid overtime in accordance with the overtime regulations.
 - (b) An employee who is required to work in excess of thirty-five (35) basic hours of work per week shall be paid overtime in accordance with the overtime regulations.

- 17.11 (a) All tour schedules shall be posted on Thursday at least fourteen (14) days in advance of the date they are to become effective. When an employee's tour schedule is changed without seven (7) days notice, the employee shall be paid two (2) times **their** basic rate of pay on the first tour of the change only. Normal pay treatment will apply on the second and subsequent days of the tour schedule. Should the first tour be a Sunday, Holiday, Christmas or New Year's Eve, the foregoing payment will be additional to the premium and/or holiday special payment. A change of tour payment is not applicable when an employee returns to their originally scheduled tour after a previous tour change that resulted in a change of tour payment. Neither will such payment be applicable when an employee is required to work overtime, or when the change of tour involves no change in the starting and/or ending of the tour, but simply a change of lunch or break periods within the tour.
 - (b) Christmas week and New Year's week schedules shall be posted four
 (4) weeks in advance of December 25th.

(c) Occasional employees who are scheduled or called in shall not have their scheduled or call in tours cancelled.

- 17.12 Employees on days off or holidays shall be notified by the Supervisor in charge of any tour changes made in the posted schedule. Posted tours should not be changed by Management unless by mutual consent of the employee concerned and payment as in Clause 17.11 (a) will apply.
- 17.13 All tour schedules shall be so arranged as to allow for regular indiscriminate rotation of tours within their respective groups.

ARTICLE 18 - FAMILY DAYS (Clerical)

- 18.01 Eligibility and Pay Entitlement
 - (a) Every January 1, all regular full-time and regular part-time clerical employees will be credited with three (3) Family Days.
 - (b) Regular employees commencing clerical employment will be credited with Family Days as follows:
 - (1) Between January 1 and April 30 three (3) days.
 - (2) Between May 1 and August 31 two (2) days.
 - (3) Between September 1 and December 31 one (1) day.
 - (c) Regular full-time clerical employees will receive their basic rate of pay for each Family Day.

- (d) Regular part-time clerical employees will receive their basic hourly rate for five (5) hours for each Family Day.
- (e) All Family Days are to be taken prior to December 31 of the year in which they were credited.
- (f) Family Days may be taken in half day increments.
- 18.02 (a) An employee may request their Family Day(s) anytime throughout the year. Arrangements for Family Days will be worked out between the manager and the employee. All reasonable efforts will be made to accommodate employee requests for Family Days.
 - (b) In the event management was unable to accommodate requests for Family Days, the day(s) not used will be paid out to the employee at year end. If the employee has not requested to use their Family Days they will not be carried over nor paid out.

ARTICLE 19 - OVERTIME - APPLICATION OF OVERTIME REQUIREMENTS

Where service affecting conditions or operational needs of an unforeseeable or unpreventable nature occur **the Company** expects that employees will co-operate and respond to callouts and operational needs that may involve additional hours above the normal.

Management shall make every reasonable effort to canvass for volunteers before assigning overtime.

If it becomes necessary to assign overtime, an employee's reasonable excuse for not working overtime, that may arise out of unforeseen personal difficulties shall be given every reasonable consideration - judged by circumstances and availability of other qualified staff or alternate methods of meeting the overtime requirement.

19.01 Overtime will be paid at the rate of two (2) times the basic hourly rate.

19.02 Callout Overtime

A minimum of two (2) hours pay at the applicable overtime rate shall be allowed to any employee who is called out for immediate reporting to the job and shall include travelling time to and from home. Any subsequent call in this two (2) hour period will be considered part of the first call-out. This Clause does not apply to any employee who is held over to work overtime at the end of their regular tour, nor does it apply to any employee who is called and works continuously into the beginning of their regular tour. In the latter case, overtime at the applicable rate will be paid from the time of callout to the beginning of the tour.

- 19.03 An employee required to work overtime for a complete tour prior to or after their regular tour or on their day off, shall be paid two (2) times the employee's basic rate of pay, regardless of the length of the tour.
- 19.04 An employee who works two (2) or more hours of overtime preceding or following their regular tour shall be entitled to a paid fifteen (15) minute break between their overtime hours and regular tour. Where possible, the break shall be given during the regular hours.

Employees working overtime shall be entitled to breaks in accordance with 16.02 (e) for Clerical employees, and 17.02 (f) for Operator Services.

- **19.05** Overtime will be paid for only when approved by the authorizing Supervisor in charge.
- 19.06 Rest Period

The provisions of this Clause shall not apply in the case of a callout overtime work period of less than four (4) hours duration, nor shall it apply when an employee is required to work on their days off.

An employee who is required to work overtime for a continuous period of four (4) hours or more shall be granted a rest period of eight (8) consecutive hours with no loss in their basic rate of pay in accordance with the following conditions:

- (a) An employee shall return to work at the end of their eight (8) hour rest period, when the rest period ends prior to the beginning of the last session of their scheduled tour of duty.
- (b) When the eight (8) hour rest period extends into the last session of their tour, the employee shall not be required to report for work.

- (c) When due to the demands of service, an employee is required to work during the complete rest period or any part of the rest period or last session of their tour stipulated in this Clause or following the end of such tour, they shall be paid at the rate of two (2) times their basic hourly rate for the hours so worked in lieu of the rest period payment for such hours.
- (d) When a sustained overtime period of four (4) hours or more extends into an employee's scheduled tour of duty they shall be paid at the rate of two (2) times their basic hourly rate for such hours worked in their scheduled tour and shall receive a rest period and payment at their basic hourly rate for the balance of their scheduled tour of duty not worked.
- **19.07** Vacation Overtime (V.O. Time)

V.O. time permits a regular full-time or regular part-time employee working overtime to elect to be paid for such overtime in accordance with the following:

- (1) such overtime to be paid for at the overtime rate; or,
- (2) such overtime to be paid for at straight time rates for the overtime worked and the employee granted time off equivalent to and in lieu of the remainder of their overtime entitlement; or,
- (3) all overtime to be "banked" and taken as paid time off at a later date.

V.O. time permits an occasional employee working overtime to elect to be paid for such overtime in accordance with the following:

- (1) such overtime to be paid for at the overtime rate; or,
- (2) such overtime to be paid for at straight time rates for the overtime worked and the equivalent hours banked to be taken as a pay out at a later date; or,
- (3) all overtime to be "banked" and taken as a pay out at a later date.
- (a) Election of method of payment shall be made by the employee prior to the overtime being reported to the Accounting Department.

- (b) Both the overtime and the premium portion of callout overtime payment as well as all or one-half (1/2) of the premium portion of any Company holiday may be banked. An employee may choose to bank the premium portion of the Sunday premium.
- (c) All V.O. time compensation either for time off or payments in lieu thereof, shall be based on the employee's V.O. accrued earnings.
- (d) An employee may bank and maintain unlimited accumulated time off.
- (e) All accrued V.O. earnings in excess of forty (40) hours shall be paid out at the employee's basic hourly rate in the pay period containing April 30th.
- (f) Scheduling of V.O. time off:
 - (1) A request for V.O. time off shall be made at least fourteen (14) days in advance wherever possible. A reply shall be conveyed to the employee as early as possible. Lack of fourteen (14) days notice shall not preclude the granting of V.O. time off at the employee's request and agreement by the Supervisor. Where the demands of service necessitate, the Company shall have the right to defer a request for V.O. time off.
 - (2) The scheduling of annual vacations as per Article 21 shall take precedence over V.O. time off.
 - (3) It shall not be required to use all or any annual vacation prior to requesting V.O. time off.
 - (4) An employee may request a V.O. time payout at any time. **The minimum payout shall be 0.25 hours.**
- **19.08** Part-time employees working less than the normal basic hours per day of fulltime employees and who are required to work longer than their regular working day, shall be paid at their basic hourly rate for the hours so worked up to the normal hours for full-time employees, after which the overtime provisions of Clause 19.01 shall apply.

ARTICLE 20 - HOLIDAYS AND HOLIDAY PAY

20.01 (a) Company Holidays shall be as follows:

New Year's Day Good Friday Easter Monday Victoria Day Canada Day Labour Day Thanksgiving Day Remembrance Day Christmas Day Boxing Day Christmas Float Holiday

and any other holiday approved by the Company.

- (b) A Company Holiday which falls on any day Monday to Friday inclusive, shall be observed on the day on which it falls. When any Company Holiday falls on a Saturday or Sunday, the following Monday or preceding Friday shall be declared by the Company as a lieu day.
- (c) When Christmas Day is observed on a Monday, Boxing Day will be observed on the Tuesday following.

Clerical:

- (d) For regular tour workers only, Christmas Day will be observed on December 25.
- (e) The Christmas Float Holiday shall be observed so as to give five (5) consecutive days off including the weekend as follows:
 - (1) On December 24 when Christmas Day falls on Tuesday, Thursday, Friday or Saturday.
 - (2) On December 27 when Christmas Day falls on Monday or Wednesday.
 - (3) On December 28 when Christmas Day falls on Sunday.

Operator Services

(d) Tour workers will observe the actual day of the Company Holiday rather than the lieu day where a day is declared in lieu of a Company Holiday. This is not applicable to an employee declaring an alternate day in lieu of Easter Monday.

- (e) The Christmas Float Holiday shall be observed on the following days and employees scheduled in accordance with Operator Services scheduling procedures:
 - (1) On December 24 when Christmas Day falls on Tuesday, Thursday, Friday or Saturday.
 - (2) On December 27 when Christmas Day falls on Sunday, Monday or Wednesday.
- 20.02 Civic Holidays
 - (a) Any full or half-day Civic Holiday proclaimed under the Municipal Government Act by the local authorities of any city, town or village will be given immediate consideration by the Company upon receipt of the official proclamation. Proclamations should be forwarded to the Vice-President, Human Resources for consideration.
 - (b) Civic Holidays for travelling employees will be those declared at their headquarters.
 - (c) Employees scheduled to work on a Civic Holiday shall, at the employee's discretion, be granted equivalent time off with pay at a later date or be paid double their basic rate of pay for that day. Time off shall be arranged at a time suitable to the employee and their supervisor. Overtime shall not be paid.
 - (d) Should the Civic Holiday be the first Monday of August and it occurs during a vacation period, one (1) additional day will be allowed. The foregoing will also apply in those locations which proclaim an alternate day in lieu of the first Monday of August.
 - (e) Employees not scheduled to work on a Civic Holiday who are called out to work on the holiday will receive overtime payment in addition to their basic rate of pay.
 - (f) An employee whose normal day off falls on a Company approved Civic Holiday, will receive equivalent time off later.

20.03 Easter Monday

- (a) Employees shall be entitled to choose an alternate day in lieu of Easter Monday which may be used to observe a religious holiday.
- (b) An employee electing to take a day in lieu of Easter Monday must declare their intention to do so not less than five (5) weeks prior to Easter Monday, otherwise Easter Monday will remain as the employee's holiday. A specific date need not be requested at the time the employee declares their intention to take a day in lieu.
- (c) The day in lieu of Easter Monday must be scheduled between January 2nd and December 20th inclusive in any year. Requests to have the lieu day scheduled between December 21 and January 1 will be considered and may be granted subject to the demands of service.
- (d) A request for the day in lieu of Easter Monday that is submitted not less than five (5) weeks in advance of the date requested shall be granted subject to 20.03 (c). Requests submitted less than five (5) weeks in advance will be considered and may be granted subject to demands of service.
- (e) The day in lieu of Easter Monday must be taken within the calendar year and cannot be carried forward to the next year. The day will not be paid out except in the case of layoff or dismissal of the employee.
- (f) An employee who has elected to take another day in lieu of Easter Monday and who is scheduled to work on Easter Monday may only change the Easter Monday tour with another qualified employee who has also elected to take another day in lieu of Easter Monday and at the discretion of the Supervisor. The other qualified employee shall be responsible for the Easter Monday tour at no extra cost to the Company.
- 20.04 A Company Holiday shall have the effect of reducing the calendar work week by one (1) day on the occurrence of one (1) holiday or two (2) days on the occurrence of two (2) holidays, etc.
- 20.05 Holiday Pay
 - (a) An employee scheduled and who works a tour of duty on a Company Holiday shall be paid two (2) times their basic rate of pay. This payment is in addition to any other that the employee may be entitled to receive under 20.05 (b), (c), (d), (e), ((f) applies to Operator Services only).

- (b) There shall be no deduction in the pay of a regular full-time employee for a holiday not worked except in the case of an employee on furlough or leave without pay which has exceeded thirty (30) days. When a holiday occurs at the end of a furlough period or leave without pay, the employee must notify their immediate supervisor of their availability for work on that day to be paid for the holiday.
- (c) Regular Part-Time employees shall be paid holiday pay in accordance with Clause 30.02.
- (d) A casual employee who has received wages on at least fifteen (15) of the thirty (30) days preceding the holiday is entitled to their basic rate of pay for the holiday.

A casual employee who has received wages on less than fifteen (15) days in the thirty (30) days preceding the holiday is entitled to one-twentieth (1/20) of their wages earned in the thirty (30) days preceding the holiday.

(e) An occasional employee is entitled to the average of their regular earnings in the thirty (30) days preceding the holiday.

Operator Services:

- (f) An employee taken ill while at work during a holiday session or tour, shall be assumed as having worked the session or tour as the case may be, and shall be paid the holiday payment per Clause 20.05(a).
- (g) A holiday night tour shall start at 12:01 A.M. on the day of the holiday.
- 20.06 An employee called in to work overtime shall be paid in accordance with the overtime regulations.

ARTICLE 21 - ANNUAL VACATIONS

- 21.01 **For purposes of vacation selection,** the vacation year will extend from May 1st of one year until April 30th of the following year.
- 21.02 Vacation Credits
 - (a) Regular employees shall accrue one twenty-sixth (1/26th) of their annual vacation entitlement in each pay period. Employees accruing increased vacation entitlement in a milestone year will be able to select vacation based on their new vacation entitlement during the current May 1st April 30th vacation year (which may fall before the actual milestone service date when increased vacation entitlement occurs).
 - (b) Regular employees absent for maternity leave or child care leave (for the purpose of adoption) shall continue to accrue vacation for the first twenty (20) weeks of such absence. Accrual of vacation credits will then stop until the employee returns to work.
 - (c) Regular employees absent on leave of absence (i.e. Education Leave) with partial pay shall accrue vacation at a prorated rate (e.g. twenty-five per cent (25%) of normal credits when receiving twenty-five per cent (25%) regular pay).
 - (d) In accruing vacation, no reduction shall be made for approved furlough, leaves of absence or layoffs when the total time off during the twelve (12) months previous to May 1 is two (2) months or less.
 - (e) Regular employees absent on leave of absence without pay, other than those above, shall cease accruing vacation credits the first working day of the leave of absence. Accrual of vacation credits shall start again the first day the employee returns to work.
 - (f) Regular employees who have their vacation credits reduced shall have the option of taking approved furlough for all or part of the credits not earned. (e.g. Employee entitled to four (4) weeks of vacation but only accrues two (2) weeks of vacation, the employee is entitled to schedule up to two (2) weeks of approved furlough in addition to their two (2) weeks of vacation.) Scheduling of such "vacation furlough" shall be done in accordance with the vacation scheduling clauses.

21.03 An employee who has completed the required continuous years of service shall be entitled to vacation as **shown below.**

Service Completed	Vacation Entitlement
Less than one (1) year	One and one-quarter (1.25) days for each full month of service completed
One (1) to eight (8) years	three (3) weeks
Nine (9) to fifteen (15) years	four (4) weeks
Sixteen (16) to twenty-four (24) years	five (5) weeks
Twenty-five (25) or more years	six (6) weeks

Upon hire, an employee shall accrue vacation on a biweekly pay period basis and shall move to the next tier of vacation accrual one (1) year prior to their anniversary date. The vacation accrual shall be incremented as shown in the following charts. In the conversion to hourly accrual no employee shall have their entitlement adversely affected for any reason except as provided for elsewhere in this Collective Agreement. Any system adjustments required shall be made manually.

Clerical	
Service Completed	Vacation Accrual
Less than one (1) year	accrue 4.33 hours biweekly
One (1) to eight (8) years	accrue 4.33 hours biweekly
Nine (9) to fifteen (15) years	accrue 5.77 hours biweekly
Sixteen (16) to twenty-four	accrue 7.22 hours biweekly
(24) years	
Twenty-five (25) or more years	accrue 8.66 hours biweekly

<u>Operator Services</u> <u>Service Completed</u>	Vacation Accrual
Less than one (1) year	accrue 4.04 hours biweekly
One (1) to eight (8) years	accrue 4.04 hours biweekly
Nine (9) to fifteen (15) years	accrue 5.39 hours biweekly
Sixteen (16) to twenty-four (24) years	accrue 6.74 hours biweekly
Twenty-five (25) or more years	accrue 8.08 hours biweekly

- 21.04 Vacation Pay, **Regular Employees:**
 - (1) Regular employees shall be entitled to paid vacation at the employee's regular rate of pay at the time vacation is taken.
 - (2) When four per cent (4%), or six per cent (6%) after six (6) years of service, of the employee's gross earnings in the preceding vacation year (beginning with the first pay period of the year and ending in the last one) exceeds the total amount of their vacation pay at their regular rate of pay at the time of taking their vacation, any excess shall be paid to the employee in the pay period which includes January 15th.
 - (3) Gross earnings shall include regular earnings, differentials, commissions, overtime pay, holiday pay and vacation pay.
- 21.05 Vacation Wage Advance:

An employee may request their vacation pay in advance for the regular payroll period(s) which terminate during their vacation. Such requests shall be made verbally to the Manager or an appointed representative not less than fourteen (14) days prior to the date the advance is required by the employee.

- 21.06 Vacation Scheduling:
 - (a) Employees shall have preference to dates for vacation in order of their seniority.
 - (b) The Employer shall post or circulate a vacation planner on or before February 15 of each year. Regular employees shall select their choice for vacation in order of seniority by March 15. After this date vacation will be selected on an as requested basis and granted subject to the demands of service. Management will reply to the request within five (5) working days.
 - (c) Employees shall be granted their posted vacation except in cases of extraordinary local or general disturbances caused by storms, flood, fires or accidents. If the employee's vacation is cancelled for any of the above reasons, the employee shall be compensated for the loss of any deposit or prepayment upon presentation of proof of loss, and the employer will attempt to reschedule the vacation on the employee's preferred alternate dates.
 - (d) Employees shall not be scheduled any Saturday and/or Sunday tours at the beginning and/or end of their vacation period.
 - (e) If an employee who has approved vacation scheduled, transfers, the Employer shall honour this scheduled vacation.
 - (f) Employees' vacation schedules shall be arranged exclusive of Management's vacation schedules.

Clerical

(g) Vacations will be arranged in accordance with the requirements of service by the Supervisor in charge. A minimum of fifteen percent (15%) of regular employees in each classification in a functional work group / office at each location will be allotted vacation at any given time. Where there are fewer than ten (10) employees in the group / office, there shall be a minimum of ten percent (10%) of regular employees allotted vacation at any given time. This does not preclude allotting more than the ten percent (10%) or fifteen percent (15%) minimums if requirements of service allow.

Where the **fifteen percent (15%)** minimum vacation allotment has not been filled by April 1, the remaining percentage shall be available for other time off requests by Regular employees on an as requested basis.

Operator Services

(g) Vacation will be arranged by Management in charge in accordance with the requirements of service. A minimum of ten percent (10%) of regular employees in each office will be allotted vacation at any given time. This does not preclude allotting more than the ten percent (10%) **minimum** if requirements of service allow.

Where the ten percent (10%) minimum vacation allotment has not been filled by April 1, the remaining percentage shall be available for other time off requests by Regular employees on an as requested basis.

Clerical

- (h) Vacation periods may consist of any number of days and may commence on any day of the week providing this can be accommodated. Single day selection shall be limited to five (5) working days at the time of scheduling. However, single day selection will not be affected where the employee is unable to schedule a full week due to a lack of vacation days remaining on the vacation schedule.
- e.g. Monday, Tuesday, Wednesday available in a week and employee schedules all three (3) days (single day selection not reduced).

Monday, Tuesday, Wednesday available in a week and employee schedules less than three (3) days available (single day selection reduced).

Operator Services

- (h) With the approval of the immediate supervisor, vacations may be taken:
 - (i) in one (1) continuous period or,
 - (ii) in separate or combined periods of not less than one (1) week (five (5) working days)
- (i) All vacations shall commence at the beginning of the calendar week unless unavoidable due to the demands of the service. Upon an employee's request, vacations may start on any day of the calendar week subject to supervisory discretion.
- **21.07** (a) Vacation entitlement must be taken by an employee before expiration of the vacation year in which the vacation falls due.

(b) If an employee requests or due to an unforeseen situation, the employee is unable to take all or part of their annual vacation in the vacation year in which it becomes due, the employee may, with the approval of the Employer, carry over their unused vacation in accordance with the following:

Completed Years of <u>Continuous Service</u> one (1) to six (6) years	<u>Vacation Carry Over</u> one (1) week
seven (7) and eight (8) years	no carry over
nine (9) to fifteen (15) years	one (1) week
sixteen (16) or more years	two (2) weeks

See **Example** for vacation carry over calculations.

- (c) Vacation which has been carried over may only be scheduled after March 15 in accordance with Clause 21.06 (b) and (g).
- **21.08** If a vacation period comes open for any reason, the next employee on the seniority list shall be given the opportunity to utilize that period.
- **21.09** An employee working away from their headquarters will be granted travelling time once a year at the start and end of their annual vacation. If an employee chooses to split their vacation, subject to the provisions of clause 21.06, the employee shall be allowed travelling time to and from their headquarters twice during any vacation year with transportation supplied.
- **21.10** When a holiday (Company or Civic) occurs during an employee's vacation period, the employee will be allowed one (1) additional day of vacation for each holiday (Company or Civic) which may be taken at a time suitable to the employee and their Supervisor.
- **21.11** An employee who is unable to use their vacation entitlement before the end of the vacation year because of illness will have their vacation entitlement and vacation pay carried over to the following vacation year. Under extenuating circumstances and if all sick benefits have been exhausted, the employee may have vacation entitlement and pay advanced to them with the authorization of the Employer.

EXAMPLE - VACATION CARRY OVER CALCULATIONS

Pursuant to Article 21 employees may, with management approval, carry over vacation to the following year. The amount of carry over is calculated as follows.

The Canada Labour Code stipulates that employees with less than seventy two (72) months of employment MUST take ten (10) days vacation each year, and employees with more than seventy two (72) months of employment MUST take fifteen (15) days of vacation.

TELUS employees with one (1) to six (6) years of service are eligible for three (3) weeks (15 days) of vacation and the Code states the minimum to be taken is ten (10) days.

The amount of carry over is calculated as follows:

vacation entitlements - Code minimum = carry over

in this example: 15 days - 10 days = 5 days. Therefore the employee may carry over 5 days vacation.

An employee with seven and eight years of service: vacation days = 15, Code minimum = 15 15 - 15 = 0 days. Therefore employees with seven or eight years of service are not eligible to carry over any vacation days.

An employee with nine to fifteen years of service: vacation days = 20, Code minimum = 15 20 - 15 = 5 days. Therefore the employee may carry over 5 days of vacation.

An employee with sixteen to twenty-four years of service: vacation days = 25, Code minimum = 15 25 - 15 = 10 days. Therefore the employee may carry over 10 days of vacation.

ARTICLE 22 - PAY IN LIEU OF VACATION

22.01 Casual and Occasional employees shall be entitled to vacation pay on each pay period as listed below:

Service Completed	Vacation Pay
less than six years	4% of gross earnings
six years or more	6% of gross earnings

Vacation Time Off

- (a) Casual and Occasional employees will be entitled to time off for vacation in accordance with **21.03**.
- 22.02 Regular employees whose employment ceases for whatever reason, shall be paid in accordance with 21.04 for all earned and outstanding vacation entitlement up to and including the last day worked. An adjustment shall be made to the final pay of an employee who has taken vacation entitlement in excess of that to which they were actually entitled at the date of termination of employment.

ARTICLE 23 - MATERNITY LEAVE

- 23.01 An employee who has completed six (6) months of service is entitled to a maximum of twenty (20) weeks of maternity leave without pay.
- 23.02 An employee who intends to take maternity leave shall notify the Employer in writing at least four (4) weeks in advance of the commencement date and length of such leave. At that time, the employee will provide a doctor's certificate verifying the pregnancy.
- 23.03 An employee who intends to change the length of maternity leave shall give not less than four (4) weeks notice in writing of any change in the length of leave intended to be taken.
- 23.04 The period of maternity leave shall not exceed twenty (20) weeks in length and may commence not earlier than fourteen (14) weeks prior to the estimated date of confinement and shall end not later than twenty (20) weeks following the actual day of confinement.

- 23.05 The Employer shall not require an employee to take a leave of absence from employment unless the employee is unable to perform an essential function of their job and there is no appropriate alternate job available. The employee shall only be required to take the leave of absence for such time as she is unable to perform that essential function.
- 23.06 An employee returning from maternity leave shall be reinstated in the position that they occupied when the maternity leave commenced. Where for any valid reason the Employer can not reinstate the employee in their former position, the Employer shall reinstate the employee in a comparable position with the same wages and benefits and in the same headquarters area.
- 23.07 Alberta Health Care Insurance Plan, Supplementary Health, Vision and Dental Care Plans, Group Insurance Contributions
 - (a) An employee who has been granted maternity leave, shall authorize the Employer to deduct from their final pay, the necessary funds required to cover the employee's portion of contributions to Alberta Health Care Insurance Plan, Supplementary Health, Vision and Dental Care Plans and Group Insurance for the whole period of maternity leave plus one (1) month's deduction.
 - (b) The Employer will continue to pay the Employer's portion of contributions to Alberta Health Care Insurance Plan, Supplementary Health, Vision and Dental Care Plans and Group Insurance for the whole period of maternity leave.
- 23.08 Suspension of Increases

A wage increase falling due during the period of maternity leave will be suspended during the period of leave with credit for time worked being applied in the establishment of a new scheduled date, upon return to work.

Fifteen (15) days or less in a month will not be counted as any portion of a month and more than fifteen (15) days will be counted as a full month for purposes of increases only.

- 23.09 (a) An employee who has been granted maternity leave, shall authorize the Employer to deduct from their final pay prior to commencing the leave, the necessary funds required to cover one (1) month's Union dues.
 - (b) When the employee returns to work, the Employer will resume deduction of Union dues.

(c) Union dues voluntarily contributed for the entire period of leave will be forwarded by the Employer to the Union.

ARTICLE 24 - CHILD CARE LEAVE

- 24.01 An employee who has completed six (6) months of service and who adopts a child or has care and custody of a newborn child shall be entitled to leave without pay of up to twenty-four (24) weeks. This period, in the case of adoption, commences when the child comes into the employee's care. Otherwise the period may commence on the day the child is born, or the day the child comes into the employee's care, or on the expiration of the mother's maternity leave.
- 24.02 (a) An employee who intends to take a child care leave shall notify the Employer in writing at least four (4) weeks prior to commencement of such leave and include the length of leave intended to be taken. The Employer recognizes that in cases of adoption, it may not be possible to provide four (4) weeks prior notice.
 - (b) An employee who intends to change the length of the leave shall provide four(4) weeks notice in writing of any change to the length of the leave.
- 24.03 An employee returning from child care leave shall be reinstated in the position occupied when the child care leave commenced. Where for any valid reason the Employer cannot reinstate the employee in their former position, the Employer shall reinstate the employee in a comparable position with the same wages and benefits and in the same headquarters area.
- 24.04 The aggregate amount of unpaid child care leave that may be taken by two (2) employees in respect of the birth or adoption of any one (1) child shall not exceed twenty-four (24) weeks.

- 24.05 Alberta Health Care Insurance Plan, Supplementary Health, Vision and Dental Care Plans, Group Insurance Contributions
 - (a) An employee who has been granted child care leave, shall authorize the Employer to deduct from their final pay, the necessary funds required to cover the employee's portion of contributions to Alberta Health Care Insurance Plan, Supplementary Health, Vision and Dental Care Plans and Group Insurance for the whole period of child care leave plus one (1) month's deduction.
 - (b)The Employer will continue to pay the Employer's portion of contributions to Alberta Health Care Insurance Plan, Supplementary Health, Vision and Dental Care Plans and Group Insurance for the whole period of child care leave.
- 24.06 Suspension of Increases

A wage increase falling due during the period of child care leave will be suspended during the period of leave with credit for time worked being applied in the establishment of a new scheduled date, upon return to work.

Fifteen (15) days or less in a month will not be counted as any portion of a month and more than fifteen (15) days will be counted as a full month for purposes of increases only.

- 24.07 (a) An employee who has been granted a child care leave shall authorize the Employer to deduct from the employee's last pay prior to commencing the leave and forward to the Union one (1) month's dues.
 - (b) When the employee returns to work, the Employer will resume deduction of Union dues.
 - (d) Union dues voluntarily contributed for the entire period of leave will be forwarded by the Employer to the Union.

ARTICLE 25 - ADOPTION LEAVE

- 25.01 An employee who has completed six (6) months of service may request a leave of absence without pay or benefits (except as provided in 25.05) not to exceed twenty (20) weeks for the purpose of adopting a child. Such leave must be taken immediately following any leave taken pursuant to Article 24: Child Care Leave. Approval for adoption leave will not be unreasonably withheld.
- 25.02 Requests for adoption leave must be submitted in writing to the employee's Supervisor not less than four (4) weeks prior to the anticipated commencement date of the adoption leave.
- 25.03 An employee returning from adoption leave shall be reinstated in the position occupied when the adoption leave commenced. Where for any valid reason the Employer cannot reinstate the employee in their former position, the Employer shall reinstate the employee in a comparable position with the same wages and benefits and in the same headquarters area.
- 25.04 An employee who has been granted adoption leave, shall authorize the Employer to deduct from their final pay or pay to the Employer, the necessary funds required to cover the employee's portion of premiums to Alberta Health Care Insurance Plan, Supplementary Health, Vision and Dental Care Plans and Group Insurance for the entire period of adoption leave plus one (1) month.
- 25.05 The Employer will continue to pay the Employer's portion of premiums to Alberta Health Care Insurance Plan, Supplementary Health, Vision and Dental Care Plans and Group Insurance for the whole period of the adoption leave.
- 25.06 When the employee returns to work, the Employer will resume deduction of Union dues.
- 25.07 Union dues voluntarily contributed for the entire period of leave will be forwarded by the Employer to the Union.

ARTICLE 26 - SAFETY

- 26.01 Employees shall abide by all Company safety regulations and wear or use protective equipment provided by the Company.
- 26.02 The Company shall ensure the safety of employees while at work.
- 26.03 The Union fully supports safety initiatives and will endeavor to ensure its members follow any safety procedures. The Union may make recommendations

from time to time related to safety matters which will be considered by the Company.

- 26.04 Safety Training
 - (a) The Company shall provide health and safety training and education, including regular upgrade training, sufficient to enable each employee to work with minimum risk at all jobs they may be assigned.
 - (b) No employee shall be required to perform any job or use any tools or equipment for which they have not received proper instruction.
 - (c) All tools and equipment provided by the Company and used by employees shall be inspected in accordance with Company regulations.
- 26.05 Where an employee while at work has reasonable cause to believe that:
 - (a) the use or operation of a tool or equipment constitutes a danger to the employee or another employee; or,
 - (b) a condition exists in any place that constitutes a danger to the employee,

the employee may refuse to use or operate the tools or equipment or to work in that place.

- 26.06 An employee may not refuse to use or operate the tool or equipment or to work in a place if:
 - (a) the refusal puts the life, health or safety of another person directly in danger; or,
 - (b) the danger referred to in Clause 26.05 is inherent in the employee's work or is a normal condition of employment.
- 26.07 Safety Footwear

Employees required to work in designated foot protection areas will be eligible for reimbursement, upon proof of purchase, of CSA approved footwear to a maximum of one hundred dollars (\$100.00) per calendar year. Confirmation of the requirement for safety footwear is required from the supervisor prior to purchase.

26.08 Video Display Terminals

A pregnant employee who is required to work directly with Video Display Terminals may elect to take furlough by forwarding a written request to the Company together with a certificate from a qualified medical practitioner verifying that she is pregnant.

26.09 Safety representatives shall be allowed access to all work locations.

ARTICLE 27 - ACCIDENT COMPENSATION

- 27.01 All accidents sustained by an employee while on duty will be reported as soon as practicable to the employee's Supervisor in order that the Workers' Compensation Board may be notified.
- 27.02 When an employee suffers a disabling work injury for which they are entitled to compensation under the Workers' Compensation Act, basic wages will be paid from the first day after the work injury in accordance with the Accident Compensation Schedule. Thereafter direct payments will be made by the Workers' Compensation Board. An employee's sick leave will not be affected while on compensation.

Accident (Compensation Schedule
------------	-----------------------

During The Following Service Year	Applicable Accident Compensation Credits (Weeks at Full Pay)	Clerical <u>(Hours)</u>	Operator Services <u>(Hours)</u>
7-12 Months.	One (1)	37.5	35.0
2nd Year.	Two (2)	75.0	70.0
3,4, & 5 Years.	Four (4)	150.0	140.0
6-15 Years. (inclusive)	Twenty (20)	750.0	700.0
16-20 Years. (inclusive)	Twenty-six (26)	975.0	910.0
21-25 Years. (inclusive)	Thirty-nine (39)	1462.5	1365.0
Over 25 Years	Fifty-two (52)	1950.0	1820.0

- 27.03 A casual or occasional employee who suffers a disabling injury while working for the Employer which qualifies for Workers' Compensation will receive payment directly from the Workers' Compensation Board.
- 27.04 If an employee resumes duty before having exhausted their accident compensation benefits to which the employee is entitled and in a period less than **their regular hourly equivalent of** sixty-five (65) working days is again

absent due to a related or non-related accident, a continuing accident compensation claim will be presumed. Benefits will be resumed from the point at which the first portion of the claim ceased and continued until the end of the continuing claim or until the employee's rights have been exhausted.

- 27.05 After their regular hourly equivalent of sixty-five (65) working days, accident compensation credits will be re-established. For the purpose of accumulating their regular hourly equivalent of a sixty-five (65) working day period, compensation furlough will not be considered as lost time, but will extend the time required to re-establish credits by the amount of compensation furlough taken during this period.
- 27.06 Anniversary date passed while on accident compensation if the employee is entitled to increased credits according to regulations, the employee will be allowed, if required, the difference in excess of the credits in the previous service year without having to return to work and normal accident compensation regulations will apply when the employee returns to work.
- 27.07 Anniversary date passed between disabling work injury absences if the employee is entitled to increased credits according to the regulations, the employee will be allowed, if required, the difference in excess of the credits in the previous service year or will be allowed the new credits after a period of **their regular hourly equivalent of** sixty-five (65) working days between disabling work injury absences.
- 27.08 Anniversary date passed but not establishing increased credits the normal regulations will apply and **their regular hourly equivalent of** sixty-five (65) working days must elapse between disabling work injury absences to establish new accident compensation credits.

ARTICLE 28 - SICK LEAVE REGULATIONS

28.01 Clerical

If an employee is unable to report for work because of sickness, they will notify their Supervisor at the start of their shift or immediately following the start of their shift. Failure to report and in the absence of satisfactory reason(s), then disciplinary action pursuant to Article 15 Discipline, Resignation and Dismissal may result.

Operator Services

If an employee is unable to report for work because of sickness, they will notify the supervisor before the start of their tour. Failure to report and in the absence of satisfactory reason(s), then disciplinary action may result.

28.02 Sick leave credits shall be computed based on an employee's service date.

28.03 Sick Leave Schedule

Sick Leave will be allowed in accordance with the following table:

During The Following Service Year	Days @ Full Pay	Days @ 70% Pay
	N 111	
0 -3 Months	Nil	Nil (see Note 1)
4-6 Months	Nil	75 (see Note 2)
7-12 Months	5	70
2nd Year	10	65
3-5 Years	20	55
6-10 Years	65	65
11-15 Years	100	160
16-20 Years	130	130
21-25 Years	195	65
Over 25 Years	260	Nil

In the conversion and administration of an hourly reporting system, no employee shall have their entitlement adversely affected for any reason except as otherwise provided for in this Collective Agreement. The SAP time reporting system administers Sick Leave in hours in accordance with the following tables:

<u>Clerical</u>

During The			
Following	Hourly	Hourly	
Service	Equivalent	Equivalent	
Year	@ Full Pay	@ 70% Pay	
0-3 Months	Nil	Nil (see Note 1)	
4-6 Months	Nil	562.5 (see Note 2)	
7-12 Months	37.5	525.0	
2nd Year	75.0	487.5	
3-5 Years	150.0	412.5	
6-10 Years	487.5	487.5	
11-15 Years	750.0	1200.0	
16-20 Years	975.0	975.0	
21-25 Years	1462.5	487.5	
Over 25 Years	1950.0	Nil	
Operator Services			
During The			
Following	Hourly	Hourly	
Service	Equivalent	Equivalent	
Year	@ Full Pay	@ 70% Pay	
0-3 Months	Nil	Nil (see Note 1)	
4-6 Months	Nil	525.0 (see Note 2)	
7-12 Months	35.0	490.0	
2nd Year	70.0	455.0	
3-5 Years	140.0	385.0	
6-10 Years	455.0	455.0	
11-15 Years	700.0	1120.0	
16-20 Years	910.0	910.0	
21-25 Years	1365.0	455.0	
Over 25 Years	1820.0	Nil	

- Note 1: Employees with less than four (4) months service may apply for sickness benefits directly from **EIC**. Should any employee in this category require sickness benefits from the **EIC** and because of the duration of the illness, pass the anniversary point of the fourth month, the employee shall continue to receive the benefit from **EIC** until the benefit is exhausted or return to work is possible. Any subsequent illness will be administered under the **TELUS** schedule of benefits.
- Note 2: During the fourth to sixth month of service a waiting period of ten (10) consecutive working days of sickness absence must elapse before the initial benefit will be applied starting at the eleventh day of absence. Any subsequent illnesses will be paid for from the first day of such illness.
- Note 3: For the purpose of Clause 28.03 a full day's pay **or** seventy percent (70%) day's payment for all employees in receipt of commission shall be calculated at their basic rate of pay on the day before they become entitled to sick leave plus one-two hundred **and sixtieth (1/260)** of their total commission paid in the previous calendar year.

For an employee who has not worked a full calendar year, the employee's commission shall be manually calculated by going back up to one (1) full year from the day before they become entitled to sick leave. The sick leave allowed shall be calculated at their basic rate of pay plus one-two hundred **and sixtieth (1/260)** of this commission.

28.04 In accordance with the preceding schedule an employee who is absent because of illness shall be considered to be on sick leave from the first day providing they notify their Supervisor in compliance with Clause 28.01.

In doubtful circumstances, the Supervisor may require a physician's certificate and shall advise the employee of this requirement prior to their return to work. Upon return to work the employee has the option of submitting the certificate to their Supervisor or directly to **the Company's** Occupational Health and Wellness **Advisor**.

When a Supervisor requires that an employee obtain any physician's documentation, the employee shall be reimbursed for the portion of the physician's fee not covered by the Provincial Health Care Plan for obtaining such documentation. A receipt will be required.

28.05 When an employee has advised their Supervisor that they are unable to continue work because of illness, they shall continue to receive their basic rate of pay for the balance of the work day with no reduction of sick leave credits.

- 28.06 Notwithstanding the provisions of the foregoing clauses, where the absence is frequent, extended or seriously interferes with the effectiveness of a production unit, upon request, a physician's certificate must be produced. Likewise, a medical consultation may be required by a physician appointed by the Company.
- 28.07 Cases of quarantine will be dealt with individually, as occasion demands and may be paid for subject to the limiting conditions of the Sick Leave Regulations.
- 28.08 An employee who has totally exhausted their sick leave credits and after resuming duty for **their regular hourly equivalent of** twenty-two (22) working days is again ill with an unrelated illness, will be eligible to receive, after ten (10) consecutive working days of illness, fifteen (15) weeks' payment at the rate of **seventy per cent (70%)**.
- 28.09 An employee who has partially used their sick leave credits and after resuming duty for **their regular hourly equivalent of** twenty-two (22) working days is again ill with a new illness, will receive the balance of the original credits plus the number of weeks required to provide fifteen (15) weeks of benefit. The two (2) week waiting period is waived in these cases.
- 28.10 If an employee resumes duty before having exhausted the sick leave benefits to which they are entitled and in a period less than **their regular hourly equivalent of** sixty-five (65) working days is again ill, a continuing Sick Benefit Claim will be presumed. Benefits will be resumed from the point at which the first portion of the claim ceased and continued until the end of the continuing claim or until the employee's rights have been exhausted.
- 28.11 An employee who exhausts all sick leave credits to which they are entitled and in a period less than **their regular hourly equivalent of** sixty-five (65) working days is again ill due to the same illness, will not be eligible for further sick leave credits until the employee has been returned to work for **their regular hourly equivalent of** sixty-five (65) working days.
- 28.12 After **the regular hourly equivalent of** sixty-five (65) working days regular service, should the employee again be sick, a new sickness will be assumed and paid for in accordance with Clause 28.03.

28.13 (a) Anniversary Date Passed While On Sick Leave

If the employee is entitled to increased credits according to regulations, they shall be allowed, if required, the difference in excess of the credits in the previous service year without having to return to work and normal sick leave regulations shall apply when they return to work.

(b) Anniversary Date Passed Between Sicknesses

If the employee is entitled to increased credits according to regulations, they shall be allowed, if required, the difference in excess of the credits in the previous service year or shall be allowed the new credits after a period **consisting of their regular hourly equivalent** of sixty-five (65) working days between sicknesses in accordance with the Sick Leave Regulations.

(c) Anniversary Date Passed But Not Establishing Increased Credits

The normal regulations shall apply and **their regular hourly equivalent of** sixty-five (65) working days must elapse between sicknesses to establish new sick leave credits.

(d) Exhaustion Of Sick Leave Credits

If an employee has exhausted their sick leave credits and is absent on sick furlough during the sixty-five (65) working days needed to re-establish new credits, this sick furlough shall not be considered as lost time for the purpose of accumulating **their regular hourly equivalent of the** sixty-five (65) working day period, but shall extend the time required to re-establish credits by the amount of sick furlough taken during this period.

- 28.14 When an employee is absent under the Sick Leave Regulations, the said employee shall not be granted vacation, Holidays (Civic or Company) or wages in lieu thereof during the sick leave absence. Under extenuating circumstances and if all sick leave benefits are exhausted, the employee may use their vacation credits with the authorization of the Employer.
- 28.15 (a) Sick Leave Regulations shall apply if an employee becomes incapacitated by reason of sickness or accident during their vacation or for the first two (2) weeks of any authorized furlough.

Clerical

(b) Upon their return to work, the employee must provide to the Medical Department a physician's documentation of the illness and the name, address and the phone number of the doctor who provided treatment. The physician's documentation may be verified by the Employer's medical advisor if necessary.

Operator Services

- (b) In order to be eligible for the unused portion of the vacation, the employee must immediately notify their supervisor at the beginning and end of their illness. This employee may claim the direct cost of this notification. (i.e. telephone call, wire, etc.). Upon their return to work, the employee must provide to the manager or medical department a physician's documentation of the illness and the name, address and phone number of the doctor who provided treatment. The physician's documentation may be verified by the employer's medical advisor if necessary.
- 28.16 The Company may stop payment of these benefits if, in their opinion, disability is caused by willful negligence or misconduct of the employee.
- 28.17 The sick leave regulations in this Article shall be subject to amendment at any time by any legislation or regulation changes enacted and put into effect by the **Employment** Insurance Commission.

ARTICLE 29 - ALBERTA HEALTH CARE INSURANCE PLAN, SUPPLEMENTARY HEALTH, VISION, DENTAL CARE PLANS AND GROUP INSURANCE PLAN

29.01 Alberta Health Care Insurance Plan

For all regular employees covered by this Agreement who are enrolled or become enrolled in the Alberta Health Care Insurance Plan, the Company shall pay seventy-five percent (75%) of the contract premium. The employee shall pay the balance of the premium through payroll deduction.

29.02 Supplementary Health Plan

For all regular employees covered by this Agreement who are enrolled or become enrolled in the Supplementary Health Plan, the Company shall pay seventy-five percent (75%) of the contract premium. The employee shall pay the balance of the premium through payroll deduction.

29.03 Dental Plan

For all regular employees covered by this Agreement who are enrolled or become enrolled in the Dental Plan, the Company shall pay eighty -five percent (85%) of the Dental Contract premium and fifty percent (50%) of the Orthodontics Care premium. The employee shall pay the balance of the premiums through payroll deduction.

29.04 Vision Care Plan

For all regular employees covered by this Agreement who are enrolled or become enrolled in the Vision Care Plan, the Company shall pay seventy-five percent (75%) of the contract premium. The employee shall pay the balance of the premium through payroll deduction.

- 29.05 Group Insurance Plan
 - (a) The Company will pay the premium on the first four thousand dollars (\$4,000.00) of insurance for these employees as of June 30, 1973 who elected to remain on the basic coverage.
 - (b) For all regular employees who are enrolled or become enrolled in the basic Group Insurance, the Company shall pay fifty percent (50%) of the contract premium. The employee shall pay the balance of the premium through payroll deduction.
- 29.06 Employee Benefits All benefits currently enjoyed by employees covered by this Agreement shall not be reduced or withdrawn during the life of this Agreement.

ARTICLE 30 - REGULAR PART-TIME BENEFITS

30.01 Benefit Premiums

Benefit premiums will not be prorated.

- 30.02 Company Holidays
 - (a) An employee who works on a Company holiday shall be paid at two (2) times their basic hourly rate for scheduled hours worked on the holiday. This payment is in addition to that provided in either (b) or (c).
 - (b) An employee who has received wages on at least fifteen (15) of the thirty (30) days preceding the holiday is entitled to their regular earnings for the holiday.

- (c) An employee who has received wages on less than fifteen (15) days in thirty (30) days preceding the holiday is entitled to one-twentieth (1/20th) of their wages earned in the thirty (30) days preceding the holiday.
- 30.03 Paid Absences

Regular part-time employees are entitled to these benefits but only on scheduled work days and pay treatment can not exceed scheduled hours. These benefits include such things as; Serious Distress (DS), Funeral Leave (FL) and Bereavement Leave (BL).

30.04 Progressional Increases

Regular part-time service is not prorated.

30.05 Seniority

Regular part-time service is not prorated.

30.06 Seventy Five Percent (75%) Rule

Regular part-time employees can be scheduled, within the guidelines of this Agreement, up to seventy-five percent (75%) of the hours of a regular full-time employee in the payroll year (a fixed twelve (12) month period of time).

30.07 Sick Leave

Regular part-time employees are entitled to sick leave on scheduled days of work only. Pay treatment is not to exceed hours scheduled. Sick leave entitlements will be earned based on calendar years of service and credits will be equivalent to seventy-five percent (75%) of those available to regular full-time employees based on the regular hourly equivalent applicable to Clerical or Operator Services. Sick leave credits are drawn down based on hours scheduled and reestablished based on the employee receiving pay for sixty-five (65) days (exclusive of overtime and sick leave), regardless of the number of hours worked in the day.

In the conversion and administration of an hourly reporting system, no employee shall have their entitlement adversely affected for any reason except as otherwise provided for in this Collective Agreement.

The SAP time reporting system administers Sick Leave in hours in accordance with the following tables:

<u>Clerical</u>		
During The Following	Hourly	Hourly
Service	Equivalent	Equivalent
Year	@ Full Pay	@ 70% Pay
	le rui ray	@ 70% Fay
0-3 Months	Nil	Nil
4-6 Months	Nil	421.88
7-12 Months	28.13	393.75
2nd Year	56.25	365.63
3-5 Years	112.50	309.38
6-10 Years	365.63	365.63
11-15 Years	562.50	900.00
16-20 Years	731.25	731.25
21-25 Years	1096.88	365.63
Over 25 Years	1462.50	Nil
Operator Services		
During The		
Following	Hourly	Hourly
Service	Equivalent	Equivalent
Year	@ Full Pay	@ 70% Pay
0-3 Months	Nil	Nil
4-6 Months	Nil	393.75
7-12 Months	26.25	367.50
2nd Year	52.50	341.25
3-5 Years	105.00	288.75
6-10 Years	341.25	341.25
11-15 Years	525.00	840.00
16-20 Years	682.50	682.50
21-25 Years	1023.75	341.25
Over 25 Years	1365.00	Nil
Two (2) Pay Periods		

30.08 Two (2) Pay Periods

The two (2) pay periods (four (4) continuous weeks), referred to in the respective Clauses of this Agreement are fixed blocks of time (mutually exclusive). The first two (2) pay periods of any payroll year will constitute the first regular part-time block.

30.09 Vacations

Vacation entitlements will be based on years of service.

ARTICLE 31 - TECHNOLOGICAL CHANGE (Clerical)

- 31.01 If a decrease of work necessitating layoffs results from technological change, such as: new equipment, office upgrading or new techniques and not from adverse economic conditions or voluntary resignations, then the Employer shall advise the Union and use its best efforts to train all affected employees desiring such training, so the employee may be qualified to accept employment in other phases of the operation of the system.
- 31.02 Regular employees who have completed one (1) but less than five (5) years of service, who suffer loss of employment as a direct result of technological changes shall be paid a separation allowance based on employee service and basic rate of pay at the time of layoff.
- 31.03 Regular employees who have completed five (5) or more years of service, who suffer loss of employment as a direct result of technological changes shall be paid a separation allowance of two (2) weeks pay per year of service to a maximum of forty (40) weeks, based on employee service and basic rate of pay at the time of layoff.
- 31.04 If any regular employee who has been paid a separation allowance is subsequently re-engaged and again laid off, separation allowance in the instance of the second, or any subsequent layoff will be based on the date of last re-engagement.

ARTICLE 32 - TECHNOLOGICAL CHANGE (Operator Services)

- 32.01 If a decrease of work necessitating layoffs results from technological change, such as: new equipment, office upgrading or new techniques and not from adverse economic conditions or voluntary resignations, then the Employer shall advise the Union and use its best efforts to train all affected employees desiring such training, so the employee may be qualified to accept employment in other phases of the operation of the system.
- 32.02 Regular employees who have completed one (1) year but less than five (5) years of service, who suffer loss of employment as a direct result of technological changes shall be paid severance pay pursuant to the Canada Labour Code.
- 32.03 Regular employees who have completed five (5) or more years of service, who suffer loss of employment as a direct result of technological changes shall be paid a separation allowance of two (2) weeks pay per year of service to a maximum of forty (40) weeks, based on employee service and basic rate of pay at the time of layoff.
- 32.04 If any regular employee who has been paid a separation allowance is subsequently re-engaged and again laid off, separation allowance in the instance of the second, or any subsequent layoff will be based on the date of last re-engagement.

ARTICLE 33 - LAYOFF (Clerical)

- 33.01 In the event of a force reduction of regular employees, the Company will effect the layoff on a last in, first out basis.
- 33.02 Should the Company decide to reduce the number of employees as referred to in Clause 33.01, casual and occasional employees shall be laid off prior to regular employees.
- 33.03 Prior to the Company exercising their rights in accordance with Clause 33.01, the Company shall:
 - (a) Notify the Union in writing not less than sixteen (16) weeks prior to the reduction. The notice will include the number of employees affected, their job title and pay group, their seniority date, and the reasons for the proposed reduction; and,
 - (b) establish a joint committee with the Union to consider alternatives to and otherwise mitigate the effects of the proposed downsizing. The committee

shall be a maximum of six (6) people equally represented by the Company and the Union and will meet within two (2) weeks of the notice. Alternatives may include voluntary resignation incentives, targeted resignation incentives, or other appropriate alternatives such as retraining/reskilling of employees or redeployment to other jobs; and,

- (c) jointly with the Union consider an outplacement program to assist laid off employees; and,
- (d) verify employee seniority, job title and pay group of those affected prior to the layoff.
- 33.04 Not less than ten (10) weeks prior to the layoff, the Company will:
 - (a) Send a notice to each affected employee containing:
 - (1) employee entitlements at time of layoff (e.g. UI, etc.),
 - (2) options being considered by the joint committee,
 - (3) if known, the expected return to work date,
 - (4) obligations and rights of employees with respect to recall, and change of address.
 - (b) Notify the Union in writing of the name, job title and pay group, seniority date and headquarters of the employees to be laid off.
- 33.05 Order of recall for employees who were laid off in accordance with Clause 33.01, shall be last out first in, through the medium of registered letter return receipt requested to the employee's last address of record, at any time within twelve (12) months from layoff, provided they are physically qualified and equipped by training and experience to perform the duties of the work available.
- 33.06 It will be the responsibility of former employees who desire re-employment within the terms above, to keep the Company informed as to their correct address and to advise the Company within one (1) week of the date of any offer of reemployment as to their acceptance. The Company will assume that failure on the part of any laid off employee to notify the Company within one (1) week concerning acceptance of an offer of employment or to report for duty within fifteen (15) days from the date of offer constitutes a rejection of the offer and a severance of the employment relationship.
- 33.07 (a) Employees who have been laid off and have been recalled to work within a period of twelve (12) months from the date of layoff shall retain all former credits.

(b) If the layoff period is thirty (30) or more consecutive days, an employee should refer to the Telus Corporation Pension Plan.

33.08 Severance Pay

A regular employee who is terminated following expiry of the twelve (12) month recall period shall be entitled to one (1) week of severance pay for each completed year of service up to twenty (20) years inclusive and two (2) weeks pay for each completed year of service over twenty (20) years. Severance pay will be pro-rated for any partial year of service.

33.09 An employee who has been laid off, may, within seven (7) days of the layoff, waive their right to recall and collect their severance pay.

ARTICLE 34 - LAYOFF (O.S.)

- 34.01 In the event of a force reduction of regular employees, the Company will effect the layoff on a last in, first out basis.
- 34.02 Should the Company decide to reduce the number of employees as referred to in Clause 34.01, casual and occasional employees shall be laid off prior to regular employees.
- 34.03 Prior to the Company exercising their rights in accordance with Clause 34.01, the Company shall:
 - (a) Notify the Union in writing not less than sixteen (16) weeks prior to the reduction. The notice will include the number of employees affected, their job title and pay group, their seniority date, and the reasons for the proposed reduction; and,

- (b) establish a joint committee with the Union to consider alternatives to and otherwise mitigate the effects of the proposed downsizing. The committee shall be a maximum of six (6) people equally represented by the Company and the Union and will meet within two (2) weeks of the notice. Alternatives may include voluntary resignation incentives, targeted resignation incentives, or other appropriate alternatives such as retraining/reskilling of employees or redeployment to other jobs; and,
- (c) jointly with the Union consider an outplacement program to assist laid off employees; and,
- (d) verify employee seniority, job title and pay group of those affected prior to the layoff.
- 34.04 Not less than ten (10) weeks prior to the layoff, the Company will:
 - (a) Send a notice to each affected employee containing;
 - (1) employee entitlements at time of layoff (e.g. UI, etc.),
 - (2) option being considered by the joint committee,
 - (3) if known, the expected return to work date,
 - (4) obligation and rights of employees with respect to recall, and change of address.
 - (b) Notify the Union in writing of the name, job title and pay group, seniority date and headquarters of the employees to be laid off.
- 34.05 Order of recall for employees who were laid off because of a force surplus as outlined in Clause 34.01, shall be last out first in, through the medium of a registered letter, return receipt requested, to the employee's last address of record, at any time within twelve (12) months from the date of layoff, provided they are physically qualified and equipped by training and experience to perform the duties of the work available.
- 34.06 It will be the responsibility of the former employees who desire re-employment within the terms above, to keep the Company informed as to their correct address and to advise the Company within one (1) week of the date of any offer of re-employment as to their acceptance. The Company will assume that failure on the part of any laid off employee to notify the Company within one (1) week concerning acceptance of an offer of employment or to report for duty within fifteen (15) days from the date of the offer constitutes a rejection of the offer and a severance of the employment relationship.
- 34.07 (a) Employees who have been laid off and have been recalled to work within a period of twelve (12) months from the date of layoff shall retain all former credits.

- (b) If the layoff period is thirty (30) or more consecutive days, and employee should refer to the Telus Corporation Pension Plan.
- 34.08 Severance Pay

A regular employee who is terminated following expiry of the twelve (12) month recall period shall be entitled to one (1) week of severance pay for each completed year of service up to twenty (20) years inclusive and two (2) weeks pay for each completed year of service over twenty (20) years. Severance pay will be pro-rated for any partial year of service.

34.09 An employee who has been laid off, may, within seven (7) days of the layoff, waive their right to recall and collect their severance pay.

ARTICLE 35 - JOB EVALUATION

- 35.01 (a) The Employer shall evaluate newly created bargaining unit jobs and any existing bargaining unit job which has had duties added, deleted or changed.
 - (b) The Employer shall establish a provisional rate for any new job prior to the job being filled.
 - (c) Employees may participate with their Manager in the preparation of the employee's job description.
- 35.02 Requests for Job Evaluation and Job Descriptions
 - (a) An employee or Supervisor may request an evaluation of an existing job by submitting a written request together with a current job description for the position to the compensation area of Human Resources.
 - (b) Upon request an employee shall be provided with a current job description for their position. The Employer shall provide the Union with a copy of the current job description for each bargaining unit position.

35.03 Job Evaluation Appeals

- (a) An employee or Supervisor who disputes the evaluation of a job may send a letter giving the reasons for challenging the evaluation to the compensation area of Human Resources requesting an appeal.
- (b) The appeal will be heard by the Job Evaluation Steering Committee. Those involved in presenting the appeal will be excluded from the presence of the Committee during its deliberations. The Committee hearing the appeals shall at all times have an equal number of Employer and Union representatives. The Job Evaluation Steering Committee shall communicate their decision to the employee and their Supervisor.
- (c) Should the decision of the Job Evaluation Steering Committee fail to resolve the matter, a grievance may be filed at the arbitration step of the grievance procedure. Final resolution may be arbitrable before a single arbitrator. The cost of the arbitrator shall be shared equally between the Employer and the Union.
- 35.04 (a) A Job Evaluation Steering Committee comprised of three (3) representatives chosen by the Employer and three (3) chosen by the Union shall be established. Each party shall designate one (1) alternate representative. The Committee shall be co-chaired by an Employer and Union representative and two (2) representatives from each party shall constitute a quorum.
 - (b) The Job Evaluation Steering Committee shall meet as required for the purpose of:
 - (1) Hearing job evaluation appeals and making determinations related thereto;
 - (2) Reviewing quarterly job evaluation summaries and initiating series reviews if deemed necessary;
 - (3) Reviewing issues related to the job evaluation process and forwarding recommendations to the Employer and Union;
 - (4) Reviewing evaluation of benchmark jobs;
 - (5) Reviewing the pay structure for expansion as may be deemed necessary.
 - (c) Job Evaluation Steering Committee members and alternates shall be appointed for two (2) year periods and may be re-appointed. Members shall be given training on the job evaluation plan and will be provided with all relevant documentation and any other necessary information.
 - (d) The Job Evaluation Steering Committee work shall be considered as time worked.

- (e) The Job Evaluation Steering Committee may submit issues to an arbitrator. The costs of the arbitrator shall be shared equally between the Company and the Union.
- 35.05 Wage Adjustments
 - (a) Any wage adjustment to a higher rate resulting from the evaluation of a newly created position shall be retroactive to the date the position was filled.
 - (b) Any wage adjustment to a higher rate made as the result of an evaluation request, appeal or arbitrator's decision shall be retroactive to the date the original request was submitted to the compensation area of Human Resources.
 - (c) Wage adjustments to a higher rate shall be either one (1) increment higher than the previous rate or the minimum rate of the new group, whichever is higher.
 - (d) A wage adjustment to a lower rate shall be effective the date of the determination. Incumbent employees shall have their rate of pay maintained at the current rate until either the adjusted job rate exceeds their maintained rate or the employee permanently leaves the pay group. Where possible, the Employer will reassign affected employees to vacancies in a pay group consistent with their maintained rate of pay. If the employee refuses the job offered the Employer is under no further obligation to offer the employee jobs consistent with their maintained rate of pay.
- 35.06 The Job Evaluation Plan and breakpoints now in effect shall form part of this Agreement. Any changes must be negotiated between the Union and the Employer.

ARTICLE 36 - UPGRADING COURSES

36.01 Where upgrading courses are required for employee development, or to meet the requirements of the job, the matter of payment of wages, travel time, expenses, tuition and course material shall be arranged for when approved by the appropriate Director.

ARTICLE 37 - JOB POSTINGS, TRANSFERS, SURPLUS PLACEMENTS AND REDEPLOYMENT

- 37.01 Job Postings:
 - (a) All regular clerical job vacancies in pay groups 46 and above shall be posted without geographic restriction.
 - (b) Clerical job vacancies in pay groups 44 and 45 shall be posted and the posting may restrict applications to the headquarters where the vacancy is located.
 - (c) Clerical job vacancies in pay groups 41 to 43 inclusive may be posted.
 - (d) All regular Operator Services job vacancies in pay groups 47 and above shall be posted without geographic restriction.
 - (e) Operator Services job vacancies pay groups 46 and below shall be posted however applications may be restricted to the headquarters where the vacancy is located.

(f) Where possible, occasional vacancies in all pay groups shall be posted, however, geographic restrictions may apply.

37.02 Job Postings: Out-of-Scope Vacancies

Job postings for out-of-scope (non-bargaining unit) positions shall not be subject to the selection criteria contained in this Collective Agreement.

- 37.03 Job Posting Restrictions
 - (a) In some cases, selection may be limited to regular employees.
 - (b) Applications for clerical job vacancies when clerical employees are on layoff shall be restricted to active clerical employees.

- (c) Applications for operator services job vacancies when operator services employees are on layoff shall be restricted to active operator services employees.
- (d) An "active employee" is one who has not been laid off.
- 37.04 Job postings for bargaining unit job vacancies shall be accessible to all employees and shall include the job functions, qualifications, required knowledge, educational and training requirements, skills, classification, wage range, headquarters where the vacancy is located and closing date for receipt of applications.
 - (a) The following order for consideration of applicants shall apply:
 - (i) regular full or part-time employees
 - (ii) next, if no one is chosen from (i), then occasional employees
 - (iii) next, if no one is chosen from (ii), then casual employees
 - (iv) next, if no one is chosen from (iii), then outside applicants
 - (b) The qualifications for the position shall be consistent with the responsibilities specified in the Job Description.
- 37.05 A copy of all postings for bargaining unit job vacancies will be sent to the Union.
- 37.06 The Company shall place an employee in another position if it is found that the employee cannot fulfill the position to which they were transferred.
- 37.07 Posting Exceptions

Job vacancies may be filled without posting in the following circumstances:

- (1) Through the compassionate transfer process; or,
- (2) by a regular employee whose current position has been declared surplus; or,
- (3) by a regular employee whose current position has been downgraded as a result of job evaluation; or,
- (4) by a regular employee who is occupying a lower rated job as a result of redeployment; or,
- (5) by the lateral transfer of an employee when the transfer is within a department within the employee's headquarters.

Priority for placement in vacancies being filled without posting shall be to first consider those employees approved for a compassionate transfer followed in order by those in surplus positions, those in positions which have been downgraded as a result of job evaluation, those occupying a lower rated job as a result of redeployment and finally through lateral transfer.

- 37.08 In filling job postings, the job shall be awarded based on seniority, qualifications and ability.
- 37.09 (a) All applicants shall be notified of the results of the job posting within thirty (30) days of the closing of the posting. Notification of the appointment shall be forwarded to the Local Union.
 - (b) A mutually agreeable transfer date for any successful applicant must be negotiated between the departments concerned. It is expected that under normal circumstances, the time taken for release of the successful applicant should not exceed thirty (30) days of their being chosen as the successful applicant. Successful applicants held beyond this thirty (30) day release period due to business requirements, will have their pay rate adjusted to the rate that will be applicable in their new position.
- 37.10 Transfers Operator Services
 - (a) An employee's written request for a change in either headquarters where the employee's work can move with the employee or employment status from regular full-time to regular part-time will be considered and may be granted at the discretion of and at no cost to the Employer. A copy of the employee's request shall be provided to the Union.
 - Note: Movement from regular part-time to regular full-time requires a job posting.
 - (b) When it becomes necessary to transfer Operator Services employees between offices within a headquarters area for periods of more than six (6) months, the following procedure will be followed:
 - (1) First, by canvassing for volunteers. Senior employees who volunteer and are qualified by training and experience to perform the duties of the work available will be given preference; and,
 - (2) Secondly, when there are insufficient volunteers, employees will be transferred in inverse order of seniority, provided they are qualified by training and experience to perform the duties of the work available.

This does not apply to developmental assignments.

37.11 Redeployment

- (a) Redeployment is the term used to describe the movement of employees as a result of:
 - (1) the employee's position being declared surplus; or,
 - (2) a layoff which has created vacancies that require filling to meet business needs; or,
 - (3) the closure or relocation of an office or department.

Other situations not satisfied by (1) (2) or (3) above, will be mutually agreed to by the Company and the Union.

- (b) Job vacancies may be filled with employees through redeployment without posting, however, prior to such vacancies being filled the matter will be reviewed between the Company and the Union.
- (c) There shall be no backfilling of positions which have been declared surplus.
- (d) Pay Treatment for Redeployed Employees
 - (1) The Employer will first attempt to place employees being redeployed to another position having either the same pay group as, or one pay group above the employee's current position.
 - (2) If the employee is redeployed to, or volunteers to accept a higher rated position, the employee's rate of pay will be increased by one increment, or the minimum wage of the new group, whichever is higher.
 - (3) If the employee is redeployed to or volunteers to accept a lower rated position, their salary will continue to be administered as if they had remained in their previous position for two (2) years from the date of transfer. The employee will then remain at their current pay rate until their rate is equal to or less than the maximum of the new pay group. If the employee refuses a job within their headquarters that is consistent with their maintained rate of pay, the Employer is under no further obligation to find a job consistent with the maintained rate.
- (e) Redeployment Process

The Employer will redeploy employees in accordance with the following sequence:

(1) For employees whose job is relocated, incumbents will be given first right to accept or reject the job in the new headquarters.

- (2) By canvassing for volunteers within all identified surplus positions within the headquarters requiring the redeployment. The most senior volunteers shall be selected provided the volunteers have the ability and qualifications to perform the required job.
- (3) By canvassing for volunteers within all identified surplus positions requiring redeployment. The most senior volunteers shall be selected provided the volunteer has the ability and qualifications to perform the required job.
- (4) If the necessary redeployment has not been achieved through volunteers, the Employer will transfer employees from positions requiring redeployment in inverse order of seniority, provided they have the ability and qualifications to perform the duties of the work available.
- (5) If more than one employee is to be redeployed, preference for available positions will be granted by seniority to those employees being redeployed.
- (f) If an employee is unable to accept permanent redeployment, the employee will be deemed to have resigned their employment.
- (g) Employees being redeployed by the Employer will be covered by the Company's Relocation Policy dated March 1995, and any revisions in progress.

ARTICLE 38 - TEMPORARY MOVEMENT OF EMPLOYEES

- 38.01 Temporary Assignments to Management Clerical
 - (a) When an employee is assigned to relieve either a Supervisor or another MP position for a temporary period not to exceed sixty (60) working days, they shall receive their basic rate of pay and in addition, shall receive from the first day, or portion thereof a differential of ten dollars (\$10.00) per day. An employee on a temporary assignment shall not be expected to perform the duties of their clerical position.
 - (b) (1) Management shall canvass at the beginning of each calendar year and establish a list of qualified volunteers in accordance with their seniority. During the calendar year, volunteers who become qualified or employees transferring into the Supervisory group that wish to volunteer and who are qualified, shall be added to the bottom of the rotation.
 - (2) Where there is more than one (1) qualified volunteer, the Employer will select employees on a rotational basis from those on the list to fill

temporary assignments. If there is no qualified volunteer, the selection process may be expanded to other groups.

- (3) An employee who has served sixty (60) working days in temporary assignment(s) to Management shall be removed from the rotation for the remainder of the calendar year.
- (c) Such temporary increases shall not interfere in any way with any progressional wage increase which may fall due during the period the employee is assigned to a temporary position.
- 38.02 Temporary Assignments to Management Operator Services

Service Assistant and Support

- (a) An employee performing the duties of a Service Assistant and Support shall be paid a differential of ten dollars (\$10.00) for each day or portion of a day worked in this capacity.
- (b) An employee shall not perform the duties of a Service Assistant and Support for more than a total of sixty (60) working days in a year.
- (c) Employees will be selected for Service Assistant and Support on the basis of having the necessary qualifications and leadership abilities.
- (d) Service Assistant and Support opportunities shall be posted on a per office basis.
- (e) Service Assistant and Support shall not perform Feedback and Development Reviews or associated processes (e.g. Performance Samples), impose formal discipline or have access to any other employee personnel files.
- 38.03 Temporary Assignments to a Higher Rated Job

<u>Clerical</u>

- (a) Management, where deemed necessary, may appoint a qualified clerical employee to act in a higher rated clerical job for a temporary period not to exceed sixty (60) working days.
- (b) Management shall canvass for volunteers and shall ensure equal opportunity for qualified volunteers. Assignments shall be in order of seniority. No employee will be assigned more than once until all qualified volunteers have been assigned.

- (c) An employee when appointed to assume the primary responsibilities of a higher position shall receive payment in accordance with the following provisions:
 - (1) Where because of service requirements an employee is appointed to work in a higher rated clerical job for a temporary period, the employee will receive a differential of four dollars (\$4.00) per day or the minimum wage of the new group, whichever is higher, from the first day of the appointment. At the conclusion of the temporary appointment, the employee will revert to their former rate of pay. It is agreed that the appointment shall not be made more than three (3) groups higher without mutual consent of the Union and the Employer.
 - (2) Temporary assignments may exceed sixty (60) working days in cases of replacements for: maternity leave, child care leave, adoption leave, extended sick leave, furlough or other specific circumstances mutually agreed to by the Employer and the Union. Where these replacements involve more than one (1) move, all the moves will be deemed to be as a result of the above, (e.g. When the absent employee is a Group 47, then the 46 to 47, 45 to 46, 44 to 45 etc. are considered part of the original move).

Operator Services

- (a) Management where deemed necessary, may appoint a qualified Operator Services employee to act in a higher rated Operator Services job for a temporary period not to exceed sixty (60) working days.
- (b) Management shall canvass for volunteers and shall ensure equal opportunity for qualified volunteers.

- (c) An employee when appointed to assume the primary responsibilities of a higher position shall receive payment in accordance with the following provisions:
 - (1) Where because of service requirements an employee is appointed to work in a higher rated Operator Services job for a temporary period, the employee will receive a differential of four dollars (\$4.00) per day or the minimum wage of the new group, whichever is higher, from the first day of the appointment. At the conclusion of the temporary appointment, the employee will revert to their former rate of pay. It is agreed that the appointment shall not be made more than three (3) groups higher without mutual consent of the Union and the Employer.
 - (2) Temporary assignments may exceed sixty (60) working days in cases of replacement for: maternity leave, child care leave, adoption leave, extended sick leave, furlough or other specific circumstances mutually agreed to by the Employer and the Union. Where these replacements involve more than one (1) move, all the moves will be deemed to be as a result of the above, (e.g. When the absent employee is a Group 47, then the 46 to 47, 45 to 46, 44 to 45 etc. are considered part of the original move).
- 38.04 Temporary Assignments to Other Bargaining Units

The Company and Union will review on a case by case basis opportunities and conditions for temporarily assigning bargaining unit employees to perform work in another bargaining unit. No bargaining unit employees will be temporarily assigned work outside the bargaining unit without the agreement of the Union.

38.05 Other Temporary Assignments Within the Bargaining Unit

The Company and Union will review on a case by case basis opportunities and conditions for temporarily assigning bargaining unit employees to perform work not otherwise addressed in this Article. Exceptions to the temporary assignment conditions contained herein must be agreed to by the Union.

- 38.06 Acting Assignments Clerical/Operator Services
 - (a) Acting assignments shall be filled by way of job postings in accordance with Article 37.
 - (b) When an employee is successful on a posting to relieve a manager or other position not covered by this agreement for a period in excess of ninety (90) working days, they shall be paid in accordance with the following provisions:

- (1) For the duration of the acting assignment, their compensation shall be administered in the higher level position as though they occupied that position on a permanent basis; and,
- (2) When an employee who has been in an acting assignment returns to their regular position, their salary and anniversary date shall be readjusted to that which would have been in effect if they had continuously occupied that position.
- (c) Union dues, based either on the employee's rate of pay in their last bargaining unit position or bargaining unit acting position, shall be remitted to the Union for each month of the acting assignment, or; if the employee is working in another position covered by the IBEW then dues shall be remitted in accordance with that Agreement.
- (d) Acting assignments shall not exceed a period of two (2) years.
- (e) There shall be a one year break for every employee between acting assignments.
- (f) An employee on an acting assignment shall remain a bargaining unit employee and shall retain their seniority, right of grievance and be subject to dues deduction and layoff provisions. All other terms and conditions of this agreement shall be suspended for the duration of the acting assignment.

ARTICLE 39 - DIFFERENTIALS AND SPECIAL PAYMENTS (Clerical)

- 39.01 Tour Differentials
 - (a) An employee who is required to work an evening or night tour as defined in Article 16.02, will be paid a tour differential of eighty-five cents (\$0.85) per hour for each hour or portion thereof that they are required to work between the hours of 6:01 p.m. and 8:00 a.m.
 - (b) Clause (a) notwithstanding, tour differentials shall not be paid on days off, Sundays, Holidays (Company and Civic), Christmas Eve, New Year's Eve or when employees are paid on an overtime basis.
- 39.02 An employee working a night tour shall receive a differential of two dollars and fifty cents (\$2.50) for every tour where relief is not provided.

39.03 Saturday Differential

An employee who, at the direction of the Employer, works consecutive Saturdays shall be paid a differential of two dollars and fifty cents (\$2.50) for each hour or portion thereof worked on the second and any subsequent consecutive Saturday. A differential shall not be paid when an employee works a consecutive Saturday as the result of an employee arranged tour change.

- **39.04** Sunday Premium
 - (a) An employee who works a Sunday tour of duty shall receive their basic rate of pay and in addition they shall receive a Sunday premium of one-half (1/2) times their basic hourly rate of pay for each hour or portion thereof worked.
 - (b) Sunday premium shall not be paid when an employee works overtime, a Holiday (Company or Civic), Christmas Eve or New Year's Eve.

39.05 Training Differential

- (a) An employee who is temporarily assigned training duties will be paid a differential of sixty-five cents (\$0.65) per hour for each hour or portion thereof that is spent in providing training.
- (b) Training shall be defined as:
 - (1) Instruction given in new techniques or procedures; or,
 - (2) demonstration of job duties.
- (c) A training differential shall not be paid when:
 - (1) An employee's normal duties as defined in the job description includes training activities; or,
 - (2) a differential as per Clause 38.01 (a), is paid; or,
 - (3) a temporary assignment differential is being paid as per Clerical Clause 38.03 (c) (1), and the higher rated job, as defined in the job description, includes training activities.

39.06 Christmas and New Year's Eve Special Payment

For work performed on December 24 and December 31, the following special payment shall apply:

- (a) An employee working after 6:00 p.m. on the evening of either day shall receive **their** basic rate of pay and in addition shall receive Christmas or New Year's Eve special payment of one (1) times **their** basic hourly rate for each hour worked after 6:00 p.m. up to the ending time of duty.
- (b) An employee called in to work overtime shall be paid in accordance with the overtime regulations.
- NOTE: Temporary Assignment Differentials See Clause 38.03 Clerical

ARTICLE 40 - DIFFERENTIALS AND SPECIAL PAYMENTS (Operator Services)

- 40.01 (a) An employee who is required to work an afternoon, or evening tour, as defined in Clause 17.02 (c) and (d) will be paid a tour differential of eighty-five cents (\$0.85) per hour for each hour or portion thereof that they are required to work between the hours of 4:00 p.m. and 12:00 midnight.
 - (b) An employee who is required to work a day, evening, or night tour, as defined in Clause 17.02 (b), (d) and (e), will be paid a tour differential of eighty-five cents (\$0.85) per hour for each hour or portion thereof that they are required to work between the hours of 12:00 midnight and 7:00 a.m.
 - (c) Tour differentials shall not be paid on days off duty, Sundays, Holidays, Christmas and New Year's Eve or when employees are paid on an overtime basis nor to any day tour employee, as defined in Clause 17.02(b) for hours which fall between the hours of 7:00 a.m. and 6:00 p.m.
- 40.02 (a) An employee who is required to work a Sunday tour of duty shall receive their basic rate of pay and in addition they shall receive a Sunday premium calculated as follows: One-half (1/2) times their basic hourly rate of pay for each hour or portion thereof worked.

Sunday premium shall not be paid when an employee works overtime, on a Holiday, Christmas Eve or New Year's Eve.

- (b) An employee taken ill while at work during a Sunday session or tour, shall be assumed to have worked the session or tour as the case may be, and shall be paid the Sunday premium per Clause 40.02 (a).
- (c) A Sunday night tour shall be from 12:00 midnight on Saturday to 7:30 a.m. on the following Sunday and paid for in accordance with the foregoing clauses.
- 40.03 An employee assigned to work split tours shall receive a differential of three dollars and fifty cents (\$3.50) for each completed tour.
- 40.04 An employee working the night tour shall receive a differential of two (2) hours pay at their basic rate of pay for every tour where no relief is provided.
- 40.05 Service Assistant (Special Services)
 - (a) An employee performing the duties of a Service Assistant, shall be paid a differential of one dollar twenty cents (\$1.20) for each hour or portion thereof worked in this capacity.
 - (b) An employee shall not perform the duties of a Service Assistant for more than a total of one hundred twenty (120) days/eight hundred forty (840) hours in a year.
 - (c) The employee shall be selected on the basis that they have the fundamental knowledge of operator practices and demonstrate good communication skills.
 - (d) These assignments shall be done on an equitable basis.
 - (e) Service Assistant opportunities shall be posted on a per office basis.

40.06 Christmas Eve and New Year's Eve Special Payment

For work performed on December 24 and December 31, the following special payment shall apply:

(a) An employee working after 6:00 p.m. on the evening of either day shall receive their basic rate of pay and in addition shall receive Christmas Eve or New Year's Eve special payment calculated as follows:

One-times their basic hourly rate for each hour worked after 6:00 p.m. up to the end of their tour of duty.

(b) An employee called in to work overtime shall be paid in accordance with the overtime regulations.

- **40.07** An employee who, at the direction of the Employer, works consecutive Saturdays, shall be paid a differential of two dollars and fifty cents (\$2.50) for each hour or portion thereof worked on the second and any subsequent consecutive Saturday. A differential shall not be paid when an employee works a consecutive Saturday as the result of an employee arranged tour change.
- **40.08** Training Differential
 - (a) An employee who is temporarily assigned training duties will be paid a differential of sixty-five cents (\$0.65) per hour for each hour or portion thereof that is spent in providing training.
 - (b) Training shall be defined as:
 - (1) Instruction given in new techniques or procedures; or,
 - (2) demonstration of job duties.
 - (c) A training differential shall not be paid when:
 - (1) An employee's normal duties as defined in the job description includes training activities; or,
 - (2) a differential as per Operator Services Clause 38.02 (a), is paid; or,
 - (3) a temporary assignment differential is being paid as per Operator Services Clause 38.03 (c) (1), and the higher rated job, as defined in the job description, includes training activities.
- NOTE: Temporary Assignment Differentials See Operator Services Clause 38.02, 38.03.

ARTICLE 41 - PAY PROVISIONS FOR EMPLOYEES ABSENT FROM DUTY

41.01 Court Witness

- (a) An employee absent from work as a result of a subpoena or summons to appear as a witness shall be allowed time off with pay for **TELUS** scheduled hours of work. The witness fees received shall be turned over to the immediate Manager. The employee will not be allowed to claim expenses such as meals, lodging, and transportation from **TELUS** where they are covered by the Court.
- (b) An employee acting as a voluntary witness is not paid for such absence.
- (c) An employee appearing as a witness on behalf of the Company or as a result of employment with the Company, will receive the normal **TELUS** salary and expenses.
- 41.02 Jury Duty

An employee receiving a Court notice to appear for jury duty will inform their Supervisor. The employee may contact the Law Department if advice is required.

- (a) Time off with pay will be allowed for jury duty.
- (b) Jury Duty fees received shall be turned over to the immediate Manager.
- (c) Employees will not be allowed to claim expenses (i.e. meals, lodging, transportation or incidentals) from **TELUS** where they are covered by the Court.
- (d) The employee may elect to retain the jury fee and have the time off as approved furlough.
- 41.03 Elections

Employees eligible to vote shall be allowed sufficient time off with pay to meet legal requirements for voting purposes. Where employees are registered to vote in localities other than where they are working, the provisions of the applicable legislation governing the specific municipal, provincial or federal election shall apply. Where possible, such employees will be encouraged to vote at advance polls.

41.04 Bereavement Leave

In the case of a death in the immediate family, an employee shall if required, be given time off with pay up to a maximum of three (3) working days. Additional time with pay for up to two (2) days shall be granted if required for either travelling or where an employee is charged with the responsibilities of making funeral arrangements. The term "immediate family" shall be interpreted to mean a mother, father, spouse (including common-law), sister, brother, son, daughter, mother-in-law, father-in-law, grandfather, grandmother, sister-in-law, brother-in-law, son-in-law, daughter-in-law, granddaughter and grandson and any relative of the employee who resides permanently in the employee's household or with whom the employee permanently resides.

- (1) Where circumstances may warrant furlough, in addition to bereavement leave, approval will be given immediate consideration by the employee's immediate manager.
- (2) Should a holiday fall during a period of bereavement leave, the day shall be paid as holiday pay and will not extend the bereavement leave.
- (3) Bereavement occurring during an employee's vacation period shall extend the vacation by the amount of days allowed for bereavement leave in accordance with the foregoing.

41.05 Funeral Leave

In the case of a death of a friend, working associate or relative not covered under Clause 41.04, an employee may be given time off with pay not to exceed one (1) day to attend the funeral. This will be subject to service requirements and management approval. Such requests shall not be unreasonably denied.

41.06 Pallbearer Leave

When an employee serves as a pallbearer, the employee shall be allowed time off work with pay up to a maximum of one (1) day.

41.07 Serious Distress

"Serious Distress" has reference to a serious, critical condition which requires the personal attention or presence of the employee at a time over which the employee has no control and which cannot properly be served by others, or attended to by the employee at any time when the employee is off duty. Circumstances of employee request for leave under serious distress shall receive just, reasonable and immediate consideration and granted where required, accordingly:

(a) Short Term Leave With Pay

Short term leave up to a maximum of one (1) day may be granted by the immediate Supervisor. Additional time may be granted if necessary, by the Manager's Manager who will decide what time off, with pay, will be granted.

(b) Use of Advance Vacation

If necessary, up to one (1) week of advance vacation may be granted by the employee's immediate Supervisor.

41.08 Medical, Dental or Other Special Appointments

An employee requesting time off for a medical or dental or other special appointment shall be granted time off with pay up to a maximum of one (1) day by the immediate manager subject to demands of service. If any additional expense to the Company is involved, the employee may be allowed furlough.

41.09 Work for Another Company

An employee without Company approval accepting employment with a firm, agency, or other person on the employee's own time with remuneration shall not be deemed to be an employee of the Company during the period of such employment.

41.10 Time Lost Other Than Sickness

Payment of wages will not be allowed for time lost, by any employee, for reasons other than sickness or those covered under the terms of this Agreement.

41.11 Furlough

- (a) The Employer may grant furlough without pay or benefits for personal reasons, subject to the demands of service. Such furlough will not be unreasonably withheld.
- (c) Requests for furlough must be submitted in writing to the employee's Supervisor not less than four (4) weeks prior to the anticipated commencement date of the furlough. Lack of four (4) weeks written notice shall not preclude the granting of furlough at the employee's request and agreement by the Supervisor.

41.12 Absence (Clerical)

An employee who is absent from work for any reason, without prior authorization is required to contact their Supervisor at their place of work, giving the reason for their absence, within the first two (2) hours of such absence. Failure to report absence or an absence which is not authorized by the Supervisor may result in disciplinary action pursuant to Article 15: Discipline, Resignation and Dismissal.

41.13 Absence (Operator Services)

If an employee is unable to report for work for any reason, they will notify the Supervisor, wherever possible, before the start of their tour. Failure to report and in the absence of satisfactory reason(s) then disciplinary action may result.

ARTICLE 42 - MEALS, LODGING, TRANSPORTATION AND TRAVELLING TIME

- 42.01 Expenses set out in this Article are to cover expenditures necessarily incurred by employees in the performance of their duties while working away from their headquarters. Where no expense is incurred, no claim shall be made. An employee who has been authorized by their Supervisor shall be allowed expenses as stated herein.
- 42.02 Meals
 - (a) Employees working outside their headquarters who are absent overnight shall be allowed reasonable and actual expenses for meals (breakfast, lunch and supper). Receipts are required.
 - (b) An employee who commutes daily from their headquarters may only claim lunch.
- 42.03 Lodging
 - (a) For overnight absences, reasonable and actual expenses will be allowed for commercial lodging upon the production of receipts.
 - (b) An employee who makes private arrangements for accommodation other than their place of domicile, may claim an allowance of twenty-five dollars (\$25.00) per night. The employee may claim meals in accordance with Clause 42.02 (a).
- 42.04 Incidental Expenses

An incidental expense of five dollars (\$5.00) for each overnight stay will be granted from the first day for an employee travelling away from their headquarters for twenty-four (24) hours or more. Incidental expenses are intended to cover such items as laundry, gratuities, etc.

42.05 Extraordinary Expenses

When extraordinary expenses are incurred which are above the allowances stated in this Article, the employee with appropriate justification, will be allowed actual living expenses upon production of receipts.

- 42.06 Overtime Meal Allowance
 - (a) An employee may claim an overtime meal allowance of nine dollars (\$9.00) without a receipt in accordance with the following conditions:
 - (1) Employees working overtime within their headquarters for two (2) or more hours;
 - (i) following the end of a regular or overtime tour, or;
 - (ii) preceding a regular or overtime tour.
 - (iii) when an Operator Services employee works overtime for two (2) hours or more between the two portions of a regular or overtime split tour.
 - (2) An employee claiming lodging away from headquarters may claim an overtime meal allowance if they have worked four (4) or more hours of overtime following or preceding their regular or overtime tour.
 - (3) Employees working away from headquarters not incurring lodging:

An employee working two (2) or more hours of overtime following or preceding their regular or overtime tour at a location outside their headquarters and who will not be incurring an overnight stay, may claim the appropriate meal allowance in accordance with Clause 42.02 (a).

- (4) An employee who is called out on overtime for immediate reporting to work on their day off or after they have completed their regular tour may claim an overtime meal allowance after four (4) hours and a similar allowance after each additional continuous four (4) hours of overtime worked.
- (b) An employee may claim an overtime meal allowance for each additional continuous four (4) hours of overtime worked.
- (c) In special circumstances, the Supervisor may authorize payment of an overtime meal payment not covered by the foregoing or when the amount is exceeded.
- (d) Overtime meal breaks may be scheduled at any time subject to the welfare of the employee and the requirements of service.
- (e) Overtime will not be paid during any meal period.
- 42.07 Travel Time

An employee travelling at the direction of the Employer shall be paid at their basic hourly rate of pay or if applicable, the overtime rate.

- 42.08 An employee who resigns, is dismissed or laid off when working outside their headquarters, will be provided return transportation to their headquarters.
- 42.09 For the purpose of conducting Company business, transportation will be furnished or allowed all employees covered by this Agreement where same is considered necessary by the Company.

ARTICLE 43 - PAY PROVISIONS AND WAGE SCHEDULES

43.01 Non-permanent Experienced Employee

When a non-permanent experienced employee transfers to the permanent staff, either regular full-time or regular part-time, reasonable allowance will be made for past service in setting the wage rate as a permanent employee.

43.02 Payments

Employees shall be paid on alternate Tuesdays throughout the year, for all basic wages due up to and including the Saturday following the previous pay date.

In the case of missing, delayed or incorrect work reports, at the time of closing off the payroll, it will be assumed that the basic number of hours were worked. Any necessary adjustments for overtime, differentials, or absences will be made on the following pay. When a regular pay day falls on a holiday, payment of wages shall be made on the preceding working day.

43.03 Wage increases granted in accordance with progression steps of the wage schedules contained in this Agreement shall be effective **on the actual date the increase is due.**

- 43.04 When an employee is given a wage increase or adjustment for reason of efficiency or promotion, the normal date of the scheduled increase will still apply, providing they have not reached their maximum rate. If they have reached their maximum rate and are promoted to a new job paying a higher rate, their scheduled increase date shall be calculated from the time of such a move. The provisions of this Clause will also apply when an employee's job is re-evaluated to a higher group.
- 43.05 Wage Adjustment Upon Promotion
 - (a) Promotion to a higher group adjust wage to:
 - (1) One (1) increment (money wise) higher than the former wage; or,
 - (2) the minimum wage of the new group, whichever is higher.
 - (b) The Employer may authorize the payment of a wage higher than stipulated by Clause (a) provided such wage shall not exceed the maximum in the new group.
- 43.06 Suspension of Increases
 - (a) A salary increase falling due when an employee is off duty on sick leave or furlough will be suspended during the period of absence.
 - (b) A salary increase falling due when an employee is on sick leave for three (3) months or less, will take effect when the employee resumes duty; provided however, that for subsequent increases the normal date of scheduled increase will apply.
 - (c) When an employee is on sick leave for a period of more than three (3) months, their subsequent increases will be set back the length of time corresponding to the duration of the period of absence in excess of three (3) months.
 - (d) When an employee is granted furlough for a period of more than one (1) month, their subsequent increases will be set back the length of time corresponding to the duration of the furlough.

43.07 Administration of Progression Wage Schedule

Progression through the semi-annual steps of the wage schedules as contained in this Agreement shall be subject to the periodic appraisal by Management of an employee's work performance. Should Management consider that the employee has not qualified for a semi-annual increase, the employee will be notified in writing with a copy to the Union, of the reasons not less than thirty (30) days prior to the increase due date. An employee so affected may, within ten (10) days of the notice in writing, review the matter with the Supervisor. Should the employee, following the review feel that their increase should not be withheld, they may take the matter up as a grievance.

- 43.08 The amount of increases granted shall be those indicated in the wage schedules and increases shall not be granted for amounts less than those indicated on the appropriate step in the schedule, subject to the provisions of Clause 43.07.
- 43.09 Should a special increase be warranted during the progression period, it will be given every consideration by Management.
- 43.10 While supervisors will attempt, as far as possible, to assign the employee to work for which the employee has been trained, no part of this Agreement shall be construed as meaning that an employee shall do only work of the classification in which they are employed, nor shall any part of this Agreement be construed as meaning that certain work shall be performed only by a certain classified employee.
- 43.11 Variable Pay
 - (a) All regular full-time, regular part-time and occasional employees will participate in the Company's Variable Compensation Plan. If the plan's goals for a year are met, employees will receive a payment in the first quarter of the following year equal to three percent (3%) of the employee's total regular earnings (exclusive of all overtime, premiums and differentials) in the previous payroll year. "Payroll year" means the pay periods used by TELUS to determine the gross earnings for the purposes of producing yearly statements for income tax purposes.
 - (b) This plan will continue from year to year.
 - (c) Changes to the plan, will be discussed with the union prior to implementation.
 - (d) Variable pay is separate and apart from the Wage Schedules.

43.12 Wage Schedules

The conditions of this Agreement apply to all employees designated in the wage schedules which follow.

The daily and monthly rates are based on:

Clerical: Hourly Rate X 7.5 = Daily Rate Hourly Rate X **162.5** = Monthly Rate

Operator Services: Hourly Rate X 7 = Daily Rate Hourly Rate X **151.666** = Monthly Rate

Wage Schedules

Note: The interval from one step to the next shall be six (6) months. For occasional employees, the step shall be based on the employee working:

Clerical

the greater of:

(a) **975** hours (including overtime hours @ actual hours worked) OR

(b) six (6) months

Operator Services

the greater of:

(a) **910** hours (including overtime hours @ actual hours worked) OR

(b) six (6) months

Clerical and Operator Services Wage Schedules

Effective May 1, 1998 (Hourly Rates)

STEP	40	41	42	43	44	45	46	47	48	49	50	51	52
А	10.92	11.36	11.76	12.18	12.89	13.31	13.69	13.96	14.40	14.83	15.31	15.85	16.33
В	11.36	11.76	12.18	12.55	13.31	13.69	13.96	14.40	14.83	15.31	15.85	16.33	16.92
С	11.76	12.18	12.55	12.89	13.69	13.96	14.40	14.83	15.31	15.85	16.33	16.92	17.59
D	12.18	12.55	12.89	13.31	13.96	14.40	14.83	15.31	15.85	16.33	16.92	17.59	18.17
E	12.55	12.89	13.31	13.69	14.40	14.83	15.31	15.85	16.33	16.92	17.59	18.17	18.74
F	12.89	13.31	13.69	13.96	14.83	15.31	15.85	16.33	16.92	17.59	18.17	18.74	19.18
G		13.69	13.96	14.40	15.31	15.85	16.33	16.92	17.59	18.17	18.74	19.18	19.63
Н			14.40	14.83	15.85	16.33	16.92	17.59	18.17	18.74	19.18	19.63	20.05
1				15.31	16.33	16.92	17.59	18.17	18.74	19.18	19.63	20.05	20.49

Effective May 1, 1999 (Hourly Rates) Note: Rates already include 2% increase for April 30, 1999

STEP	40	41	42	43	44	45	46	47	48	49	50	51	52
А	11.36	11.82	12.24	12.67	13.41	13.85	14.24	14.52	14.98	15.43	15.93	16.49	16.99
В	11.82	12.24	12.67	13.06	13.85	14.24	14.52	14.98	15.43	15.93	16.49	16.99	17.61
С	12.24	12.67	13.06	13.41	14.24	14.52	14.98	15.43	15.93	16.49	16.99	17.61	18.30
D	12.67	13.06	13.41	13.85	14.52	14.98	15.43	15.93	16.49	16.99	17.61	18.30	18.90
E	13.06	13.41	13.85	14.24	14.98	15.43	15.93	16.49	16.99	17.61	18.30	18.90	19.50
F	13.41	13.85	14.24	14.52	15.43	15.93	16.49	16.99	17.61	18.30	18.90	19.50	19.95
G		14.24	14.52	14.98	15.93	16.49	16.99	17.61	18.30	18.90	19.50	19.95	20.42
Н			14.98	15.43	16.49	16.99	17.61	18.30	18.90	19.50	19.95	20.42	20.86
1				15.93	16.99	17.61	18.30	18.90	19.50	19.95	20.42	20.86	21.32

Effective May 1, 2000 (Hourly Rates)

STEP	40	41	42	43	44	45	46	47	48	49	50	51	52
А	11.59	12.06	12.49	12.92	13.68	14.13	14.53	14.81	15.28	15.74	16.25	16.82	17.33
В	12.06	12.49	12.92	13.32	14.13	14.53	14.81	15.28	15.74	16.25	16.82	17.33	17.96
С	12.49	12.92	13.32	13.68	14.53	14.81	15.28	15.74	16.25	16.82	17.33	17.96	18.67
D	12.92	13.32	13.68	14.13	14.81	15.28	15.74	16.25	16.82	17.33	17.96	18.67	19.28
E	13.32	13.68	14.13	14.53	15.28	15.74	16.25	16.82	17.33	17.96	18.67	19.28	19.89
F	13.68	14.13	14.53	14.81	15.74	16.25	16.82	17.33	17.96	18.67	19.28	19.89	20.35
G		14.53	14.81	15.28	16.25	16.82	17.33	17.96	18.67	19.28	19.89	20.35	20.83
Н			15.28	15.74	16.82	17.33	17.96	18.67	19.28	19.89	20.35	20.83	21.28
				16.25	17.33	17.96	18.67	19.28	19.89	20.35	20.83	21.28	21.75

Note: The interval from one step to the next shall be administered as per Article 43 (Pay Provisions and Wage Schedules).

Effective May 1, 1998

Clerical (Based on a 7.5 hour day)

day)

[Daily	ſ	Monthly		Yearly		Н	ourly	D	aily	М	onthly	Yearly
\$	81.90	\$	1,774.77	\$	21,294.00		\$	10.92	\$	76.44	\$	1,656.45	\$ 19,874.40
\$	85.20	\$	1,846.28	\$	22,152.00		\$	11.36	\$	79.52	\$	1,723.20	\$ 20,675.20
\$	88.20	\$	1,911.29	\$	22,932.00		\$	11.76	\$	82.32	\$	1,783.87	\$ 21,403.20
\$	91.35	\$	1,979.55	\$	23,751.00		\$	12.18	\$	85.26	\$	1,847.58	\$ 22,167.60
\$	94.13	\$	2,039.69	\$	24,472.50		\$	12.55	\$	87.85	\$	1,903.71	\$ 22,841.00
\$	96.68	\$	2,094.95	\$	25,135.50		\$	12.89	\$	90.23	\$	1,955.28	\$ 23,459.80
\$	99.83	\$	2,163.21	\$	25,954.50		\$	13.31	\$	93.17	\$	2,018.99	\$ 24,224.20
\$	102.68	\$	2,224.97	\$	26,695.50		\$	13.69	\$	95.83	\$	2,076.64	\$ 24,915.80
\$	104.70	\$	2,268.85	\$	27,222.00		\$	13.96	\$	97.72	\$	2,117.59	\$ 25,407.20
\$	108.00	\$	2,340.36	\$	28,080.00		\$	14.40	\$	100.80	\$	2,184.34	\$ 26,208.00
\$	111.23	\$	2,410.25	\$	28,918.50		\$	14.83	\$	103.81	\$	2,249.56	\$ 26,990.60
\$	114.83	\$	2,488.26	\$	29,854.50		\$	15.31	\$	107.17	\$	2,322.37	\$ 27,864.20
\$	118.88	\$	2,576.02	\$	30,907.50		\$	15.85	\$	110.95	\$	2,404.29	\$ 28,847.00
\$	122.48	\$	2,654.03	\$	31,843.50		\$	16.33	\$	114.31	\$	2,477.10	\$ 29,720.60
\$	126.90	\$	2,749.92	\$	32,994.00		\$	16.92	\$	118.44	\$	2,566.59	\$ 30,794.40
\$	131.93	\$	2,858.81	\$	34,300.50		\$	17.59	\$	123.13	\$	2,668.23	\$ 32,013.80
\$	136.28	\$	2,953.08	\$	35,431.50		\$	18.17	\$	127.19	\$	2,756.21	\$ 33,069.40
\$	140.55	\$	3,045.72	\$	36,543.00		\$	18.74	\$	131.18	\$	2,842.67	\$ 34,106.80
\$	143.85	\$	3,117.23	\$	37,401.00		\$	19.18	\$	134.26	\$	2,909.41	\$ 34,907.60
\$	147.23	\$	3,190.37	\$	38,278.50		\$	19.63	\$	137.41	\$	2,977.67	\$ 35,726.60
\$	150.38	\$	3,258.63	\$	39,097.50		\$	20.05	\$	140.35	\$	3,041.38	\$ 36,491.00
\$	153.68	\$	3,330.14	\$	39,955.50		\$	20.49	\$	143.43	\$	3,108.13	\$ 37,291.80
)aily, r	nonthly and	year	ly rates are no	ot exa	ct due to rou	ndir	ng.						

Effective May 1, 1999

Clerical (Based on a 7.5 hour day)

day)

Operator Services (Based on a 7 hour
----------------------------	-------------------

l	Daily	I	Monthly		Yearly		ŀ	lourly	D	Daily	М	onthly	Yearly
\$	85.20	\$	1,846.28	\$	22,152.00		\$	11.36	\$	79.52	\$	1,723.20	\$ 20,675.20
\$	88.65	\$	1,921.05	\$	23,049.00		\$	11.82	\$	82.74	\$	1,792.98	\$ 21,512.40
\$	91.80	\$	1,989.31	\$	23,868.00		\$	12.24	\$	85.68	\$	1,856.69	\$ 22,276.80
\$	95.03	\$	2,059.19	\$	24,706.50		\$	12.67	\$	88.69	\$	1,921.91	\$ 23,059.40
\$	97.95	\$	2,122.58	\$	25,467.00		\$	13.06	\$	91.42	\$	1,981.07	\$ 23,769.20
\$	100.58	\$	2,179.46	\$	26,149.50		\$	13.41	\$	93.87	\$	2,034.16	\$ 24,406.20
\$	103.88	\$	2,250.97	\$	27,007.50		\$	13.85	\$	96.95	\$	2,100.91	\$ 25,207.00
\$	106.80	\$	2,314.36	\$	27,768.00		\$	14.24	\$	99.68	\$	2,160.07	\$ 25,916.80
\$	108.90	\$	2,359.86	\$	28,314.00		\$	14.52	\$	101.64	\$	2,202.54	\$ 26,426.40
\$	112.35	\$	2,434.62	\$	29,211.00		\$	14.98	\$	104.86	\$	2,272.32	\$ 27,263.60
\$	115.73	\$	2,507.76	\$	30,088.50		\$	15.43	\$	108.01	\$	2,340.58	\$ 28,082.60
\$	119.48	\$	2,589.02	\$	31,063.50		\$	15.93	\$	111.51	\$	2,416.42	\$ 28,992.60
\$	123.68	\$	2,680.04	\$	32,155.50		\$	16.49	\$	115.43	\$	2,501.37	\$ 30,011.80
\$	127.43	\$	2,761.30	\$	33,130.50		\$	16.99	\$	118.93	\$	2,577.21	\$ 30,921.80
\$	132.08	\$	2,862.07	\$	34,339.50		\$	17.61	\$	123.27	\$	2,671.26	\$ 32,050.20
\$	137.25	\$	2,974.21	\$	35,685.00		\$	18.30	\$	128.10	\$	2,775.93	\$ 33,306.00
\$	141.75	\$	3,071.72	\$	36,855.00		\$	18.90	\$	132.30	\$	2,866.94	\$ 34,398.00
\$	146.25	\$	3,169.24	\$	38,025.00		\$	19.50	\$	136.50	\$	2,957.96	\$ 35,490.00
\$	149.63	\$	3,242.37	\$	38,902.50		\$	19.95	\$	139.65	\$	3,026.22	\$ 36,309.00
\$	153.15	\$	3,318.76	\$	39,819.00		\$	20.42	\$	142.94	\$	3,097.51	\$ 37,164.40
\$	156.45	\$	3,390.27	\$	40,677.00		\$	20.86	\$	146.02	\$	3,164.25	\$ 37,965.20
\$	159.90	\$	3,465.03	\$	41,574.00		\$	21.32	\$	149.24	\$	3,234.03	\$ 38,802.40
Daily, ı	nonthly and	year	ly rates are no	ot exa	ct due to rour	ndin	g.						

Effective May 1, 2000

Clerical (Based on a 7.5 hour day)

Operator Services (Based on a 7 hour

day)

I	Daily	I	Monthly	,	Yearly		ŀ	lourly	D	aily	M	onthly	•	Yearly
\$	86.93	\$	1,883.66	\$	22,600.50		\$	11.59	\$	81.13	\$	1,758.09	\$	21,093.80
\$	90.45	\$	1,960.05	\$	23,517.00		\$	12.06	\$	84.42	\$	1,829.38	\$	21,949.20
\$	93.68	\$	2,029.94	\$	24,355.50		\$	12.49	\$	87.43	\$	1,894.61	\$	22,731.80
\$	96.90	\$	2,099.82	\$	25,194.00		\$	12.92	\$	90.44	\$	1,959.83	\$	23,514.40
\$	99.90	\$	2,164.83	\$	25,974.00		\$	13.32	\$	93.24	\$	2,020.51	\$	24,242.40
\$	102.60	\$	2,223.34	\$	26,676.00		\$	13.68	\$	95.76	\$	2,075.12	\$	24,897.60
\$	105.98	\$	2,296.48	\$	27,553.50		\$	14.13	\$	98.91	\$	2,143.38	\$	25,716.60
\$	108.98	\$	2,361.49	\$	28,333.50		\$	14.53	\$	101.71	\$	2,204.06	\$	26,444.60
\$	111.08	\$	2,407.00	\$	28,879.50		\$	14.81	\$	103.67	\$	2,246.53	\$	26,954.20
\$	114.60	\$	2,483.38	\$	29,796.00		\$	15.28	\$	106.96	\$	2,317.82	\$	27,809.60
\$	118.05	\$	2,558.14	\$	30,693.00		\$	15.74	\$	110.18	\$	2,387.60	\$	28,646.80
\$	121.88	\$	2,641.03	\$	31,687.50		\$	16.25	\$	113.75	\$	2,464.96	\$	29,575.00
\$	126.15	\$	2,733.67	\$	32,799.00		\$	16.82	\$	117.74	\$	2,551.43	\$	30,612.40
\$	129.98	\$	2,816.56	\$	33,793.50		\$	17.33	\$	121.31	\$	2,628.79	\$	31,540.60
\$	134.70	\$	2,918.95	\$	35,022.00		\$	17.96	\$	125.72	\$	2,724.35	\$	32,687.20
\$	140.03	\$	3,034.34	\$	36,406.50		\$	18.67	\$	130.69	\$	2,832.05	\$	33,979.40
\$	144.60	\$	3,133.48	\$	37,596.00		\$	19.28	\$	134.96	\$	2,924.58	\$	35,089.60
\$	149.18	\$	3,232.62	\$	38,785.50		\$	19.89	\$	139.23	\$	3,017.11	\$	36,199.80
\$	152.63	\$	3,307.38	\$	39,682.50		\$	20.35	\$	142.45	\$	3,086.89	\$	37,037.00
\$	156.23	\$	3,385.40	\$	40,618.50		\$	20.83	\$	145.81	\$	3,159.70	\$	37,910.60
\$	159.60	\$	3,458.53	\$	41,496.00		\$	21.28	\$	148.96	\$	3,227.96	\$	38,729.60
\$	163.13	\$	3,534.92	\$	42,412.50		\$	21.75	\$	152.25	\$	3,299.26	\$	39,585.00
Deiber			hu ratao are re	at a x=		م الب	~		 					
Daily, ı	monthly and	year	ly rates are n	ot exa	ct due to rour	ndin	g.							

ARTICLE 44 - DURATION AND NEGOTIATIONS

- 44.01 This Agreements shall become effective as of November 29, 1999 unless otherwise specified and shall remain in full force and effect until December 31, 2000 inclusive, and shall automatically be renewed for one (1) year unless written notice is given by either party to the other to negotiate a new Agreement or to terminate this Agreement. Such notice must be given not less than thirty (30) days nor more than ninety (90) days prior to the expiry date of this Agreement or any continuation thereof.
- 44.02 Where a notice to negotiate a new Agreement has been served, the first negotiating meeting shall be held within twenty (20) days following the date of service.
- 44.03 Where a notice to negotiate or notice to terminate has been served, this Agreement or any continuation thereof shall remain in full force and effect until the provisions of the Canada Labour Code have been met.
- 44.04 MEDIATION / CONCILIATION

If during negotiations for a Collective Agreement or revisions or renewal of this Agreement, the Parties fail to agree on the terms thereof, either Party may request the assistance of the conciliation or mediation service having legal jurisdiction over the matter.

BETWEEN

TELUS

AND

THE INTERNATIONAL BROTHERHOOD OF ELECTRICAL WORKERS

LOCAL 348

CLERICAL/OPERATOR SERVICES

LETTERS OF UNDERSTANDING/LETTERS OF INTENT

It is agreed by the Parties that all letters of understanding/letters of intent, that are issued as a result of Clerical/Operator Services Negotiations will remain in effect for the duration of this Agreement.

Note: Letters of Intent/Letters of Understanding will be printed at the back of the Collective Agreement.

Signed by:

Signed by:

L. Emanuel TELUS Grace Thostenson LOCAL 348 International Brotherhood of Electrical Workers

Dated: October 23, 1999 Dated: October 23, 1999

BETWEEN

TELUS

AND

THE INTERNATIONAL BROTHERHOOD OF ELECTRICAL WORKERS

LOCAL 348

CLERICAL/OPERATOR SERVICES EMPLOYEES

AND

CRAFT AND SERVICES EMPLOYEES

BENEFITS COMMITTEE

The Parties agree to form a committee to meet twice per year to discuss concerns and review information regarding benefit plans applicable to employees covered by the above agreements.

signed by:

Don Windsor TELUS signed by:

Mike Semeniuk International Brotherhood of Electrical Workers Local Union 348

Dated: March 28, 1999 Dated: March 28, 1999

BETWEEN

TELUS

AND

THE INTERNATIONAL BROTHERHOOD OF ELECTRICAL WORKERS

LOCAL 348

CLERICAL

COMPRESSED HOURS OF WORK PROGRAM INFORMATION TECHNOLOGY (COMPUTER OPERATORS)

The Company and the Union agree that within sixty (60) days following the date of ratification, or as otherwise mutually agreed, the Compressed Hours of Work Program Information Technology (Computer Operators) shall be addressed regarding integration into the IBEW 348 Clerical Compressed Work Week schedule or agree to develop an alternative if this is not feasible. The requirement of stand-by service and pay, shall also be reviewed and addressed.

Signed by:

Signed by:

L. Emanuel TELUS

Grace Thostenson LOCAL 348 International Brotherhood of Electrical Workers

Dated: October 23, 1999 Dated: October 23, 1999

BETWEEN

TELUS

AND

THE INTERNATIONAL BROTHERHOOD OF ELECTRICAL WORKERS

LOCAL 348

CLERICAL

COMPRESSED WORK WEEK

Where a clerical employee or group of employees volunteers and their manager agrees, the Company and Union agree that the employees may work one (1) of the following compressed work week schedules:

Clerical Compressed Work Week Scheduling Options

- 2. Each week containing three ten hour shifts and one seven and one-half hour shift. (10, 10, 10, 7.5)
- 3. Each week containing three nine and one-half hours shifts and one nine hour shift. (9.5, 9.5, 9.5, 9)
- 4. Each week containing three twelve and one-half hour shifts. (12.5, 12.5, 12.5). This option must be approved by both the Union and the Director of Employee Relations.
- Two (2) week rotation consisting of any combination of one (1) week containing three (3) eight and one-half (8.5) hour shifts plus two (2) eight (8) hour shifts, and one (1) week containing three (3) eight and one-half (8.5) hour shifts plus one (1) eight (8) hour shift. (8.5, 8.5, 8.5, 8.8, 8.5, 8.5, 8.5, 8)

The following provisions shall apply to clerical employees working any of the compressed work week options. These shall supersede or prevail where there may be a conflict with any other article of the collective agreement.

- Annual Vacation: Vacation credits will be expressed and taken in hours and paid based on hours. (Example: 15 vacation days will be credited as 15 x 7.5 = 112.5 hours)
- 2. Sick Leave Credits: Sick leave credits will be stated and paid in hours. (Example: 20 days will be credited as 20 x 7.5 = 150 hours). An employee scheduled to work a ten (10) hour shift but who is absent because of illness will be paid for ten (10) hours from their sick leave credits. Employees working a compressed work week schedule shall have their sick leave credits reestablished based on hours. (Example: 487.5 hours is the equivalent of 65 working days and 165 hours is the equivalent of 22 working days.)
- 3. Holidays: Employees will be paid holiday pay for each hour they would have been scheduled to work if the day had not been a holiday. If the holiday falls on a day that the employee is not scheduled to work, the work day in the same week which is closest to the actual holiday should be scheduled as the day in lieu of the holiday. (Example: If the holiday falls on a Saturday and the employee's last scheduled work day that week is Thursday, the holiday lieu day would be scheduled for the Thursday.) Employees will be paid at two times their basic hourly rate for all hours including overtime worked on a holiday.
- 4. Overtime: Overtime will be paid for hours worked in excess of the employee's scheduled daily hours or those worked on a day off.
- 5. Meal Periods: Shifts of 11 hours or more shall include 2 paid twenty minute meal periods: those that are less than 11 hours will have one unpaid meal period of not less than thirty minutes or longer than one hour.
- 6. Bereavement Leave: Employees may take up to five working days for bereavement leave; paid bereavement leave is limited to a maximum of 37.5 hours.
- 7. Training Courses: An employee working option 2, 3, 4, who is scheduled to take three or more days of training in a week will be scheduled for five consecutive days of 7.5 hours each that week. For employees working option 1 or 5, there is no change in scheduled hours unless the training is for five days and coincides with the four day work week. In this case the employee's four day week should be scheduled to another week within the rotation.
- 8. Schedule Posting: Tour schedules will be posted at least five weeks in advance of the day they are to become effective.

- 9. Days Off: Additional days off resulting from the implementation of a compressed work week schedule are to be scheduled consecutively with other days off unless a <u>proven</u> business need exists or the employee requests it on another day.
- 10. Change of Tour: Employees will be eligible for a change of tour payment if their schedule is changed without seven days' prior notice.
- 11. Exchange of Tour: A qualified employee may work a tour on behalf of another employee if approved by the manager and at no additional cost to the Company.
- 12. Termination of CWW Schedule: Either an employee or manager can revert to a five day per week, seven and one-half hours of work per day schedule if four weeks notice of termination has been given from one to the other.

Signed by:

Signed by:

L. Emanuel TELUS Grace Thostenson LOCAL 348 International Brotherhood of Electrical Workers

Dated: October 23, 1999 Dated: October 23, 1999

BETWEEN

TELUS

AND

THE INTERNATIONAL BROTHERHOOD OF ELECTRICAL WORKERS

LOCAL 348

ELECTRONIC MONITORING

A Clerical employee will be notified prior to any electronic monitoring of their work unless the employee advises in writing that they do not want to be notified of such monitoring.

Signed by:

Signed by:

L. Emanuel TELUS Grace Thostenson LOCAL 348 International Brotherhood of Electrical Workers

Dated: October 23, 1999

BETWEEN

TELUS

AND

THE INTERNATIONAL BROTHERHOOD OF ELECTRICAL WORKERS

LOCAL 348

CLERICAL/OPERATOR SERVICES

FAMILY DAYS

The Union and the Company agree that for the balance of 1999 employees who were governed by the Civic Service Union 52 Collective Agreement with TELUS Communications (Edmonton) Inc. under their June 19, 1997 – December 18, 1998 Collective Agreement, shall be entitled to Family Days as provided therein. Subsequent to that, these employees shall be eligible for Family Days (Clerical) as indicated in Article 18 of this Collective Agreement.

Signed by:

Signed by:

L. Emanuel TELUS Grace Thostenson LOCAL 348 International Brotherhood of Electrical Workers

Dated: October 23, 1999

BETWEEN

TELUS

AND

THE INTERNATIONAL BROTHERHOOD OF ELECTRICAL WORKERS LOCAL 348

CLERICAL/OPERATOR SERVICES

JOB ALLOCATION APPEAL PROCEDURE

I. JOB ALLOCATION

The Parties agree that the following procedure will be followed when excluded jobs at or below MP (Management or Professional) Pay Band 2 (or equivalent) are created or reclassified to a new pay band at or below Pay Band 2 (or equivalent):

- 1. The Employer will notify the Union in writing when a job has been created or reclassified within fifteen (15) days of the date that the job is first filled or reclassified. The notice will include the name(s) and locations of the current incumbent(s) and the date filled or reclassified.
- 2. Within thirty (30) days of the notification, the Employer will provide the Union with a copy of the Job Description.
- 3. If the Union disputes the exclusion of the job, it will be given an opportunity to interview any incumbent of the job regarding the nature of their duties if:
 - (a) the Union has given the Employer written notice that it disputes the exclusion and,
 - (b) not less than thirty (30) and not more than seventy-five (75) days have passed since notification;

- 4. If the Union continues to dispute the exclusion of the job, it must notify the Employer in writing not later than seventy-five (75) days after notification. The Parties will then schedule a meeting within one hundred five (105) days of notification to attempt to agree on the future status of the job;
- 5. After meeting, if the Parties are not able to agree on the status of the job, the Union may refer the dispute to the Job Allocation Appeal Board for a final and binding determination of whether the job will continue to be excluded or is to be included in the Union's bargaining unit;
- 6. The Job Allocation Appeal Board shall be composed as follows:
 - (a) One Union representative who is a member of the bargaining unit or who is employed by the Union;
 - (b) One Employer representative who is employed by the Employer; and
 - (c) The Chairperson who shall be agreed upon by the Union's and the Employer's representatives;
- 7. If the Union wishes to refer a job dispute to the Job Allocation Appeal Board it shall, within one hundred fifteen (115) days of notification, notify the Employer in writing of its intent to do so and of the name of its representative on the Board;
- 8. On receiving notice of a reference to the Job Allocation Appeal Board, the Employer shall advise the Union of its representative to the Board within seven (7) days;
- 9. Unless agreed otherwise, the Job Allocation Appeal Board may conduct its own investigation and shall hold a hearing within sixty (60) days of the Union's reference of the dispute and will hear such evidence as the Parties may desire to present to assure a full, fair hearing;
- 10. In making its determination, the Job Allocation Appeal Board; shall consider first whether the job is occupied by an "employee" within the meaning of Part 1 of the <u>Canada Labour Code</u> and, if the job is occupied by an "employee", whether it shares a greater community of interest with jobs included in the Union's bargaining unit, or with excluded jobs;

- 11. The Job Allocation Appeal Board shall render a decision in writing to the Parties within thirty (30) days after the hearing advising them of whether the job is to continue to be excluded or is to be included within the Union's bargaining unit. The Chairperson shall have the authority to render the decision with the concurrence of either of the other members and a decision then rendered shall be final and binding on the Parties. If no majority decision is reached, then the Chairperson shall render the final and binding decision;
- 12. If the Job Allocation Appeal Board rules that the job falls within the scope of the Union's bargaining unit, it shall be treated as a new job effective the beginning of the next pay period following the date of publication of the Board's decision and the appropriate wage rate shall be determined pursuant to the Job Evaluation system under the applicable Collective Agreement or, in the absence of such a system, the Board shall determine which existing bargaining unit classification the job most closely resembles and that classification will not preclude the Parties from negotiating a new rate for the job;
- 13. Unless waived by written agreement, the time limits under this procedure are mandatory and either party failing to comply with those limits shall be deemed to have accepted the position of the other party. All references to "days" are to calendar days.
- 14. Each of the Parties shall bear the expense of its representative on the Job Allocation Appeal Board. The fees and expenses of the Chairperson shall be borne equally by the Parties.

Signed by:

Signed by:

L. Emanuel TELUS Grace Thostenson LOCAL 348 International Brotherhood of Electrical Workers

Dated: October 23, 1999

BETWEEN

TELUS

AND

THE INTERNATIONAL BROTHERHOOD OF ELECTRICAL WORKERS LOCAL 348

CLERICAL/OPERATOR SERVICES

JOB SHARING

- I. Job Sharing
 - This letter of understanding establishes provision for either two Clerical or two Operator Services employees to voluntarily "Job Share" a full-time position. Job Sharing is an alternative work arrangement whereby the duties and responsibilities of a full-time position may be restructured, in a manner that would accommodate the employment of two employees to fill a full-time position. A job sharing arrangement between a Clerical and an Operator Services employee will not be permitted.
 - 2. A "Job Sharing Arrangement" refers to a specific written agreement setting out the names of the participants, the position to be shared, how duties will be divided, and the length of time agreed upon for the trial period.
 - 3. No "Job Sharing Arrangement" shall be entered into by any employees or party which conflicts with the terms of the collective agreement between the parties named above.
 - 4. No more than five (5) percent of regular full-time positions covered by the Clerical/Operator Services Agreement may be job shared at any given time.
 - 5. Employees qualified to enter a job share arrangement shall be from the same collective agreement.
 - 6. Employees cannot enter a job share arrangement for a position which has a higher pay group.

- 7. Job share opportunities shall not be posted in accordance with the posting provisions of the collective agreement.
- 8. The contents of this Letter of Understanding may be revised at anytime by agreement of the parties hereto.
- II. Participation/Eligibility
 - 1. Participation of employees under these provisions shall be voluntary and no employee shall be denied the opportunity to volunteer for a job sharing arrangement.
 - 2. Participation shall be limited to existing regular full or regular part-time employees who have completed their probationary period.
 - 3. Employees wishing to establish a job sharing arrangement must submit a written proposal to management with a copy to the Union outlining the reasons for the arrangement, its feasibility, and operating details necessary to implement it such as commencement date, days and hours to be worked, absence and vacation coverage and sharing of responsibilities.
 - 4. Employees participating in a job share arrangement will be re-classified to regular part-time-job share (RPT-JS).
- III. Maintenance of Full Time Positions
 - 1. Shared positions shall, in all respects with the exception that they are held by two individuals, be treated as though they were single positions with regard to scheduling and job descriptions.
 - 2. Upon expiry of the "Job Sharing Arrangement" the shared position will revert to full-time with the incumbent who was in the position before the job share arrangement started having ownership to the position.
- IV. Approval
 - 1. Any "Job Sharing Arrangement" shall be subject to approval by management.
- V. Schedules and Job Descriptions
 - 1. A work schedule will be set out in advance showing days, hours or tours to be worked and not worked by the job sharing partners.

- 2. Job Descriptions for the job sharing partners will be identical unless otherwise approved vis-a-vis Job Sharing 1.
- 3. The employer agrees not to increase production levels expected of job sharers for the sole reason that the position is shared.
- VI. Benefits
 - 1. An employee entering into a job shared position shall have benefits as provided elsewhere under this agreement pro-rated as applicable to part-time employees.
 - 2. Previously accumulated leave credit will not be effected.
- VII. Sickness/Vacation Relief
 - 1. Job sharing partners have the option of providing each other with sickness or vacation relief.
 - 2. A job partner shall not be required to provide relief on more than three (3) consecutive days for which they were not scheduled to work or more than six (6) days in one month for which they are not scheduled to work.
- VIII. Training
 - 1. Job sharers will have the same access to training required to perform their regular duties as other full-time employees under the collective agreement.
- IX. Evaluation
 - 1. In addition to other provisions set out in this agreement respecting performance evaluation, job sharing employees may request a separate evaluation of both shared and individual performance.

- X. Termination of Job Share Arrangement
 - 1. A "Job Sharing Arrangement" may be terminated upon 30 days notice for any of the following reasons:
 - (a) The employment of a job sharing partner terminates.
 - (b) The employer serves written notification withdrawing its approval of the job share arrangement to the employees with a copy to the Union.
 - 2. An employee who no longer wishes to continue their job sharing arrangements must successfully apply on a job posting to exit from the job share arrangement.
 - 3. Where the employment of a job sharing partner terminates or is about to terminate or for any other reason one partner is unable to continue the arrangement, the remaining partner will become the full-time incumbent until a suitably qualified individual is located to continue in another job sharing arrangement.
 - 4. Where the employer or Union serves notice to terminate the job share arrangement, both employees shall revert to their former or equivalent positions.
- XI. Information
 - 1. Written information materials, subject to their approval by the parties to this agreement, shall be made freely available to all employees.
- XII. Termination of Letter of Understanding

This Letter of Understanding shall be terminated thirty days following written notification by either party to the other of its intent to terminate this Letter of Understanding.

Signed by:

Signed by:

L. Emanuel TELUS Grace Thostenson LOCAL 348 International Brotherhood of Electrical Workers

Dated: October 23, 1999

BETWEEN

TELUS

AND

THE INTERNATIONAL BROTHERHOOD OF ELECTRICAL WORKERS LOCAL 348

CLERICAL/OPERATOR SERVICES

JOINT COMMITTEE GRIEVANCE REVIEW

The Employer and the Union agree that within sixty (60) days of ratification of the collective agreement they will meet to review the potential establishment of a Joint Committee for Grievance Review.

The responsibilities of the Committee would include the joint review of all grievances within sixty (60) days of the decision at Step 3 (before filing to Arbitration) for the purposes of attempting settlement or narrowing the issues advanced to Arbitration. All available information pertinent to the grievance would be shared, however discussions will be privileged and not to be introduced at Arbitration.

Signed by:	Signed by:
L. Emanuel TELUS	Grace Thostenson LOCAL 348
	International Brotherhood of Electrical Workers

Dated: October 23, 1999

BETWEEN

TELUS

AND

THE INTERNATIONAL BROTHERHOOD OF ELECTRICAL WORKERS LOCAL 348

CLERICAL/OPERATOR SERVICES

MEDICAL AND COMPASSIONATE TRANSFERS

Medical and Compassionate transfer requests will be sent to the Director, **Labour** Relations. A sub-committee of the Company and Union will review the transfer requests on the basis of:

> Compassionate reasons, or Medical reasons, or Isolation reasons.

Subject to mutual agreement between the Management and Union representatives on the sub-committee, an employee who has requested a transfer for any of the above reasons may be placed in a vacancy without the vacancy being posted. Any expenses incurred as a result of the transfer will be paid by the Employer in accordance with the relocation policy.

If an employee's transfer request cannot be accommodated, the employee will be informed in writing.

Signed by:

L. Emanuel TELUS Signed by:

Grace Thostenson LOCAL 348 International Brotherhood of Electrical Workers

Dated: October 23, 1999

BETWEEN

TELUS

AND

THE INTERNATIONAL BROTHERHOOD OF ELECTRICAL WORKERS LOCAL 348

CLERICAL

MICROCOMPUTER CONSULTANT POSITIONS - PLAnet HELP DESK

It is agreed by the parties that the terms and working conditions of Microcomputer Consultants working in the PLAnet Help Desk area will be addressed within sixty (60) days of ratification of this Collective Agreement, or as the parties may mutually agree. Should the parties fail to reach an agreement, the Microcomputer Consultants shall be covered by the terms of settlement reached through the 1999 Clerical / Operator Services Negotiations.

Signed by:

Signed by:

L. Emanuel TELUS

Grace Thostenson LOCAL 348 International Brotherhood of Electrical Workers

Dated: October 23, 1999

BETWEEN

TELUS

AND

THE INTERNATIONAL BROTHERHOOD OF ELECTRICAL WORKERS LOCAL 348

CLERICAL

OCCASIONAL EMPLOYEES - RESIDENTIAL SALES & SERVICE

Occasional employees are employees employed on an hourly or daily basis, in accordance with Article 3, to cover situations that cannot reasonably be handled by Regular Full-Time, Regular Part-Time or Casual employees. Occasional employees may be hired to cover peak activity periods as they relate to the business needs of the department. In Residential Sales & Service the employer identifies that those times <u>typically</u> are:

- Mondays and Fridays
- Evenings and weekends
- First week of the month
- Last week of the month
- 14th, 15th and 16th of the month
- Month of September

Occasional employees are expected to be available for work at least two (2) weekends per month and a minimum of three (3) days per week when required. Failure to meet these availability requirements or declaring "Not Available" for specific days or periods, on a continuous basis without an acceptable reason, may be considered by the employer as "not meeting the business needs of the company" and shall be dealt with accordingly.

Residential Sales & Service Trial

The Union and the Employer agree to trial the following exceptions for the Residential Sales & Service call centre occasional employees:

- Tour schedules shall be posted a minimum of two (2) weeks in advance of the day they are to become effective. Occasional employees are to identify their availability (i.e. number of days available per week and specific days or times not available) one (1) week prior to the schedule being posted. Occasional employees who are scheduled or called in shall not have their scheduled or call in tours cancelled. Once an occasional employee is scheduled, or accepts a call in, they are responsible to work the tour or to arrange a trade.
- A tour may be changed with a minimum of seven (7) days' notice as per Clause 16.06. For unexpected training or due to customer demand, the tour may be changed with a minimum of three (3) days' notice, however this change shall not alter the tour start by more than two (2) hours.
- A day shall mean the twenty-four (24) hour period beginning at midnight on one day and ending on the following midnight. Clause 16.07 shall not apply. On any given day occasional employees shall be required to work only one tour. There shall be a minimum of ten (10) hours between the end of a tour and the start of the next tour. If ten (10) hours is not provided, then overtime shall be paid for the entire tour.

The trial shall be for a period of twelve (12) months. This shall not preclude the parties from jointly addressing concerns identified during the trial. Should unforeseen difficulties arise, either party may provide thirty (30) days notice in writing to cancel the trial.

Signed by:

Signed by:

L. Emanuel TELUS

Grace Thostenson LOCAL 348 International Brotherhood of Electrical Workers

Dated: October 23, 1999

BETWEEN

TELUS

AND

THE INTERNATIONAL BROTHERHOOD OF ELECTRICAL WORKERS LOCAL 348

REGULAR PART-TIME, CASUAL AND OCCASIONAL EMPLOYEES

TELUS will provide to the IBEW **separate reports containing** information regarding the above mentioned **Clerical and Operator Services** employees as follows:

- 1. The information will be forwarded every second pay period;
- 2. employees will be identified by position **number**;
- 3. employees will be identified by personnel subgroup (classification);
- 4. employees will be identified by name;
- 5. employees will be identified by pay scale group;
- 6. employees will be identified by description of job (job title / number);
- 7. the number of hours worked by each employee in each pay period will be identified;
- 8. employee's start date will be identified;
- 9. the information will continue to be provided on a moving twenty-six pay period basis;
- 10. work schedules will be provided upon request;
- 11. where possible the company will provide the information in soft copy;

12. the format of the reports shall continue to be as are currently provided or as the parties may agree.

Signed by:

Signed by:

L. Emanuel TELUS Grace Thostenson LOCAL 348 International Brotherhood of Electrical Workers

Dated: October 23, 1999

BETWEEN

TELUS

AND

THE INTERNATIONAL BROTHERHOOD OF ELECTRICAL WORKERS LOCAL 348

CLERICAL AND OPERATOR SERVICES

SALES INCENTIVE PLAN

The November 17, 1998 / December 8, 1998 Letter of Understanding regarding the Sales Incentive Program for Customer Service Representatives in Residential Services shall be extended to December 31, 1999.

Signed by:

Signed by:

L. Emanuel TELUS

Grace Thostenson LOCAL 348 International Brotherhood of Electrical Workers

Dated: October 23, 1999

BETWEEN

TELUS

AND

THE INTERNATIONAL BROTHERHOOD OF ELECTRICAL WORKERS LOCAL 348

CLERICAL AND OPERATOR SERVICES

SAP ADMINISTRATION

To support a successful conversion process for TELUS' former administrative systems which accrued and decremented in days, to the SAP format which accrues and decrements in hours, the Company and the Union recognize that unexpected situations may arise where existing Collective Agreement provisions may not align with SAP system parameters. In seeking resolution to these issues, the Company and the Union agree that the negotiated terms and conditions of employment shall be preserved unless the parties expressly negotiate otherwise. To this end, the parties shall address the situations listed below as follows:

Furlough Adjustment

• Employees taking furlough in increments of less than one day shall reestablish sick leave credits as if attending a complete day of work. The Company will pursue a specific code to designate those circumstances when employees are coded for approved furlough periods of less than a day.

Temporary Movement between Clerical and Operator Services

- Employees in temporary assignments shall accrue vacation credits based on the position they are temporarily assigned to. Additional vacation accrued in a temporary assignment position shall be paid out when the employee returns to their permanent position, if it is not possible for the employee to take the vacation time off. If the temporary assignment position accrues vacation at a lower rate, an adjustment will be made to the employee's vacation upon return to their permanent position to ensure the employee has their full entitlement in accordance with Article 21.
- Employees requesting vacation while in a temporary assignment, shall take vacation at the rate and hours of the temporary assignment position (example: an Operator on loan to a clerical position will receive the clerical hourly rate and seven and one half (7.5) hours of vacation for each day taken). Upon return to the operator position, the employee will resume the operator hourly rate and if any vacation was taken while in the clerical position, their entitlement will be adjusted to reflect that time as if it were taken at the operator entitlement and rate.

Sick Leave for Temporary Movement between Clerical and Operator Services

• In accordance with Article 28, employees in temporary assignments shall reestablish and draw down sick leave credits based on the hours of work for the temporary assignment position (example: an Operator who becomes ill while assigned to a clerical position shall have seven and one-half (7.5) hours of sick leave decremented). Upon return to their regular position, the employee shall have their sick leave entitlement adjusted to ensure their entitlement is no less than had they taken the sick leave in their regular position.

Accident Compensation

• In accordance with Article 27, employees in temporary assignments shall receive the accident compensation credits corresponding to the position they are working in at the time the injury occurs. In the event the employee returns to their regular position and is again absent due to a related or non-related accident, a continuing accident claim will be presumed in accordance with Article 27.04. In that case, the appropriate accident compensation credits and position they would be ascribed to will be determined through the Workers' Compensation Board's assessment of the claim.

Permanent Movement Between Clerical and Operator Services

- For those circumstances where there are permanent movements between Operator Services and Clerical, an adjustment will be made to the employee's vacation entitlement as follows:
 - Clerical employees moving to Operator Services will receive the additional vacation earned as a result of working seven and one-half (7.5) hours per day.
 - Operator Services employees moving to Clerical, will have an adjustment made to credit them with the equivalent vacation they would have been entitled to had they worked a seven and one-half (7.5) hour day.

Regular Part-time Vacation

Regular Part-time employees shall accrue paid vacation based on the following formula:

<u>Hours Paid</u> = Percent of full time vacation accrual Hours available (full time equivalent)

Example (Clerical):

1170	= 60% of full time paid vacation
1950	(as per Article 21)

• The employee will be able to designate the days or weeks which will be paid or unpaid. The regular rate of pay for vacation accrual will be based on the entitlements set out in Article 21.

Should any unforeseen difficulties arise, the parties shall jointly address the situation with the guiding principle being that employees shall not be negatively impacted due to system requirements and / or limitations.

Signed by:

Signed by:

L. Emanuel TELUS

Grace Thostenson LOCAL 348 International Brotherhood of Electrical Workers

Dated: October 23, 1999

BETWEEN

TELUS

AND

THE INTERNATIONAL BROTHERHOOD OF ELECTRICAL WORKERS LOCAL 348

OPERATOR SERVICES

SERVICE ASSISTANT (SPECIAL SERVICES), CUSTOMER SERVICE ASSISTANT AND SERVICE ASSISTANT AND SUPPORT DUTIES

It is agreed that the Service Assistant (Special Services), Customer Service Assistant and Service Assistant and Support responsibilities do not include formal disciplining of employees or monitoring for the purpose of evaluation.

Signed by:

Signed by:

L. Emanuel TELUS

Dated: October 23, 1999 Grace Thostenson LOCAL 348 International Brotherhood of Electrical Workers

BETWEEN

TELUS

AND

THE INTERNATIONAL BROTHERHOOD OF ELECTRICAL WORKERS LOCAL 348

CLERICAL/OPERATOR SERVICES

SEVERANCE PAY

Regular employees who are unable to accept a permanent redeployment to a position in another headquarters will be eligible to receive severance pay in accordance with either Clerical Article 33.08 or Operator Services Article 34.08.

Note: Where the Company and Union agree to an alternative severance payment, it shall supersede this Letter of Understanding. Under no circumstances shall both severance payments apply.

Signed by:

Signed by:

L. Emanuel TELUS Grace Thostenson LOCAL 348 International Brotherhood of Electrical Workers

Dated: October 23, 1999

BETWEEN

TELUS

AND

THE INTERNATIONAL BROTHERHOOD OF ELECTRICAL WORKERS LOCAL 348

CLERICAL / OPERATOR SERVICES

SUPPLEMENTARY HEALTH AND VISION CARE PLANS

Effective thirty (30) days following the date of ratification, the company shall amend the Supplementary Health Plan and the Vision Care Plan to include provisions as follows:

Supplementary Health Plan

- Usual and reasonable charges for the supplies required for the administration of insulin (syringes and needles) and testing materials used by diabetics, upon written order of a physician.
- The Plan shall pay fifty percent (50%) of the costs of purchase and repair of hearing aids as prescribed by a physician to a maximum of five hundred dollars (\$500.00) in any five (5) consecutive calendar year period. Maintenance, batteries and recharging devices are excluded.

Signed by:

Signed by:

L. Emanuel TELUS Grace Thostenson LOCAL 348 International Brotherhood of Electrical Workers

Dated: October 23, 1999

BETWEEN

TELUS

AND

THE INTERNATIONAL BROTHERHOOD OF ELECTRICAL WORKERS LOCAL 348

CLERICAL/OPERATOR SERVICES

TAXIS

The employer will pay for taxi service within city limits where available for tours starting or ending between 11:00 P.M. and 6:45 A.M. inclusive.

Signed by:

Signed by:

L. Emanuel TELUS Grace Thostenson LOCAL 348 International Brotherhood of Electrical Workers

Dated: October 23, 1999

BETWEEN

TELUS

AND

THE INTERNATIONAL BROTHERHOOD OF ELECTRICAL WORKERS LOCAL 348

CLERICAL/OPERATOR SERVICES

Telecommunications Concession

The Company will provide all Clerical and Operator Services employees their choice of up to five (5) custom calling features (Smart Touch Services) per month at no cost to the employee.

To be eligible for the telecommunications concessions, the employee must use TELUS as their telecommunications provider.

Signed by:

Signed by:

L. Emanuel TELUS

Dated: October 23, 1999 Grace Thostenson LOCAL 348 International Brotherhood of Electrical Workers

BETWEEN

TELUS

AND

THE INTERNATIONAL BROTHERHOOD OF ELECTRICAL WORKERS LOCAL 348

OPERATOR SERVICES

TRADING OF TOURS BETWEEN SERVICES

- 1. Trading will be allowed provided the two (2) employees work in the same headquarters area.
- 2. Trading is subject to Supervisory approval. Such approval shall not be unreasonably withheld.
- 3. There shall be no restrictions on the number of trades.

The current practice of trading of tours will not be affected by the above.

Signed by:

Signed by:

L. Emanuel TELUS Grace Thostenson LOCAL 348 International Brotherhood of Electrical Workers

Dated: October 23, 1999

LETTER OF UNDERSTANDING BETWEEN TELUS AND THE INTERNATIONAL BROTHERHOOD OF ELECTRICAL WORKERS LOCAL 348 CLERICAL/OPERATOR SERVICES

VACATION ENTITLEMENT

The Company and the Union agree that employees formerly governed by the Civic Service Union 52 Collective Agreement shall retain the vacation entitlement earned under that Agreement (including any milestone attained in 1999) until the end of 1999. Thereafter, this entitlement will be frozen, and any increased vacation entitlement will be aligned with the service requirements set out in the IBEW collective agreement.

Example: An employee with a vacation entitlement of 25 days (at 17 years) will retain this entitlement until they reach the next milestone (30 days at 25 years) rather than at 23 years, as provided for in the Civic Service Union 52 collective agreement.

Signed by:

Signed by:

L. Emanuel TELUS

Grace Thostenson LOCAL 348 International Brotherhood of Electrical Workers

Dated: October 23, 1999

BETWEEN

TELUS

AND

THE INTERNATIONAL BROTHERHOOD OF ELECTRICAL WORKERS LOCAL 348

CLERICAL AND OPERATOR SERVICES

WORK EXPERIENCE PROGRAMS

It is agreed by the parties that TELUS may participate in Work Experience Education Programs and that the wages or compensation and working conditions of the individuals participating in such programs shall be determined by TELUS, the Union, the applicable School Board and the affected individual (or guardian).

It is further agreed that the participation of such individuals shall in no way affect the job security of employees falling within the scope of this Agreement.

TELUS shall advise the Union of those individuals participating in this program prior to their actual commencement.

Signed by:

Signed by:

L. Emanuel TELUS

Grace Thostenson LOCAL 348 International Brotherhood of Electrical Workers

Dated: October 23, 1999

APPENDIX A CLERICAL

Supplementary Appendix to, and not Constituting a Part of the Clerical Employees' Agreement.

TRAVEL AND TRANSFER EXPENSES

- 1. GENERAL
- 1.01 This Appendix is issued to advise those concerned of the regulations and allowances covering employees travelling within or out of the Province to attend Company sponsored or authorized courses, or on transfer.
- 2. TRAVEL AND EXPENSES OUTSIDE THE PROVINCE
- 2.01 (a) An employee who has obtained authorization to travel out of the Province, will be allowed reasonable and actual expenditures for lodging, meals and miscellaneous expenses approved by Management.
 - (b) Prior to the employee's departure, the Department concerned shall establish guidelines for reasonable expenses, taking into account the location, length of stay, etc. Receipts are required for lodging.
- 2.02 (a)Transportation type and method will be determined and authorized by the Department approving the expenditure.
 - (b) An employee authorized to use their own vehicle, on a route approved by the Department, will be allowed the actual distance driven at the prevailing rates.
- 3. TRAVEL AND EXPENSES WITHIN THE PROVINCE
- 3.01 The Department approving the expenditure will determine the type and method of transportation for employees attending Company sponsored or authorized courses.
- 3.02 An employee authorized to use their own vehicle will be allowed actual kilometerage at the prevailing rate provided there is proper insurance coverage on the vehicle being used.
- 3.03 Employees will be allowed necessary meals while travelling to a course on Sunday.
- 3.04 Employees on Employer administered courses away from headquarters shall be allowed meal and incidental expenses as per the Agreement.

- 3.05 Prior to the employee's departure, the Department concerned shall establish the guidelines for reasonable lodging expenses that will be allowed, taking into account the location of the course. Receipts are required for lodging.
- 3.06 Out-of-town employees who cannot obtain accommodation within walking distance of where the course is being held may claim **taxi fare (receipt required).**
- 3.07 An employee receiving a class instructor's differential may be allowed necessary expenses as approved by their Employer.
- 3.08 Employees attending Company administered courses within a **three hundred and twenty (320) kilometre** radius of their headquarters may be allowed to come home every weekend at the discretion of their Supervisor.
- 3.09 Employees on courses
 - (a) If possible, the Supervisor will grant the employee an equivalent number of hours off during the work day preceding the course for travelling time.
 - (b) If the Supervisor decides, for the good of service, that time off should not be given, then payment for actual travel time will be made.
- 4. TRANSFER ALLOWANCES AND EXPENSES
- 4.01 When employees are successful on job postings or are transferred within the Province from one headquarter to another at the request of the Company and a move of residence is required, the costs associated with the move shall be borne by the Company as per the Relocation Policy.
- 4.02 Allowances, transportation and moving costs are not paid by the Company when an employee moves at their request.
- 4.03 Headquarters for the purpose of claiming transfer allowance and expenses will be the following metropolitan areas:
 - (a) City of Edmonton: City limits and includes Sherwood Park, St. Albert and Namao.
 - (b) All other Cities and Towns: City or Town limits.
- 4.04 Current transfer allowances and moving costs paid by the Company for Management or Special Skills Personnel moves or transfers will be applicable to Employer requested moves and employees successful on job postings.

5. EMPLOYEES TRAVELLING TO AND FROM OUT-OF-TOWN JOBS

- 5.01 Travelling employees is a term defining employees who spend all or the greater portion of their time in work locations outside their defined headquarters. It is not applicable when such employees are assigned to work wholly located within their respective headquarters.
- 5.02 The immediate Supervisor shall consider the following factors in order to reach a decision on whether or not daily or weekend travel is to be allowed:
 - (a) Distance from headquarters
 - (b) Road and weather conditions
 - (c) Nature and duration of the job
 - (d) Overtime requirements, if any
 - (e) Availability of transportation
 - (f) Accommodation and eating facilities
 - (g) Employee or group requirements as they may relate to reasonable circumstances
- 5.03 Travelling time when starting or completing a job in a town or district away from headquarters will be on Company time. If travelling extends into the employee's own time, the overtime rate will prevail.
- 5.04 Travelling to and from headquarters on a daily basis may be authorized by the immediate Supervisor. Where daily travel is authorized, the Company will allow the employee to travel to and from the job site on Company time.
- 5.05 Travelling employees working within a **three hundred and twenty (320) kilometre** radius of their headquarters may be allowed to return every weekend, subject to the conditions of Clause 5.02. Where weekend travel is authorized, the employee shall travel to and from the job site on Company time.
- 5.06 The Supervisor shall determine whether an employee remaining at the job site over the weekend, while others are returning home, shall be paid meals and lodging.
- 5.07 Travelling employees working outside a **three hundred and twenty (320) kilometre** radius of headquarters shall be allowed to return at least every second weekend. Where such travel is authorized, the Company will allow the employee to travel to and from the job site on Company time.
- 6. TRANSPORTATION

- (a) For the purpose of conducting Company business, transportation will be furnished or allowed all employees covered by this Agreement where same is considered necessary by the Company. It is understood and agreed that an employee cannot be required to use their personal automobile as a condition of employment; the use of employee's personal automobile is strictly at the employee's option.
- (b) Management will determine the type and method of transportation by bus, rail, car or plane and whenever possible, transportation will be in a Company vehicle.
- (c) An employee, who is authorized by Management to use their own vehicle will be allowed actual car kilometerage at the prevailing rate per kilometre providing there is proper insurance coverage on the vehicle being used.
- (d) Employees authorized to travel by private automobile will be covered by compensation, providing there is no deviation from the direct route to headquarters or any unauthorized stopovers enroute.
- (e) When two (2) or more employees are travelling in the same private automobile, the owner is the only person eligible to claim kilometerage.
- (f) Any claim arising from an accident to an employee's own vehicle will be the responsibility of the said employee.
- (g) Where Company vehicles are not available, Supervisors may arrange for the use of private automobiles (with employee's consent) **or alternate transportation** to transport employees to and from headquarters.
- (h) An employee who resigns, is dismissed or laid off in the field will be furnished or allowed transportation to their district headquarters.

Signed by: L. Emanuel TELUS

Signed by: Grace Thostenson LOCAL 348 International Brotherhood of Electrical Workers

Dated: October 23, 1999

APPENDIX B

OPERATOR SERVICES

Supplementary Appendix to, and not Constituting a Part of the Operator Services Employees' Agreement.

TRAVEL AND TRANSFER EXPENSES

- 1. GENERAL
- 1.01 This Appendix is issued to advise those concerned of the regulations and allowances covering employees travelling within or out of the Province to attend Company sponsored or authorized courses, or on transfer.
- 2. TRAVEL AND EXPENSES OUTSIDE THE PROVINCE
- 2.01 (a) An employee who has obtained authorization to travel out of the Province, will be allowed reasonable and actual expenditures for lodging, meals and miscellaneous expenses approved by Management.
 - (b) Prior to the employee's departure, the Department concerned shall establish guidelines for reasonable expenses, taking into account the location, length of stay, etc. Receipts are required for lodging.
- 2.02 (a) Transportation type and method will be determined and authorized by the Department approving the expenditure.
 - (b) An employee authorized to use their own vehicle, on a route approved by the Department, will be allowed the actual distance driven at the prevailing rates.

3. TRAVEL AND EXPENSES WITHIN THE PROVINCE

- 3.01 The Department approving the expenditure will determine the type and method of transportation for employees attending Company sponsored or authorized courses.
- 3.02 An employee authorized to use their own vehicle will be allowed actual kilometerage at the prevailing rate provided there is proper insurance coverage on the vehicle being used.
- 3.03 Employees will be allowed necessary meals while travelling to a course on Sunday.

- 3.04 Employees on Employer administered courses away from headquarters, shall be allowed meal and incidental expenses as per the Agreement.
- 3.05 Prior to the employee's departure, the Department concerned shall establish the guidelines for reasonable lodging expenses that will be allowed, taking into account the location of the course. Receipts are required for lodging.
- 3.06 Out-of-town employees who cannot obtain accommodation within walking distance of where the course is being held may claim **taxi fare (receipt required).**
- 3.07 An employee receiving a class instructor's differential may be allowed necessary expenses as approved by their Employer.
- 3.08 Employees attending Company administered courses within a **three hundred and twenty (320) kilometre** radius of their headquarters, may be allowed to come home every weekend at the discretion of their Supervisor.
- 3.09 Employees on courses
 - (a) If possible, the Supervisor will grant the employee an equivalent number of hours off during the work day preceding the course for travelling time.
 - (b) If the Supervisor decides, for the good of service, that time off should not be given, then payment for actual travel time will be made.
- 4. TRANSFER ALLOWANCES AND EXPENSES
- 4.01 When employees are successful on job postings, or are transferred within the Province from one headquarter to another at the request of the Company and a move of residence is required, the costs associated with the move shall be borne by the Company as per the Relocation Policy.
- 4.02 Allowances, transportation and moving costs are not paid by the Company when an employee moves at their request.

- 4.03 Headquarters for the purpose of claiming transfer allowances and expenses will be the following metropolitan areas:
 - (a) City of Edmonton: City limits and including Sherwood Park, St. Albert and Namao.
 - (b) All other Cities and Towns: City or Town limits
- 4.04 Current transfer allowances and moving costs paid by the Company for Management or Special Skills Personnel moves or transfers will be applicable to Employer requested moves or employees successful on job postings.
- 5. EMPLOYEES TRAVELLING TO AND FROM OUT-OF-TOWN JOBS
- 5.01 Travelling employees is a term defining Operator Services employees who spend all or the greater portion of their time in work locations outside their defined headquarters. It is not applicable when such employees are assigned to work wholly located within their respective headquarters.
- 5.02 The immediate Supervisor shall consider the following factors in order to reach a decision on whether or not daily or weekend travel is to be allowed:
 - (a) Distance from headquarters
 - (b) Road and weather conditions
 - (c) Nature and duration of the job
 - (d) Overtime requirements, if any
 - (e) Availability of transportation
 - (f) Accommodation and eating facilities
 - (g) Employee or group requirements as they may relate to reasonable circumstances
- 5.03 Travelling time when starting or completing a job in a town or district away from headquarters will be on Company time. If travelling extends into the employee's own time, the overtime rate will prevail.
- 5.04 Travelling to and from headquarters on a daily basis may be authorized by the immediate Supervisor. Where daily travel is authorized, the Company will allow the employee to travel to and from the job site on Company time.

- 5.05 Travelling employees working within a **three hundred and twenty (320) kilometre** radius of their headquarters may be allowed to return, every weekend, subject to the conditions of Clause 5.02. Where weekend travel is authorized, the employee shall travel to and from the job site on Company time.
- 5.06 The Supervisor shall determine whether an employee remaining at the job site over the weekend, while others are returning home, shall be paid meals and lodging.
- 5.07 Travelling employees working outside a **three hundred and twenty (320) kilometre** radius of headquarters shall be allowed to return at least every second weekend. Where such travel is authorized, the Company will allow the employee to travel to and from the job site on Company time.

6. TRANSPORTATION

- (a) For the purpose of conducting Company business, transportation will be furnished or allowed all employees covered by this Agreement where same is considered necessary by the Company. It is understood and agreed that an employee cannot be required to use their personal automobile as a condition of employment; the use of employee's personal automobile is strictly at the employee's option.
- (b) Management will determine the type and method of transportation by bus, rail, car or plane and whenever possible, transportation will be in a Company vehicle.
- (c) An employee, who is authorized by Management to use their own vehicle will be allowed actual car kilometerage at the prevailing rate per kilometer providing there is proper insurance coverage on the vehicle being used.
- (d) Employees authorized to travel by private automobile will be covered by compensation, providing there is no deviation from the direct route to headquarters or any unauthorized stopovers enroute.
- (e) When two (2) or more employees are travelling in the same private automobile, the owner is the only person eligible to claim **kilometerage**.
- (f) Any claim arising from an accident to an employee's own vehicle will be the responsibility of the said employee.

- (g) Where Company vehicles are not available, Supervisors may arrange for the use of private automobiles (with employee's consent) or alternate transportation to transport employees to and from headquarters.
- (h) An employee who resigns, is dismissed or laid off in the field will be furnished or allowed transportation to **their** district headquarters or place of hire.

Signed by:	Signed by:
L. Emanuel	Grace Thostenson
TELUS	LOCAL 348
	International Brotherhood of
	Electrical Workers

Dated: October 23, 1999

Dated: October 23, 1999

APPENDIX C

OPERATOR SERVICES

1. WORKING SUPERVISION POLICY

The Corporate Management committee does not expect Management personnel to do the work which is normally carried out by operators. However, occasions may arise, such as in training demonstrations, quality control checks, restoration or maintenance of service where an accident occurs, or urgent/emergency work is necessary to plant, equipment or machinery, or other unforeseeable or unpreventable circumstances occur, where Management personnel will be required to do work normally performed by an Operator. These occasions should be considered exceptions and not normal operating practice.

Signed by: L. Emanuel TELUS Signed by: Grace Thostenson LOCAL 348 International Brotherhood of Electrical Workers

Dated: October 23, 1999

Dated: October 23, 1999

Absence	10
Accident Compensation	
Bereavement Leave	
Court Witness	
Funeral Leave	
Furlough	
Jury Duty	
Medical, Dental or Other Special Appointments	
Paid (RPT)	
Pallbearer Leave	
Serious Distress	
Sick Leave	
Time Lost Other Than Sickness	
Work for Another Company	
Accident	10
Casual Employee	
Compensation Schedule	
Occasional Employee	
ACCIDENT COMPENSATION	
Acting Assignments	
Differentials	
Filling Acting Assignments	
Maximum Length	
Returning from Acting Assignments	
ADOPTION LEAVE	
Vacation Credits	
Afternoon Tour (Op Svcs)	
Alberta Health Care	
Adoption Leave	
Child Care Leave	
Maternity Leave	
Alberta Health Care Insurance Plan	
ALBERTA HEALTH CARE INSURANCE PLAN, SUPPLEMENTARY HEALTH, VISION, DENTAL CARE	
GROUP INSURANCE PLAN	49
ANNUAL VACATIONS	
See also Vacations	

Α

 Basic Hourly Rate (Clerical)
 4

 Basic Hourly Rate (Op Svcs)
 6

 Basic Rate of Pay (Clerical)
 3

 Basic Rate of Pay (Op Svcs)
 6

 Basic Wages (Clerical)
 4

 Basic Wages (Clerical)
 4

 Basic Wages (Op Svcs)
 6

 Benefits
 6

 Adoption Leave
 40

 Alberta Health Care
 49

 Child Care Leave
 39

 Dental Plan
 50

В

Employee Benefits	
Employee Benefits Group Insurance Maternity Leave Pay in Lieu, Occasional (Clerical)	
Maternity Leave	
Pay in Lieu, Occasional (Clerical)	
Pay in Lieu, Occasional (Op Svcs)	
Premiums (RPT)	
Regular Part-Time Employees	50
Stop Payment of Sick Leave	49
Supplementary Health Care	
Vision Care	
Benefits Committee (Letter of Understanding - Clerical, Op Svcs, Craft)	
Bereavement Leave	75
Boxing Day	
Boxing Day Break Entitlement - Overtime	
Break Period (Clerical)	
Break Period (Op Svcs)	
Bulletin Boards	7

С

Casual Employee	
Definition (Clerical)	2
Definition (Op Svcs)	5
Hired to RFT Probation Period (Op Svcs)	5
Hours of Work (Op Svcs)	
Pay for Holidays	28
Term of Employment (Clerical)	2
Term of Employment (Op Svcs)	
Vacation Entitlement	
Vacation Pay	36
Casual Employee (Op Svcs)	5
Christmas Day	25
Christmas Float (Clerical)	25
Christmas Float (Op Svcs)	26
Christmas Special Payment (Clerical)	71
Christmas Special Payment (Op Svcs)	72
Civic Holidays	26
CLASSIFICATIONS (Operator Services)	1
Clerical	
Absence	
Compressed Work Week (Letter of Understanding)	92
Consecutive Days Off	17
Definitions	2
Differentials & Special Payments	
Electronic Monitoring (Letter of Understanding)	95
Family Days	20
Headquarters, Definition of	4
Holidays	
Hours of Work & Tours of Duty	15
Layoff	54
Return to Work, Sick Leave	49
Sick Leave Notification	44
Technological Change	53
Travel & Transfer Expenses, Appendix A	123
Vacation Minimum 10%	32
Vacation Single Day Selection	33

Working Supervision	9
Company Holidays	
Compressed Hours of Work Program (Letter of Understanding - IT Computer Operators)	. 91
COMPRESSED WORK WEEK, CLERICAL (Letter of Understanding)	. 92
Consecutive Days Off (Clerical)	
Court Witness	

D

Day Tour (Clerical)	15 16
Day Tour (Op Svcs)	
Days off Consecutive (Clerical)	
Definitions	
Basic Hourly Rate (Clerical)	4
Basic Hourly Rate (Op Svcs)	
Basic Rate of Pay (Clerical)	
Basic Rate of Pay (Op Svcs)	
Basic Wages (Clerical)	
Casual Employee (Clerical)	
Casual to RFT (Op Svcs)	
Furlough (Clerical)	
Furlough (Op Svcs)	
Headquarters (Clerical)	
Meal Period (Op Svcs)	
Occasional Employee (Clerical)	
Occasional Employee (Op Svcs)	
Occasional to RFT (Op Svcs)	
Of a Day	
Overtime (Clerical)	
Overtime (Op Svcs)	6
Probationary Employee (Clerical)	
Probationary Employee (Op Svcs)	5
Regular Full-Time Employee (Clerical)	2
Regular Full-Time Employee (Op Svcs)	4
Regular Part-Time Employee (Clerical)	2
Regular Part-Time Employee (Op Svcs)	5
Session (Clerical)	
Session (Op Svcs)	
Tour Bidding (Op Svcs)	
Tour Employee (Clerical)	
Tour of Duty (Clerical)	
Tour of Duty (Op Svcs)	
Wages (Op Svcs)	
DEFINITIONS (Clerical)	
DEFINITIONS (Operator Services)	
Dental Plan	
Adoption Leave	
Child Care Leave	
Maternity Leave	
Differential	
Afternoon Tour (Op Svcs)	
Christmas (Clerical)	
Christmas Special Payment (Op Svcs)	
Consecutive Saturdays (Op Svcs)	
Day Tour (Op Svcs)	
Evening Tour (Op Svcs)	71

Higher Rated Job (Clerical)	
Higher Rated Job (Op Svcs)	
New Year's Eve (Clerical)	
New Year's Eve Special Payment (Op Svcs)	
Night Tour (Clerical)	
Night Tour (Op Svcs)	
Saturday (Clerical)	
Service Asistant (Op Svcs)	
Service Assistant and Support	
Split Tour (Op Svcs)	
Sunday Premium (Clerical)	
Sunday Premium (Op Svcs)	
Tour (Clerical)	
Training (Clerical)	
Training (Op Svcs)	
DIFFERENTIALS AND SPECIAL PAYMENTS (Clerical)	
DIFFERENTIALS AND SPECIAL PAYMENTS (Op Svcs)	
Directives	
DISCIPLINE, RESIGNATION AND DISMISSAL	
Discipline	
Dismissal	
DISCRIMINATION	
DUES DEDUCTIONS	
DURATION AND NEGOTIATIONS	

Ε

Faster Manday	7
Easter Monday	11 71
ELECTRONIC MONITORING, CLERICAL (Letter of Understanding)	
	15
	2
Casual (Clerical)	
Casual (Op Svcs)	
Casual to RFT (Op Svcs)	
Occasional (Clerical)	
Occasional (Op Svcs)	
Occasional (Residential Sales & Service) Letter of Understanding	
Occasional to RFT (Op Svcs)	
Probationary (Clerical)	
Probationary (Op Svcs)	
Regular Full-Time (Clerical)	2
Regular Full-Time (Op Svcs)	
Regular Part-Time (Clerical)	2
Regular Part-Time (Op Svcs)	5
Regular Part-Time, Casual & Occasional Employees (Letter of Understanding))9
Tour (Clerical)	3
Tour (Op Svcs)	18
Evening Tour (Clerical) 1	16
Evening Tour (Op Svcs)	
Exhaustion Of Sick Leave Credit	
Extraordinary Expenses7	

F

FAMILY DAYS (Clerical)	
Family Days (Letter of Understanding)	

Footwear, Safety	. 41
Funeral Leave	
Furlough	. 77
Furlough (Op Svcs)	6

G

GRIEVANCE PROCEDURE	
STEP 1	
STEP 2	
STEP 3	
STEP 4	
Group Insurance	
Adoption Leave	
Child Care Leave	
Maternity Leave	
·····	

Η

Headquarters (Clerical) Holidays	4
Christmas Float (Clerical)	
Christmas Float (Op Svcs)	26
Civic	26
Clerical	25
Company (RPT)	50
Company Holidays	
Easter Monday	
Holiday Night Tour (Op Svcs)	28
Operator Services	25
Pay	27
Reducing Calendar Work Week	
Tour Workers (Clerical)	
Tour Workers (Op Svcs)	
HOLIDAYS AND HOLIDAY PAY	25
Hours Of Work	
Afternoon Tour (Op Svcs)	18
Basic Hours (Clerical)	16
Basic Hours (Op Svcs)	19
Break Period (Op Svcs)	
Day tour (Clerical)	16
Day Tour (Op Svcs)	
Day Tour Definition (Clerical)	15
Deviation from Normal Basic Hours (Clerical)	17
Evening Tour (Clerical)	16
Evening Tour (Op Svcs)	18
Hours per Week (Op Svcs)	19
Meal Period Exceptions (Clerical)	
Meal Period Exceptions (Op Svcs)	
Minimum Hours per Tour (Op Svcs)	19
Night Tour (Clerical)	16
Night Tour (Op Svcs)	18
Overtime (Clerical)	15
Overtime (Op Svcs)	19
Relief or Meal Period (Clerical)	
Relief or Meal Period (Op Svcs)	19

	Schedule Changes (Clerical)	. 16
	Schedules Changes (Op Svcs)	. 20
	Session (Op Svcs)	. 18
	Split Tours (Op Svcs)	. 19
	Tour of Duty (Clerical)	
	Tour of Duty (Op Svcs)	. 18
	Week (Clerical)	. 16
	Week (Op Svcs)	
H	DURS OF WORK AND TOURS OF DUTY (Clerical)	. 15
н	DURS OF WORK AND TOURS OF DUTY (OPERATOR SERVICES)	. 18

I

J

Incidental Expenses

JOB ALLOCATION APPEAL PROCEDURE (Letter of Understanding)	58 59 61 62 62 64 63 61 61 61 00
JOB SHARING (Letter of Understanding)	04

Layoff	
Casual Employee (Clerical) Casual Employee (Op Svcs)	
Casual Employee (Op Svcs)	
Occasional Employee (Clerical)	
Occasional Employee (Op Svcs)	
Order of (Clerical)	
Order of (Op Svcs)	
Order of Recall (Clerical)	
Order of Recall (Op Svcs)	
Recall Retain Credits (Op Svcs)	
Recall, Retain Credits (Clerical)	
Severance Pay (Clerical)	
Severance Pay (Op Svcs)	
Waive Right to Recall (Clerical)	
Waive Right to Recall (Op Svcs)	
LAYOFF (Clerical)	
LAYOFF (Operator Services)	
Letter of Understanding	
Benefits Committee	
Compressed Hours of Work Program (IT Computer Operators)	
Compressed Work Week (Clerical)	

Electronic Monitoring (Clerical)	
Family Days	
Job Allocation Appeal Procedure Job Sharing	
Job Sharing	
Joint Committee Grievance Review	
Letters of Understanding/Letters of Intent	89
Medical & Compassionate Transfers	
Occasional Employees - Residential Sales & Service	
Regular Part-Time, Casual & Occasional Employees	
Sales Incentive Plan	
SAP Administration	
Service Assistant (Special Services), Service Assistant & Support Duties (Op Svcs)	
Severance Pay	
Supplementary Health and Vision Care Plans	117
Taxis	
Telecommunications Concession	119
Trading of Tours Between Services (Op Svcs)	
Vacation Entitlement	
Work Experience Programs	
Work Experience Programs	

Μ

MANAGEMENT RIGHTS	9
MATERINTY LEAVE	
Vacation Credits	
MATERNITY LEAVE	
Meal Period (Clerical)	
Meal Period (Op Svcs)	6
Meals	
MEALS, LODGING, TRANSPORTATION AND TRAVELLING TIME	
MEDICAL AND COMPASSIONATE TRANSFERS (Letter of Understanding)	105, 106
Medical, Dental or Other Special Appointments	
Meetings - Discipline, Resignation & Dismissal	
Monitoring, Electronic (Clerical) (Letter of Understanding)	

Ν

New Year's Eve Special Payment (Clerical)	
New Year's Eve Special Payment (Op Svcs)	
Night Tour (Clerical)	
Night Tour (Op Svcs)	
Night Tour, Holiday (Op Svcs)	
Notice of Resignation	
Notification of Results of Job Posting	

Occasional Employee 18 Break Period (Clerical) 18 Definition (Clerical) 3 Definition (Op Svcs) 5 Hired to RFT Probation Period (Op Svcs) 5 Minimum Hours per Tour (Op Svcs) 19 Pay for Holidays 28

Ο

Pay in Lieu of Benefits (Clerical)	
Pay in Lieu of Benefits (Op Svcs)	
Vacation Entitlement	
Vacation Pay	
Operator Services	
Absence	
Classifications	
Definitions	
Differentials & Special Payments	
Holiday Night Tour	
Holidays	
Layoff	
One Week Selection Periods	
Return to Work, Sick Leave	
Service Assitant (Special Services), Centralized Service Assistant & Service Assistant & Support	Duties (Letter of
Understanding)	
Sick Leave Notification	
Technological Change	
Trading of Tours Between Services (Letter of Understanding)	
Transfers	
Travel & Transfer Expenses, Appendix B	
Vacation Minimum 10%	
Working Supervision Policy, Appendix C	
Overtime	
Application of Requirements	
Authorization of	
Banked Time Off	
Break Entitlement	
Call back within 24 hours (Clerical)	
Callout	
Day Off, Required to Work	
Definition of, (Clerical)	
Definition of, (Op Svcs)	
Excess of 40 hours Banked	
Meal Allowance	
Overtime (Clerical)	
Paid for Work in Excess (Op Svcs)	
Part-time Employees	
Rate of Pay	
Rest Period	
Schedule of V.O. time off	
Vacation Overtime (V.O.)	

Pallbearer Leave	
Pay	
Basic Hourly Rate (Op Svcs)	
Basic Rate (Op Svcs)	6
Basic Rate, Definition of (Clerical)	
Change of Tour Payment (Clerical)	
Holidays	
Hourly Rate, Definition of (Clerical)	
Pav Period	
Severance (Clerical)	
Severance (Letter of Understanding)	
Severance Pay (Op Svcs)	

Ρ

Suspension of Increases	
Suspension of Increases Treatment for Redeployed Employees	
Vacation	
Vacation Pay for Casual Employee	
Vacation Pay for Occasional Employee	
Vacation Pay in Lieu	
Variable	
Wage Adjustment, Job Evaluation Wage Advance, Vacation	
Wage Advance, Vacation	
Wage Increases Wages (Op Svcs)	
Wages (Op Svcs)	
PAY IN LIEU OF VACATION	
PAY PROVISIONS AND WAGE SCHEDULES	
PAY PROVISIONS FOR EMPLOYEES ABSENT FROM DUTY	74
Premium	
Sunday (Clerical)	
Sunday (Clerical) Sunday (Op Svcs)	
Probationary Employee (Clerical)	
Probationary Employee (Op Svcs)	

Q

Quarantine, Sick Leave.....

R

Recall	
Order of (Clerical)	
Order of (Op Svcs)	
Retain Credits (Clerical)	
Retain Credits (Op Svcs)	
Waive Recall Rights (Clerical)	
Waive Right to Recall (Op Svcs)	
Redeployment	
Closure of Office / Department	
Pay Treatment for Employees	
Process	
Relocation of Office / Department	
Surplus Position	
Vacancy Created by Layoff	
Refusal to Work, Safety	
Regular Full-Time Employee (Clerical)	
Regular Full-Time Employee (Op Svcs)	
REGULAR PART-TIME BENEFITS	
Regular Part-Time Employee	
Benefit Premiums	
Benefits	
Break Period (Clerical)	
Company Holidays	
Family Days (Clerical)	
Holiday Pay	
Minimum Hours per Tour (Op Svcs)	
Paid Absences	
Pay Periods	
Progressional Increases	
Seniority	

Seventy Five Percent (75%) Rule	
Sick Leave	
Vacation	
Regular Part-Time Employee (Clerical)	2
Regular Part-Time Employee (Op Svcs)	
REGULAR PART-TIME, CASUAL AND OCCASIONAL EMPLOYEES (Letter of Understanding)	
Residential Sales & Service Occasional Employees (Letter of Understanding)	
Resignation, Notice of	

S

	40
SAFETY	
Footwear	
Refusal to Work	
Representatives	
Requirement to Work	
Training	
Video Display Terminals	
Sales Incentive Plan (Letter of Understanding)	
SAP ADMINISTRATION (Letter of Understanding)	
Saturday Differential (Clerical)	
Saturdays, Consecutive (Op Svcs)	
Schedules (Op Svcs)	
SCOPE	
SENIORITY	10
Seniority (RPT)	
Separation Allowance (Clerical)	53
Separation Allowance (Op Svcs)	54
Serious Distress	76
Service Assistant (Op Svcs)	
SERVICE ASSISTANT (SPECIAL SERVICES), CENTRALIZED SERVICE ASSISTANT & SERVICE ASSISTANT	. &
SUPPORT DUTIES (OP SVCS) (Letter of Understanding)	
Service Assistant and Support	66
SERVICE DATE	
Session, Definition of (Clerical)	
Session, Definition of (Op Svcs)	
Severance Pay (Clerical)	
SEVERANCE PAY (Letter of Understanding)	
Severance Pay (Op Svcs)	
Sick Leave	
Exhaustion of Credits	
Medical Documentation	
Notification (Clerical)	
Notification (Op Svcs)	
Quarantine	
Regular Part-Time Employee	
Return to Work (Op Svcs)	
Return to Work (Clerical)	
Schedule (Credits)	
Schedule (Credits)	
Unable to Continue Work	
Use of Vacation Credits	
SICK LEAVE REGULATIONS	
SICK LEAVE REGULATIONS	
Sunday Premium (Clerical)	
Sunday Premium (Op Svcs)	
Supplementaty realth and vision care Plans (Letter of Understanding)	. 117

Supplementary Health Care	
Adoption Leave	
Child Care Leave	
Maternity Leave	
Surplus Position	
Backfilling	
Redeployment	
Suspension of Increases	
Child Care Leave	
Maternity Leave	

Т

TAXIS (Letter of Understanding)	
FECHNOLOGICAL CHANGE (Clerical)	
FECHNOLOGICAL CHANGE (Operator Services)	. 54
Felecommunications Concession (Letter of Understanding) 1	119
Temporary Assignments	
Acting Assignments	. 68
Higher Rated Job (Clerical)	. 66
Higher Rated Job (Op Svcs)	. 67
Management (Clerical)	. 65
Management (Op Svcs)	. 66
Other Bargaining Units	. 68
Service Assistant and Support (Op Svcs)	. 66
Within Bargaining Unit	
TEMPORARY MOVEMENT OF EMPLOYEES	. 65
TIME OFF FOR UNION REPRESENTATIVES	7
Four Employee (Clerical)	3
Fours	
Afternoon Differential (Op Svcs)	. 71
Afternoon Tour (Op Svcs)	
Change of Tour Payment (Clerical)	
Day Tour (Clerical)	16
Day Tour (Op Svcs)	
Day Tour Differential (Op Svcs)	. 71
Definition of, (Clerical)	3
Definition of, (Op Svcs)	6
Differentials (Clerical)	
Evening Differential (Op Svcs)	. 71
Evening Tour (Clerical)	. 16
Evening Tour (Op Svcs)	
Evening Tour Differential (Op Svcs)	
Holiday Night Tour (Op Svcs)	
Hours of Work (Clerical)	. 16
Meal period (Clerical)	. 16
Meal Period (Op Svcs)	
Night Tour (Clerical)	. 16
Night Tour (Op Svcs)	. 18
Night Tour Differential (Op Svcs)	
Night Tour Differential (Clerical)	
Night Tour Differential (Op Svcs)	
Overtime Paid for Call-Backs (Clerical)	
Relief or Meal Period (Clerical)	
Relief or Meal Period (Op Svcs	
Rotation of (Clerical)	
Rotation of (Op Svcs)	. 20

Schedules Posted (Op Svcs)	
Schedules Posted (Op Svcs) Split Tour (Op Svcs)	
Split Tour Differential (Op Svcs)	
Tour Bidding (Op Svcs)	6
Tour Employee (Op Svcs)	
Tour of Duty (Clerical)	
Tour of Duty (Op Svcs)	
Unable to Continue due to Illness (Op Svcs)	
TRADING OF TOURS BETWEEN SERVICES, OP. SVCS (Letter of Understanding)	
Training	
Differential (Clerical)	
Differential (Clerical) Differential (Op Svcs)	73
Safety	
Upgrading Courses	61
Transfers	
Job Posting Exceptions	
Medical & Compassionate (Letter of Understanding)	105, 106
Operator Services	
Transfer Date	
TRAVEL AND TRANSFER EXPENSES APPENDIX A (Clerical)	
TRAVEL AND TRANSFER EXPENSES APPENDIX B (Op Svcs)	
Travel Time	80

U

Union	
Condition of Employment	
Dues Deductions	
Dues, Adoption Leave	
Dues, Child Care Leave	
Dues, Maternity Leave	
Job Postings	
List of Employees	
Meet New Employees	
Membership in	
Negotiating Committee	
Time off for Union Representatives	7
	7
UPGRADING COURSES	

V	
Vacation	
Accrual of (Clerical)	
Accrual of (Op Svcs)	
Accrual of (Op Svcs) Annual Vacations	
Banked Time Off	
Casual Employee	
Credits	
Entitlement	
Granted	
Holiday occurs during Minimum 10% (Clerical)	
Minimum 10% (Clerical)	
Minimum 10% (Op Svcs)	33
Occasional Employee	
One Week Selection (Op Svcs)	

	Open Period	34
	Overtime (V.O.)	23
	Pay	
	Pay for Casual Employee	
	Pay for Occasional Employee	36
	Pay in Lieu	
	Reduction of Credits	29
	Regular Part-Time Employee	53
	Schedule of V.O. time off	
	Schedule Posted or Circulated	32
	Scheduling	32
	Single Day Selection (Clerical)	33
	Transfer of Employee	32
	Wage Advance	31
	Year	29
	ACATION CARRY OVER CALCULATIONS, EXAMPLE OF	
V	ACATION ENTITLEMENT (Letter of Understanding)	121
	ariable Pay	
V	deo Display Terminals	42
V	sion Care	50
	Adoption Leave	40
	Child Care Leave	
	Maternity Leave	37

W

WAGE SCHEDULES	83
1998 Daily, Monthly and Yearly Wage Equivalents	
1999 Daily, Monthly and Yearly Wage Equivalents	86
2000 Daily, Monthly and Yearly Wage Equivalents	87
Administration of	82
HOURLY RATES (by Paygroup)	84
Work Experience Programs (Letter of Understanding)	
Work for Another Company	76
Work Week (Clerical)	15
Work Week, Calendar (Clerical)	
Work Week, Calendar (Op Svcs)	19
WORKING SUPERVISION (Clerical)	9
WORKING SUPERVISION POLICY, APPENDIX C (Op Svcs)	